ATTORNEY DOCKET NO: M-11338 US 05-24-2001 RECC PATENT AND TRADEMARK OFFIC MAY 1 6 2001 101727362 TO THE HONORAGE PARENTOR OF THE UNITED THE COPY THEREOF. ... FICE. PLEASE RECORD THE ATTACHED 1. Name of conveying party(ies): 2. Name and address of receiving party(ies): Analogy, Inc. Name: Avant! Corporation (a) 46871 Bayside Parkway Street Address: City: Fremont CA 94538 Additional name(s) of conveying party(ies) attached? Country: USA ⊠ No Yes 3. Nature of Conveyance: Name and address of receiving party(ies): Additional name(s) & address(es) attached? Merger Assignment Yes Security Agreement Change of Name Other Execution Date: April 26, 2001 4. Application number(s) or patent number(s): If this document is being filed together with a new application, the execution date of the application is: Patent No.(s) Patent Application No.(s) - 09/531,350 Scheduling Non-Integral Simulation Time For Mixed-Title: Signal Simulation \boxtimes Yes Additional numbers attached? 6. Total number of applications and patents involved: Name and address of party to whom correspondence concerning document should be mailed: Omkar K. Suryadevara Name: SKJERVEN MORRILL MacPHERSON LLP Internal Address: \$40.00 7. Total fee (37 CFR 3.41): 25 METRO DRIVE, SUITE 700 Street Address: Authorized to be charged to Deposit Account 19-2386 State <u>CA</u> SAN JOSE Charge Deposit Account 19-2386 for any additional fees required for this conveyance and credit deposit account

8. Statement and signature.

To the best of my knowledge and belief, the foregoing information is true and correct and any attached copy is a true copy of the

DO NOT USE THIS SPACE

original document.

Omkar K. Suryadevara Name of Person Signing

Total number of pages comprising cover sheet: Merger document 115 pages Recordation Cover sheet 1 page. Total of 116 pages

19-2386 any amounts overpaid

PATENT REEL: 011820 FRAME: 0298



		PAGE
Section 1	The Merger	1
1.1	The Closing	
1.2	Effective Date of Merger	1
1.3	Effects of Merger	1
Section 2	Representations and Warranties of the Company	1
2.1	Capital Stock.	1
2.2	Organization; Good Standing	3
2.3	Authority; Enforceability	3
2.4	No Violation	3
2.5	Subsidiaries, Other Interests.	3
2.6	Financial Statements.	4
	(a) Financial Statements	4
	(b) Certain Indebtedness	4
	(c) Absence of Certain Liabilities	4
	(d) Absence of Certain Changes	
2.7	Taxes.	5
2.8	Title to Properties	6
2.9	Inventories	6
2.10	Accounts Receivable	7
2.11	Leases	/
2.12	Facilities, Equipment.	7
2.13	Insurance.	7
2.14	Employment and Benefit Matters	
2.15	Contracts.	10
2.16	Officers and Directors	10
2.17	Corporate Documents.	.10
2.18	Legal Proceedings.	11
2.19	Orders and Legal Requirements	
2.20	Permits	***************************************

(Continued)

		PAGE
2.21	Intellectual Property.	11
2.22	Capital Expenditures	14
2.23	Environmental Matters	14
2.24	Illegal Payments	15
2.25	SEC Information.	15
2.26	Board of Directors Approval; Fairness Opinion	15
2.27	No Dissenters' Rights	16
2.28	Option Substitution	16
2.29	Employee Stock Purchase Plan	16
2.30	Transfers of Technology	16
2.31	Representations.	16
Section 3	Representations and Warranties of Buyer	16
3.1	Organization, Standing of Buyer and Buyer Subsidiary	
3.1	Authority; Enforceability.	
	Litigation	
3.3	Proxy Materials	
3.4		
Section 4	Conditions to Obligations of Buyer at Closing	17
4.1	Representations and Warranties.	17
4.2	Proxy Statement	18
4.3	Closing Certificate	18
4.4	Performance	18
4.5	Stockholder Approval.	18
4.6	Third-Party Action.	18
4.7	Opinion of Counsel.	18
4.8	Transactional Litigation	18
4.9	Interim Events	18
4.10	Changes Technical Employees	
4.11	Employment and Noncompetition Agreements.	

(Continued)

		PAGE
4.12	Option Holder Approval	19
4.13	Nonemployee Option Consents	19
4.14	Financial Statement Compliance	19
4.15	Corporate and Other Proceedings.	19
4.16	Cancellation of Company 401(k) Plan	19
4.17	Officer and Director Resignations	19
4.18	Advisory Opinion	19
4.19	DARPA Letter	20
Section 5	Conditions to Company's Obligations at Closing	20
5.1	Representations and Warranties	20
5.2	Closing Certificate	20
5.3	Performance	20
5.4	Stockholder Approval.	20
5.5	Third-Party Action.	20
5.6	Opinion of Counsel	20
5.7	Transactional Litigation	
5.8	Corporate and Other Proceedings.	21
Section 6	Covenants of Company, Buyer Subsidiary and Buyer	21
6.1	Non-Disclosure.	21
6.2	Nonsurvival of Representations and Warranties	21
6.3	Termination of this Agreement; Termination Fees	21
6.4	Reasonable Business Efforts, No Inconsistent Action	22
6.5	Access	22
6.6	No Solicitation or Negotiation	
6.7	Interim Financial Information	
6.8	Interior Conduct of Business	.,,,,,,,
6.9	a .: 229 Floation: Tay Status	
6.10	Option to Purchase	

(Continued)

		PAGE
6.11	SEC Reports	31
6.12	Notice of Certain Events	31
6.13	Takeover Statutes	31
6.14	Pay-Off	31
6.15	Other Filings	31
6.16	Officers and Directors Indemnification	32
Section 7	Miscellaneous	33
7.1	No Brokers, Finders	33
	(a) Company	33
	(b) Buyer	33
7.2	Expenses.	33
7.3	Complete Agreement; Waiver and Modification; No Third Party Beneficiaries	33
7.4	Notices	34
7.5	Law Governing	35
7.6	Headings; References; "Hereof;" Interpretation	35
7.7	Successors and Assigns.	35
7.8	Counterparts, Separate Signature Pages	35
7.9	Severability.	35
Section 8	Glossary	35

Agreement of Merger

This Agreement of Merger dated December 2, 1999 is entered into by Avant! Corporation, a Delaware corporation (the "Buyer"), AC Acquisition Corp., an Oregon corporation and wholly owned subsidiary of the Buyer ("Buyer Subsidiary"), and Analogy, Inc., an Oregon corporation (the "Company"). The Company is a developer and marketer of analog and mixed-technology design systems and model libraries, including without limitation the Saber and Verias simulator programs (the "Business").

Capitalized terms used herein have the meanings stated in Section 8.

The Buyer and the Company desire that the Buyer acquire the Company through a merger of Buyer Subsidiary with and into the Company (the "Merger"), and the Company desires to consummate the Merger, under the terms of this Agreement.

Therefore, in consideration of the mutual agreements contained herein, the parties hereby agree as follows:

Section 1 The Merger

- 1.1 Closing. The closing (the "Closing") under this Agreement shall take place at the offices of Ater Wynne LLP within 3 business days after the satisfaction (or waiver by the party entitled to waive) of all conditions stated in Sections 4 and 5, or at such other place or on such other date as the parties may agree in writing.
- 1.2 Effective Date of Merger. The Merger shall take effect upon filing of articles of merger with respect to the Plan of Merger in the form attached hereto as Exhibit A and incorporated herein by this reference (the "Plan of Merger") with the Oregon Secretary of State in accordance with Oregon law (the "Effective Time").
 - 1.3 Effects of Merger. The effects of the Merger are set forth in the Plan of Merger.

Section 2 Representations and Warranties of the Company

Except as otherwise set forth in the Disclosure Schedule attached hereto and incorporated herein (which Disclosure Schedule shall identify the section or sections of this Agreement to which such exceptions relate), the Company hereby represents and warrants to the Buyer as follows:

2.1 Capital Stock.

(a) The authorized and outstanding capital stock of the Company, as of December 1, 1999, is as follows:

PATENT REEL: 011820 FRAME: 0303

Designation of Class	Shares Authorized	Shares Outstanding
Common Stock, no par value	35,000,000	9,631,911
Preferred Stock, no par value	5,000,000	0

There is no capital stock of the Company authorized or outstanding except as stated in this Section 2.1(a). The outstanding Stock Rights of the Company are as follows:

Designation of Stock Right	Class of Stock	Shares Subject to Stock Right
Options under 1986 Stock Option Plan	Common	24,687
Options under the Amended and Restated 1993 Stock Incentive Plan	Common	1,578,724
Options under 1995 Stock Option Plan for Nonemployee Directors	Common	100,000
Warrants issued to Transamerica Business Credit Corporation	Common	10,000

There are no Stock Rights outstanding with respect to the Company except as set forth in this Section 2.1(a), and the terms of such Stock Rights are as set forth in Schedule 2.1. Except as disclosed in Schedule 2.1, the Company is not a party to any stockholders agreement, registration rights agreement, repurchase agreement or other Contract with respect to capital stock or Stock Right issued or to be issued by it.

- (b) All of the issued and outstanding capital stock of the Company has been duly and validly authorized and issued and is fully paid and non-assessable, and has not been issued in violation of any preemptive or similar rights of any stockholder or any applicable securities law. Except as disclosed in *Schedule 2.1*, no Person has any right to require the Company to redeem, purchase or otherwise reacquire any capital stock issued by the Company or any Stock Rights with respect to any capital stock issued by the Company. There are no preemptive or similar rights in respect of any capital stock of the Company except as set forth in *Schedule 2.1*.
- (c) The Company has never declared or paid any dividend or made any distribution in respect of any of its capital stock or any Stock Rights with respect thereto. Since September 30, 1999, except as set forth in *Schedule 2.1*, the Company has not directly or indirectly redeemed, purchased or otherwise acquired any of the capital stock issued by it or any Stock Rights with respect thereto.

- 2.2 Organization; Good Standing. The Company is a corporation duly organized, validly existing and in good standing under the laws of Oregon and has all requisite corporate power and authority to own, lease and operate its Properties and to conduct the Business as currently conducted. Schedule 2.2 sets forth (i) each jurisdiction in which the Company is qualified to do business as a foreign corporation, (ii) the jurisdiction of incorporation of each Subsidiary and (iii) each jurisdiction in which a Subsidiary is qualified to do business as a foreign corporation. The Company and each Subsidiary is in good standing in each jurisdiction shown in Schedule 2.2, and neither the Company nor any Subsidiary is required to qualify to do business as a foreign corporation in any other jurisdiction in which the failure to so qualify would have a Material Adverse Effect. The Company holds all of the outstanding shares of each Subsidiary. Neither the Company nor any Subsidiary is a partner in any general or limited partnership or a member in any limited liability company.
- 2.3 Authority; Enforceability. The Company has all requisite power and authority under applicable corporate law to execute and deliver this Agreement and to perform the transactions contemplated hereby and, subject to approval of the Plan of Merger by the stockholders of the Company as contemplated hereby (the "Stockholder Approval"), to consummate the Merger. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby have been duly authorized by all requisite corporate action on the part of the Company (subject to the Stockholder Approval), and no other approval on the part of the Company is necessary under applicable corporate law for the execution, delivery and performance of this Agreement. This Agreement constitutes a legal, valid and binding obligation of the Company, enforceable in accordance with its terms, subject to general limitations on the availability of equitable remedies and the effect of bankruptcy, insolvency, reorganization and other laws of general application affecting the enforcement of creditors' rights.
- No Violation. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby do not and will not violate or conflict with the articles of incorporation or by-laws of the Company or any Subsidiary, or, to the knowledge of the Company, violate any Legal Requirement or Order applicable to the Company or any Subsidiary. execution and delivery of this Agreement and the consummation of the transactions contemplated hereby do not and will not (i) require any Third-Party Action with respect to the Company or any Subsidiary under, or (ii) conflict with or constitute a default under, or result in the acceleration or right of acceleration of any obligations, or any termination or right of termination under, either (x) any Contract (other than a Designated Contract) where such failure to secure Third-Party Action, conflict, default, acceleration or termination would have a Material Adverse Effect or (y) any Designated Contract. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby do not and will not (i) result in the creation or imposition of any material Lien, claim, charge, restriction, equity or encumbrance of any kind upon or give any Person any interest or right in or with respect to any of the Properties, assets, business or Contracts of the Company or any Subsidiary otherwise than pursuant to a Designated Contract or (ii) result in the creation or imposition of any Lien, claim, charge, restriction, equity or encumbrance of any kind upon or give any Person any interest or right in or with respect to any of the Properties, assets, business or Contracts of the Company or any Subsidiary pursuant to a Designated Contract.
- 2.5 Subsidiaries, Other Interests. Except as set forth in Schedule 2.5, neither the Company nor any Subsidiary beneficially owns, directly or indirectly, any Equity Interest in or debt

obligation (except as a creditor in the ordinary course of business and except for debt obligations of the Company or a Subsidiary as specified in *Schedule 2.5*) of, any Person. The owner shown in *Schedule 2.5* owns the interest shown free and clear of all Third-Party Rights.

2.6 Financial Statements.

- (a) Financial Statements. Except as set forth in Schedule 2.6(a), the audited consolidated balance sheets and consolidated statements of operations and retained earnings and of cash flows for the Company at and for each of the years ended March 31, 1999 (the "Last Fiscal Year-End"), March 31, 1998 and March 31, 1997 (the "Audited Statements") and the unaudited consolidated balance sheets and consolidated statements of operations and cash flows for the Company at and for the period ended September 30, 1999 (the "Interim Statements" and, together with the Audited Statements, the "Financial Statements") fairly present the consolidated financial condition of the Company at the dates indicated and the consolidated results of operations and cash flows of the Company for the periods indicated in accordance with GAAP consistently applied throughout the periods indicated (except as stated therein and, in the case of the Interim Statements, the omission of certain footnote disclosures and subject to normal year-end adjustments).
- (b) Certain Indebtedness. Schedule 2.6(b) sets forth all obligations of the Company and its Subsidiaries with respect to borrowed money, debt securities, capitalized leases and the deferred payment of the purchase price of property or services over an original term of 6 months or more, and the Property of the Company, if any, subject to a Lien to secure any of such obligations.
- (c) Absence of Certain Liabilities. Neither the Company nor any Subsidiary has any liability or obligation of any nature, whether absolute, accrued, contingent or otherwise, arising out of acts or omissions heretofore occurring, or circumstances currently or heretofore existing, except: (i) as expressly set forth in this Agreement (including without limitation disclosures in the Schedules hereto); (ii) as accrued in the balance sheet included in the Interim Statements (the "Interim Balance Sheet"); (iii) for liabilities and obligations incurred since the date of the Interim Balance Sheet in the ordinary course of business consistent in nature and amount with past practice; and (iv) liabilities and obligations of a kind not required to be accrued in a balance sheet at the date hereof prepared in accordance with GAAP which individually (or in the aggregate for related events, transactions, defects or circumstances) will not subject the Company or any Subsidiary to obligations in excess of \$25,000.
- (d) Absence of Certain Changes. Since the date of the Interim Balance Sheet, except as set forth in Schedule 2.6(d):
 - (i) The Company and each Subsidiary has operated their consolidated business in the ordinary course.
 - (ii) There has been no change or changes which, individually or in the aggregate, has or have had or is or are reasonably likely to have a Material Adverse Effect.
 - (iii) There has not been any damage, destruction or condemnation known to the Company with respect to Property having an aggregate net book value on the Company's consolidated books in excess of \$25,000, net of any insurance recoveries.

- (iv) There has not been any material change in the accounting methods, practices or principles of the Company.
- (v) Neither the Company nor any Subsidiary has sold, transferred or otherwise disposed of (or agreed or committed to sell, transfer or otherwise dispose of) any Property other than the sale of inventory in the ordinary course, or canceled, compromised, released or assigned any debt or claim in its favor, where the aggregate amount of such sales, transfers, dispositions, cancellations, compromises, releases or assignments exceeds \$25,000.
- (vi) Neither the Company nor any Subsidiary has instituted, settled or agreed to settle any litigation, action or proceeding before any Governmental Agency.
- (vii) Neither the Company nor any Subsidiary has assumed, guaranteed, endorsed or otherwise become responsible (or otherwise agreed to become responsible) for the obligations of any other Person, except for the endorsement of negotiable instruments in the ordinary course of business.
- (viii) Neither the Company nor any Subsidiary has granted (or agreed or committed to grant) any increase in compensation or fringe benefits other than normal salary increases consistent with prior periods.
- (ix) Neither the Company nor any Subsidiary has entered into any licensing or other Contract with regard to the acquisition or disposition of any material Intellectual Property other than non-exclusive licenses granted in the ordinary course of business consistent with past practice.

2.7 Taxes. Except as set forth in Schedule 2.7:

- (a) The Company and each Subsidiary has properly completed and filed, within the time and in the manner prescribed by law, all Tax returns and other documents required to be filed in respect of all Taxes, and all such returns and other documents are true, correct and complete in all material respects. The Company has furnished to the Buyer copies of all income Tax returns of the Company for the past 3 years. The Company and each Subsidiary has, within the time and in the manner prescribed by law, paid all Taxes that are due and payable. The Company has established reserves on its consolidated books that are at least equal to those required by GAAP.
 - (b) (i) None of such returns contained a disclosure statement under Section 6662 of the Code or any similar provision of foreign law;
 - (ii) The Company has not received written notice from any federal or foreign taxing authority asserting any deficiency against the Company or any Subsidiary or claim for additional Taxes in connection therewith, other than any deficiency or claim which has been previously settled or for which appropriate reserves are included in the Interim Statements;
 - (iii) There is no pending action, audit, proceeding or investigation with respect to the assessment or collection of federal or foreign Taxes or a claim for refund made by the Company or any Subsidiary with respect to federal or foreign Taxes previously paid;

- (iv) All amounts that are required to be collected or withheld by the Company and each Subsidiary with respect to federal or foreign Taxes have been duly collected or withheld, and all such amounts that are required to be remitted to any federal or foreign taxing authority have been duly remitted;
- (v) No audit has been conducted of any federal or foreign income tax return filed by the Company or any Subsidiary. The time during which such returns remain open for examination has expired in accordance with applicable statute and regulations, except for those returns for which the normally applicable statutory/regulatory period has not yet elapsed;
- (vi) Neither the Company nor any Subsidiary has requested nor been granted any currently effective waiver or extension of any statute of limitations with respect to the assessment or filing of any federal or foreign Tax or return with respect thereto;
- (vii) No consent has been filed under Section 341(f) of the Code with respect to the Company or any Subsidiary;
- (viii) The Company is not required to include in income any adjustment pursuant to Section 481(a) of the Code (or similar provisions of foreign laws or regulations) by reason of a change in accounting method nor does the Company have any knowledge that the Internal Revenue Service (or other federal or foreign taxing authority) has proposed, or is considering, any such change in accounting method; and
- (ix) Neither the Company nor any Subsidiary is a party to or bound by nor has any continuing obligation under any tax sharing or similar agreement or arrangement with any Person.

2.8 Title to Properties.

- (a) Neither the Company nor any Subsidiary owns any real Property.
- (b) Schedule 2.8(b) is a true and complete summary based on the books and records of the Company of all items of personal Property owned by the Company or any Subsidiary with a consolidated net book value at the Last Fiscal Year-End in excess of \$5,000 per item.
- (c) Except as set forth in Schedule 2.8(c), the Company and each Subsidiary has good title to all tangible personal Properties, in each case free and clear of all Third-Party Rights.
- (d) The Company and its Subsidiaries, taken together, own all material items of non-inventory tangible and intangible personal Property that were owned as of the Last Fiscal Year-End and used in generating the revenue shown in the audited consolidated statement of operations of the Company for the fiscal year ending on the Last Fiscal Year-End, subject to any sales or dispositions of tangible personal Property since the Last Fiscal Year-End in the ordinary course of business.
- 2.9 Inventories. Except as set forth in Schedule 2.9, since the Last Fiscal Year-End, all sales of inventory by the Company and its Subsidiaries have been made in the ordinary course of business and no inventory has been pledged as collateral.

- 2.10 Accounts Receivable. The consolidated accounts receivable of the Company and its Subsidiaries (i) are bona fide and arose from valid sales in the ordinary course of business in material conformity with all applicable Legal Requirements, (ii) are valid and binding obligations of the debtors requiring no further performance by the Company or any Subsidiary, and (iii) subject to the allowance for doubtful accounts receivable in the Interim Balance Sheet, are fully collectible and not subject to any offsets or counterclaims and do not represent guaranteed sale, sell-or-return transactions or any other similar understanding. Except as shown on Schedule 2.6(b), no accounts receivable have been pledged as collateral to any Person. The amounts shown for accounts receivable in the Financial Statements reflect an allowance for doubtful accounts receivable in accordance with GAAP.
- 2.11 Leases. Schedule 2.11 lists all leases, rental agreements, conditional sales contracts and other similar Contracts under which the Company or any Subsidiary leases (as lessor or lessee) any real or personal Property with rental payments exceeding \$10,000 per year (collectively, the "Disclosable Leases"). All Disclosable Leases are, in all material respects, valid and enforceable by the Company in accordance with their terms. Neither the Company nor any Subsidiary nor, to the knowledge of the Company, any other party to any Disclosable Lease is in material breach thereof. The Company and each Subsidiary enjoys peaceable possession of all real estate premises subject to Disclosable Leases to which it is a party and to all personal Property subject to Disclosable Leases to which it is a party.
- 2.12 Facilities, Equipment. The Company owns or leases all material land, buildings and equipment used in the operation of its business. The Company has not received any notice of any material violation of any Legal Requirement or Order relating to the Company's facilities which has not been corrected, and no facility of the Company is in material violation of any Legal Requirement or Order.
- 2.13 Insurance. Schedule 2.13 lists and describes briefly all binders and policies of liability, theft, life, fire and other forms of insurance and surety bonds, insuring the Company or any Subsidiary or their respective Properties, assets and business as of the date hereof. Except as noted in Schedule 2.13, all listed policies and binders insure on an occurrence, rather than claims-made, basis. All policies and binders listed in Schedule 2.13 are valid and in good standing and in full force and effect and the premiums have been paid when due. Except for any claims set forth in Schedule 2.13, there are no outstanding unpaid claims under such policy or binder, and, except as set forth in Schedule 2.13, neither the Company nor any Subsidiary has received any notice of cancellation, general disclaimer of liability or non-renewal of any such policy or binder.

2.14 Employment and Benefit Matters.

- (a) Schedule 2.14(a) lists each of the following for each employee of the Company and each Subsidiary: name, hire date and current salary. None of the employees listed on Schedule 2.14(a) has given the Company or such Subsidiary notice of his or her intention to resign his or her position with the Company or such Subsidiary and neither the Company nor such Subsidiary has any present intention to terminate such employees.
- (b) Schedule 2.14(b) lists all of the following items which are applicable to the Company or any Subsidiary: (i) employment Contracts with any employee, officer or director; and (ii) Contracts

or arrangements with any Person providing for bonuses, profit sharing payments, deferred compensation, stock options or stock purchase rights (other than options and warrants granted pursuant to those Stock Rights listed in Section 2.1(a)), retainer, consulting, incentive, severance pay or retirement benefits, life, medical or other insurance, payments triggered by a change in control or any other employee benefits or any other payments, "fringe benefits" or perquisites which are not terminable at will without liability to the Company or any Subsidiary or which are subject to ERISA. The contracts or arrangements referred to in the foregoing clause (ii) are herein called "Benefit Plans."

- (c) Neither the Company nor any of its ERISA Affiliates has any union contracts, collective bargaining, union or labor agreements or other Contract with any group of employees, labor union or employee representative(s), nor has the Company or any ERISA Affiliate ever participated in or contributed to any single employer defined benefit plan or multi-employer plan within the meaning of ERISA Section 3(37), nor is the Company currently engaged in any labor negotiations, excepting minor grievances, nor is the Company the subject of any union organization effort. The Company and each Subsidiary is in material compliance with applicable Legal Requirements respecting employment and employment practices and terms and conditions of employment, including without limitation health and safety and wages and hours. No complaint or other proceeding by or on behalf of any current or former employee or group of employees is pending against the Company or any Subsidiary before any Governmental Agency, and no claim by any current or former employee or group of employees that the Company or any Subsidiary is not in compliance with any Legal Requirement relating to employees or employment is pending against the Company or any Subsidiary. There is no labor dispute, strike, slowdown or work stoppage pending or threatened against the Company or any Subsidiary.
- (d) True and correct copies of each Benefit Plan listed in Schedule 2.14(b) that is subject to ERISA (a "Company ERISA Plan") and related trust agreements, insurance contracts, and summary descriptions have been delivered or made available to the Buyer by the Company. The Company has also delivered or made available to the Buyer a copy of the most recently filed IRS Form 5500, with attached financial statements and accountant's opinions, if applicable, for each Company ERISA Plan. The Company has also delivered or made available to the Buyer a copy of, in the case of each Company ERISA Plan intended to qualify under Section 401(a) of the Code, the most recent Internal Revenue Service letter as to its qualification under Section 401(a) of the Code. Nothing has occurred prior to or since the issuance of such letters to cause the loss of qualification under the Code of any of such plans.
- (e) With respect to each Company ERISA Plan, (i) there has been no "prohibited transaction," as such term is defined in Section 406 of ERISA and Section 4975 of the Code, (ii) each Company ERISA Plan has been administered in accordance with its terms and in material compliance with all Legal Requirements (including without limitation ERISA and the Code), compliance with all Legal Requirements (including without limitation ERISA and timely filed (iii) the Company (or, as appropriate, an ERISA Affiliate) has prepared in good faith and timely filed all requisite governmental reports in true and correct form and has properly and timely filed and distributed or posted all notices and reports to participants and beneficiaries required to be filed, distributed or posted, (iv) no suit, administrative proceeding, action or other litigation has been brought, or to the knowledge of the Company is threatened, against any Company ERISA Plan or against the Company with respect to any Company ERISA Plan, including without limitation any audit or inquiry by the Internal Revenue Service or United States Department of Labor, (v) the

Company and each ERISA Affiliate have performed all material obligations required to be performed by them under, and are not in any material respect in default under or in violation of, and have no knowledge of any material default or violation of, any Company ERISA Plan, (vi) neither the Company nor any ERISA Affiliate is subject to any liability or penalty under Sections 4976 through 4980 of the Code or Title I of ERISA, (vii) all contributions required to be made by the Company or any ERISA Affiliate have been made on or before their due dates, (viii) no "reportable event" within the meaning of Section 4043 of ERISA (excluding any such event for which the 30-day notice requirement has been waived under the regulations to Section 4043 of ERISA) nor any event described in Section 4062, 4063 or 4041 of ERISA has occurred, (ix) no Company ERISA Plan is covered by, and neither the Company nor any ERISA Affiliate has incurred or expects to incur any material liability under, Title IV of ERISA or Section 412 of the Code, and (x) neither the Company nor any ERISA Affiliate is a party to, or has made any contribution to or otherwise incurred any obligation under, any "multi-employer plan" as defined in Section 3(37) of ERISA.

- (f) The Company has complied with (i) the applicable health care continuation and notice provisions of the Consolidated Omnibus Budget Reconciliation Act of 1985 ("COBRA") and the proposed regulations thereunder, (ii) the applicable requirements of the Family and Medical Leave Act of 1993 and the regulations thereunder, and (iii) the applicable requirements of the Health Insurance Portability and Accountability Act of 1996 and the temporary regulations thereunder. The Company has no material obligations under COBRA with respect to any former employees or qualifying beneficiaries thereunder.
- (g) There has been no amendment to, written interpretation or announcement (whether or not written) by the Company or other ERISA Affiliate relating to, or change in participation or coverage under, any Benefit Plan which would materially increase the expense of maintaining such Plan above the level of expense incurred with respect to that Plan for the most recent fiscal year included in the Audited Statements.
- (h) Schedule 2.14(h) contains a true and correct list of each employee, director or consultant who holds any stock option as of the date hereof, together with (i) the number of shares of Company Common Stock subject thereto, (ii) the date of grant, (iii) the extent to which such stock option is currently vested and, to the extent such stock option is not fully vested, the vesting schedule, (iv) the exercise price, (v) whether such stock option is intended to qualify as an incentive stock option within the meaning of Section 422(b) of the Code (an "ISO"), and (vi) the expiration date of such stock option. Schedule 2.14(h) also sets forth the aggregate number of ISO's and nonqualified stock options outstanding as of the date hereof.
- (i) Neither the Company nor any Subsidiary is a party to any Contract or plan, including, without limitation, any stock option plan, stock appreciation right plan or stock purchase plan, as to which any benefits will be increased, or the vesting of benefits will be accelerated, by the occurrence of any of the transactions contemplated by this Agreement or the value of any benefits will be calculated on the basis of any of the transactions contemplated by this Agreement.
- (j) Except as disclosed in Schedule 2.14(j), the Company and its ERISA Affiliates do not maintain any plans providing benefits within the meaning of Section 3(1) of ERISA (other than group health plan continuation coverage under Section 601 of ERISA and 4980B(f) of the Code) to former employees or retirees.

- (k) All Benefit Plans are in compliance with, and satisfy the requirements of, the Family and Medical Leave Act of 1993.
- 2.15 Contracts. Except as shown on Schedules 2.11 and 2.15, and except for Contracts fully performed or terminable at will without liability to the Company, neither the Company nor any Subsidiary is a party to any Contract which contemplates performance by the Company or such Subsidiary during a remaining period of more than 180 days or involves remaining commitments for sale or purchase in excess of \$25,000. True and complete copies of each Contract disclosable on Schedule 2.15 (a "Disclosable Contract") have been delivered to the Buyer. Each Disclosable Contract is, in all material respects, valid and enforceable by the Company in accordance with its terms. Neither the Company nor any Subsidiary nor, to the knowledge of the Company, any other party to any Disclosable Contract is in material breach thereof.

2.16 Officers and Directors. Schedule 2.16 is a true and complete list of:

- (a) the names and addresses of each of the Company's and each Subsidiaries' officers and directors;
- (b) the name of each bank or other financial institution in which the Company or any Subsidiary has an account, deposit or safe deposit box and the names of all persons authorized to draw thereon or to have access thereto; and
- (c) the name of each bank or other financial institution in which the Company or any Subsidiary has a line of credit or other loan facility.
- 2.17 Corporate Documents. The Company has furnished or made available to the Buyer or its representatives true, correct and complete copies of (i) the articles or certificate of incorporation and by-laws of the Company and each Subsidiary, (ii) the minute books of the Company and each Subsidiary containing all records required to be set forth of all proceedings, consents, actions and meetings of the stockholders and board of directors of the Company or such Subsidiary; and (iii) all material Permits and Orders with respect to the Company and any Subsidiary.

2.18 Legal Proceedings.

- (a) There is no action, suit, proceeding or investigation pending in any court or before any arbitrator or before or by any Governmental Agency against the Company or any Subsidiary or any of their respective Properties or businesses, and to the knowledge of the Company, there is no such action, suit, proceeding or investigation threatened.
- (b) Neither the Company nor any Subsidiary has ever been notified in writing that it has been subject to an audit, compliance review, investigation or like contract review by the U.S. General Services Administration or any other Governmental Entity or agent thereof in connection with any government contract (a "Government Audit"). To the Company's knowledge, no Government Audit is threatened and no basis exists for a finding of noncompliance with any material provision of any government contract or for a material refund of any amounts paid or owed to the Company or any Subsidiary by any Governmental Entity pursuant to such government contract.

10

PATENT REEL: 011820 FRAME: 0312

- 2.19 Compliance with Instruments, Orders and Legal Requirements. Neither the Company nor any Subsidiary is in material violation of, or in default in any material respect with respect to, any term or provision of its articles or certificate of incorporation or bylaws, or, to the knowledge of the Company, any Order or any Legal Requirement applicable to the Company or such Subsidiary.
- 2.20 Permits. The Company and each Subsidiary holds all Permits material to the conduct their consolidated business as and where now conducted. To the knowledge of the Company, there is not pending nor threatened any proceedings to terminate, revoke, limit or impair any material Permit.

2.21 Intellectual Property.

- (a) "Company Intellectual Property" means Intellectual Property used in the business of the Company and its Subsidiaries as currently conducted or as presently planned to be conducted, including without limitation, the development, licensing and use of the programs and model libraries currently being marketed (the "Programs"), other than Third-Party Intellectual Property. Except as set forth on Schedule 2.21(a) and except for Third-Party Intellectual Property licensed to the Company or a Subsidiary pursuant to an agreement listed in Schedule 2.21(c)(ii), the Company or a Subsidiary owns, solely and exclusively, and free and clear of any Third-Party Right, all title to and rights in all Intellectual Property that is used in the business of the Company and its Subsidiaries as currently conducted or as presently planned to be conducted, including without limitation, the development, licensing and use of the Programs. "Intellectual Property" means patents, patent rights, trademarks, trademark rights, trade names, trade name rights, service marks, copyright registrations, copyrights, and any applications for any of the foregoing, net lists, schematics, industrial models, inventions, technology, know-how, trade secrets, computer software programs or applications (in both source code and object code form), development documentation, programming tools, data, technical information, and tangible or intangible proprietary information or material. "Third-Party Intellectual Property" means Intellectual Property owned in whole or in part by any Person other than the Company or a Subsidiary.
- (b) Schedule 2.21(b) lists all patents, patent applications, trademarks, trade names, service marks and copyrights included in the Company Intellectual Property which have been registered, issued or applied for and the jurisdictions in which such Company Intellectual Property right has been issued, registered or applied for.
- (c) Schedule 2.21(c)(i) lists those licenses, sublicenses and other agreements, written or unwritten, to which the Company or any Subsidiary is a party pursuant to which any Person is authorized to use, resell, sublicense or market or distribute any product currently marketed or presently planned to be marketed by the Company or a Subsidiary or any component or predecessor of any such product and which have accounted for gross revenue in excess of \$500,000 in the aggregate during the three (3) years ended September 30, 1999. Schedule 2.21(c)(ii) lists all written, and all material unwritten, licenses, sublicenses and other agreements to which the Company or any Subsidiary is a party and pursuant to which the Company or any Subsidiary is authorized to use, resell or distribute any Third-Party Intellectual Property, including without limitation software, opensource, freeware, shareware and hardware, which are incorporated in or are a part of any products which the Company or any Subsidiary has sold, resold, licensed or sublicensed, or which is

PATENT REEL: 011820 FRAME: 0313 material to the current operations of the Company and its Subsidiaries, other than (in the case of Third-Party Intellectual Property used internally only and not embedded in products shipped to customers) readily-obtainable standard products. The Company and each Subsidiary has, and at the relevant times in the past had, all necessary rights to resell or distribute any hardware and software of a third party which it resells or distributes or has resold or distributed. Neither the Company nor any Subsidiary is in material violation of any license, sublicense or agreement described in Schedule 2.21(c)(i) or (ii). To the knowledge of the Company, neither the Company nor any Subsidiary, or any of the products or operations of either, is in material violation of or materially infringes any Third-Party Intellectual Property. Except as set forth on Schedule 2.21(c)(i) or (ii), neither the Company nor any Subsidiary has received any claim that it has lost or will lose any rights of the Company or any Subsidiary under any licenses to Third-Party Intellectual Property to which the Company or such Subsidiary is a party. The execution and delivery of this Agreement by the Company and, except as set forth in Schedule 2.21(c)(ii), the consummation of the transactions contemplated hereby will neither cause the Company or any Subsidiary to be in violation or default under any such license, sublicense or agreement nor entitle any other party to any such license, sublicense or agreement to terminate or modify such license, sublicense or agreement. Except as listed on Schedule 2.21(c)(i) and for the Company's commercially available products and preproduction and beta products licensed in the ordinary course of business, neither the Company nor any Subsidiary has assigned or licensed to any third party any right, title or interest in the Company Intellectual Property. Except as listed on Schedule 2.21(c)(ii), neither the Company nor any Subsidiary is contractually obligated to pay any compensation to any third party for the use of the Company Intellectual Property or the Third-Party Intellectual Property.

- (d) To the Company's knowledge there is no material unauthorized use, disclosure, infringement or misappropriation of any Company Intellectual Property or any Third-Party Intellectual Property licensed by or through the Company by any third party, including without limitation any employee or former employee of the Company or any Subsidiary. Neither the Company nor any Subsidiary has entered into any agreement to indemnify any other person against any charge of infringement of any Third-Party Intellectual Property, other than indemnification provisions contained in purchase orders or license agreements arising in the ordinary course of business.
- (e) To the Company's knowledge, all patents, registered trademarks, registered service marks and registered copyrights held by the Company and its Subsidiaries are valid and subsisting. To the Company's knowledge, there is no assertion or claim (or basis therefor) challenging the validity of any Company Intellectual Property. Neither the Company nor any Subsidiary has been sued in any suit, action or proceeding, or otherwise notified of any claim, which involves a claim of infringement of any patent, trademark, service mark, copyright or violation of any trade secret or other proprietary right of any third party. Neither the conduct of the business of the Company and its Subsidiaries as currently conducted nor the manufacture, sale, licensing or use of any of the products of the Company and its Subsidiaries as now manufactured, sold or licensed or used, infringes on or conflicts with, in any way, any trademark, trademark right, trade name, trade name right, service mark or copyright, or, to the Company's knowledge, any patent, patent right, industrial model or invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention, of any third party that individually or in the aggregate has or is reasonably likely to have a invention of any third party that individually or in the aggregate has or is reasonably likely to have a invention of any third party that individually or in the aggregate has or is reasonably likely to have a invention of any

has brought any action, suit or proceeding for infringement of Company Intellectual Property or breach of any license or agreement involving Company Intellectual Property against any third party. There are no pending, or to the Company's knowledge, threatened interference, re-examinations, oppositions or nullities involving any patents, patent rights or applications therefor of the Company or any Subsidiary. Except as set forth on Schedule 2.21(e), to the Company's knowledge, there is no breach or violation by a third party of, or actual or threatened, loss of rights under, any licenses to which the Company is a party.

- (f) Except as set forth in Schedule 2.21(f), the Company or a Subsidiary has secured written assignments from all current and former consultants and employees who contributed to the creation or development of the Company Intellectual Property currently being provided or marketed, or presently planned to be provided or marketed, to customers or currently being used by the Company or a Subsidiary of the rights to such contributions that the Company or such Subsidiary does not already own by operation of law, recognizing the Company's or Subsidiary's ownership of all such Company Intellectual Property and agreeing to hold such of it as is not protected by patents, patent applications or copyright ("Confidential Information") in confidence and not to use any Confidential Information except in connection with such consultant's or employee's work for the Company or a Subsidiary. No current or former consultant or employee has claimed any rights to, interest in or rights to compensation for use of any of the Company Intellectual Property currently marketed or currently proposed to be marketed, by the Company or any Subsidiary.
- (g) The Company and each of its Subsidiaries has taken all commercially reasonable steps to protect and preserve the confidentiality of all Confidential Information. Except as set forth on Schedule 2.21(g)(i), all use, disclosure or appropriation of Confidential Information by or to a third party has been pursuant to the terms of a written confidentiality or nondisclosure agreement between the Company or any Subsidiary and such third party. Schedule 2.21(g)(ii) contains copies of the Company's standard forms of confidentiality and nondisclosure agreement. Except as disclosed in Schedule 2.21(g)(iii) (which lists those agreements containing material variations from the standard forms of confidentiality and nondisclosure agreement), as of the date hereof, the Company has not entered into any confidentiality and nondisclosure agreement which contains terms materially different than as set forth in the standard forms of such agreements contained in Schedule 2.21(g)(ii).
- (h) Schedule 2.21(h) lists (including names, addresses, contact names, telephone numbers and termination date and next renewal date) all Contracts or other arrangements currently in effect pursuant to which the Company or any Subsidiary is obligated to provide installation, maintenance or other support services (collectively, the "Support Agreements"). Except for any nonstandard maintenance agreements specified as such in Schedule 2.21(h), all of the Support Agreements are in all material respects in the form of the license agreement identified as the standard maintenance agreement set forth in Schedule 2.21(h). The versions of the products currently supported by the Company are set forth in Schedule 2.21(h).
- (i) There are no material defects in the Company's products, and there are no material errors in any documentation, specifications, manuals, user guides, promotional material, internal notes and memos, technical documentation, drawings, flow charts, diagrams, source language statements, demo disks, benchmark test results, and other written materials related to, associated with or used or produced in the development of the Company's or its Subsidiaries' products, which defects or errors have or are reasonably likely to have, individually or in the aggregate, a Material Adverse Effect.

To the Company's knowledge, the current versions of the Programs, including, without limitation, any time-and-date-related codes, data entry features and internal subroutines thereof, (a) automatically accommodate the change in the date from December 31, 1999 to January 1, 2000 without negatively affecting the Programs' performance; and (b) accurately accept, reflect and calculate all dates that are relevant to the Programs' performance, when used with products that properly exchange date data with the Programs.

- (j) Schedule 2.21(j) lists, as of the date hereof, all written agreements or other arrangements under which the Company or any Subsidiary has provided or agreed to provide source code of any Program or any other Company Intellectual Property to any third Person, whether pursuant to escrow arrangements or otherwise.
- (k) The Company has made available to the Buyer copies of the Company's standard forms of end-user license. Except as disclosed in Schedule 2.21(k), as of the date hereof, the Company has not entered into any end-user license in the last three (3) years involving payments to the Company in excess of \$25,000 which contains terms materially different than as set forth in the standard forms of such agreements made available to the Buyer.
- (1) No government funding or university or college facilities were used in the development of any Program and no Program was developed pursuant to any contract or other agreement with any person or entity except pursuant to contracts or agreements listed in Schedule 2.21(1).
- (m) The Company has no customer warranty obligations currently if effect other than as set forth in end-user agreements listed in Schedule 2.21(k). The Company categorizes quality assurance incidents reported by its Subsidiaries or its Subsidiaries' customers into the following categories: Priority 1 an inherently essential product feature or application is not available or functioning; Priority 2 problem adversely affects completion of task, no workaround solution exists; Priority 3 final output of a process is correct but other errors exist; and Priority 4 errors which do not affect the accuracy of the delivered product or that are cosmetic. Schedule 2.21(m) lists all currently pending Priority 1 and 2 items related to products currently or previously marketed by the Company or a Subsidiary and the nature thereof that are pending or were made within the past twelve months. Except as set forth in Schedule 2.21(m), to the knowledge of the Company, the Company has not made any material oral or written representations or warranties with respect to its products or services.
- (n) Except as set forth in Schedule 2.21(n), neither any single customer of the Company which individually accounted for more than five percent (5%) or over \$1 million of the Company's gross revenue during the thirty-six (36) months ended September 30, 1999 nor any material supplier of the Company has cancelled or threatened to cancel its relationship with the Company.
- 2.22 Capital Expenditures. Schedule 2.22 sets forth, by nature and amount, all budgeted capital expenditures of the Company and its Subsidiaries for which commitments have been or are budgeted to be made, or for which payments or current liabilities have been made or incurred or are budgeted to be made or incurred, after the Last Fiscal Year-End in excess of \$5,000.
- 2.23 Environmental Matters. There are no Hazardous Materials used or present at any location used by the Company or a Subsidiary or any predecessor entity of either in the conduct of

the Business, except for any Hazardous Materials constituting normal office supplies. To the knowledge of the Company, no location currently or previously used by the Company or a Subsidiary or any predecessor entity of either is contaminated by any Hazardous Material or was previously used for any purpose other than office space. There are no environmental materials or conditions, including on-site or off-site disposal or releases of Hazardous Materials that could reasonably be expected to have a Material Adverse Effect. To the knowledge of the Company, no event has occurred and no activity has been or is being conducted by the Company, a Subsidiary or any other Person which has resulted or could reasonably result in contamination of any location currently or previously used by the Company or a Subsidiary or any predecessor entity of either by any Hazardous Material. Neither the Company, any Subsidiary nor any predecessor entity of either has received any written communication from any Governmental Entity alleging that the Company, Subsidiary or predecessor entity or any premises currently or previously occupied by any of such Persons is contaminated by any Hazardous Materials or in violation of any Environmental Requirement. To the knowledge of the Company, no Government Agency has commenced any investigation or proceeding with respect to the contamination of any location currently or previously used by the Company or a Subsidiary or any predecessor entity of either by any Hazardous Material.

- 2.24 Illegal Payments. To the best knowledge the Company, none of the Company, any Subsidiary or any director, officer, employee, or agent of the Company or any Subsidiary has, directly or indirectly, paid or delivered any fee, commission, or other sum of money or item of property however characterized to any broker, finder, agent, government official, or other person, in the United States or any other country, in any manner related to the business or operations of the Company or any Subsidiary, which the Company, any Subsidiary or any such director, officer, employee, or agent knows or has reason to believe to have been illegal under any law.
- dates (except as thereafter amended) all documents that the Company has filed with the SEC (the "Company SEC Documents") have complied in all material respects with the applicable requirements of the Act or the Exchange Act, as the case may be, and none of the Company SEC Documents has contained any untrue statement of a material fact or omitted to state a material fact required to be stated therein or necessary in order to make the statements made therein, in light of the circumstances under which they were made, not misleading except to the extent corrected by a subsequently filed Company SEC Document filed prior to the date hereof.

2.26 Board of Directors Approval; Fairness Opinion.

(a) The members of the Board of Directors of the Company present at the meeting called to discuss the Merger have unanimously approved this Agreement and the Plan of Merger and unanimously recommended this Agreement and the Plan of Merger to the Company's stockholders. Such approval and recommendation have not been modified or withdrawn and are in full force and effect on the date hereof. The Company's financial adviser, Broadview Int'l LLC, has rendered its opinion to the Board of Directors of the Company that the consideration to be received in the Merger is fair to the Company's stockholders from a financial point of view. Prior to the Company's entry into this Agreement, each of the Company's officers and directors has entered into a voting agreement with the Buyer to vote all Shares beneficially owned by such officer or director in favor of the Merger.

- (b) The transactions contemplated by this Agreement (i) are not subject to the "fair price," "moratorium," "control share acquisition" or other similar statute (a "Takeover Statute") of any jurisdiction other than the State of Oregon. The Oregon Control Share Act is not applicable to the Merger by virtue of ORS 60.801(4)(e)(F), which exempts a merger effected in compliance with ORS 60.481 to 60.501. The Oregon Business Combination statute does not apply to the Merger because the Merger has been approved by the Company's Board of Directors. There are no other Takeover Statutes of the State of Oregon applicable to the Company or any Subsidiary.
- 2.27 No Dissenters' Rights. No stockholder of the Company is entitled to exercise or obtain any form of dissenters' rights under the Oregon Business Corporation Act ("OBCA").
- 2.28 Option Substitution. No Person holding an option issued under the Company's Amended and Restated 1993 Stock Incentive Plan may, pursuant to the terms of such plan, object to the substitution of such options with options of the Buyer as set forth in Section 3.2 of the Plan of Merger.
- 2.29 Employee Stock Purchase Plan. The Company has taken all necessary action to provide that (i) the number of Company common shares available for issuance under the Company's 1996 Employee Stock Purchase Plan, as amended (the "Stock Purchase Plan"), has been increased from 300,000 to 600,000, (ii) the current offering period under the Stock Purchase Plan shall be terminated as of the earliest date after the required notice has been provided to each participant in the Stock Purchase Plan (the "Plan Termination Date"), (iii) each participant in the Stock Purchase Plan on the date hereof shall be deemed to have exercised his or her Purchase Right (as defined in the Stock Purchase Plan) on the Plan Termination Date and shall acquire from the Company (a) such number of whole shares of Company Common Stock as his or her accumulated payroll deductions on the Plan Termination Date will purchase at the Offering Exercise Price (as defined in the Stock Purchase Plan)(treating the last business day prior to the Plan Termination Date as the "Purchase Date" for all purposes of the Stock Purchase Plan) and (b) cash in the amount of any remaining balance in such participant's account, and (iv) the Stock Purchase Plan shall be terminated effective as of the Plan Termination Date.
- 2.30 Transfers of Technology. The Company has not licensed or otherwise transfered any versions of the Verias Simulator to any foreign firms or institutions, other than those transfers made in executable code form, all of which will be disclosed to DARPA.
- 2.31 Representations. No representation or warranty by the Company in this Agreement (including without limitation the Schedules and Exhibits attached hereto), or in any document furnished by the Company at the Closing pursuant hereto contains any untrue statement of a material fact or omits to state a fact necessary to make the statements contained in such representation or warranty not misleading.

Section 3 Representations and Warranties of Buyer

The Buyer hereby represents and warrants to the Company that, on and as of the date hereof:

3.1 Organization, Standing of Buyer and Buyer Subsidiary. The Buyer is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Buyer Subsidiary is a corporation duly organized, validly existing and in good standing

under the laws of the State of Oregon. The Buyer and Buyer Subsidiary have full power and authority under applicable corporate law to own, lease and operate their Properties and to carry on the business in which they are engaged.

- Authority; Enforceability. The Buyer and Buyer Subsidiary have all necessary power and authority under applicable corporate law to execute, deliver and perform their obligations under this Agreement. The execution, delivery and performance of this Agreement by the Buyer and Buyer Subsidiary has been duly authorized by all necessary action under applicable corporate law. This Agreement constitutes a legal, valid and binding obligation of the Buyer and Buyer Subsidiary, enforceable in accordance with its terms, subject to general limitations on the availability of equitable remedies and the effect of bankruptcy, insolvency, reorganization and other laws of general application affecting the enforcement of creditors' rights. The execution, delivery and performance of this Agreement by the Buyer and Buyer Subsidiary and the consummation by the Buyer and Buyer Subsidiary of all of the transactions contemplated hereby, (x) do not require any Third-Party Action relating to the Buyer or Subsidiary except those listed on Schedule 3.2, (y) do not violate any Legal Requirement or Order applicable to the Buyer or Buyer Subsidiary and (z) do not conflict with or constitute a default (with or without the giving of notice or the passage of time or both) under, or result in any acceleration or right of acceleration of any obligations under, any Contract to which the Buyer or Buyer Subsidiary is a party, where, in each case, the absence of such Third-Party Action or such violation, conflict, default or acceleration would in any way adversely affect the transactions contemplated hereby.
- 3.3 Litigation. There are no claims, actions, suits or other proceedings pending, or to the knowledge of the Buyer, threatened, at law or in equity, by or before any Governmental Agency or any arbitrator against the Buyer which could reasonably be expected to have an adverse effect on the ability of the Buyer to perform its obligations under this Agreement.
- 3.4 Proxy Materials. All of the information to be furnished by the Buyer or Buyer Subsidiary for inclusion in the Proxy Statement will not, on the date it is first mailed to the Company's stockholders, and on the date of the Company's stockholders' meeting, contain any statement which is false or misleading with respect to any material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading.

Section 4 Conditions to Obligations of Buyer and Buyer Subsidiary at Closing

The obligations of the Buyer and Buyer Subsidiary hereunder to be performed at the Closing are subject to the satisfaction at or prior to the Closing of the following conditions, except for any condition the Buyer may waive in writing in accordance with Section 7.3.

4.1 Representations and Warranties. The representations and warranties contained in Section 2 shall have been true in all material respects on the date of this Agreement and shall be true in all material respects at and as of immediately prior to the Closing with the same effect as though made at and as of immediately prior to the Closing. The updating of Schedules to disclose nonmaterial changes in the underlying matters disclosed therein occurring from the date hereof to the Effective Time shall not constitute a failure of the condition set forth in this Section 4.1.

- 4.2 Proxy Statement. The Proxy Statement shall comply as to form in all material respects with the applicable requirements of the Exchange Act and the rules and regulations promulgated thereunder, and shall not, at the time of (i) first mailing thereof or (ii) the stockholders' meeting to be held in connection with the Merger, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, except that this condition shall not apply with respect to information supplied by the Buyer or any affiliates or representatives of the Buyer for inclusion in the Proxy Statement.
- 4.3 Closing Certificate. The Company shall have delivered to the Buyer its certificate dated the date of the Closing that the conditions specified in Sections 4.1, 4.2 and 4.4 are satisfied. Such certificate shall be deemed a representation and warranty of the Company under Section 2 for all purposes of this Agreement.
- 4.4 Performance. The Company shall have performed and complied in all material respects with all covenants required herein to be performed or complied with by it on or before the Closing.
 - 4.5 Stockholder Approval. The Stockholder Approval shall have been given.
- 4.6 Third-Party Action. All Third-Party Action required under the Designated Contracts in order to consummate the Closing on the terms hereof shall have been taken, and all Third-Party Action required (other than under the Designated Contracts) in order to consummate the Closing on the terms hereof, other than any such Third-Party Actions which in the aggregate would not have a Material Adverse Effect, shall have been taken, and the Hart-Scott waiting period shall have expired.
- 4.7 Opinion of Counsel. The Buyer shall have received from Ater Wynne LLP, counsel to the Company, an opinion dated the date of the Closing, in form and substance substantially as set forth in Exhibit B.
- 4.8 Transactional Litigation. No action, suit or proceeding before any Governmental Agency shall have been commenced and not dismissed, and no investigation by any Governmental Agency shall have been commenced or overtly threatened, against the Company, the Buyer, Subsidiary, or any of their respective principals, officers, directors or shareholders seeking to restrain, prevent or change the transactions contemplated hereby or questioning the validity or legality of any of such transactions or seeking damages in connection with any of such transactions.
- 4.9 Interim Events. None of the events listed in Sections 6.8(a) through (v) shall have occurred without the Buyer's written consent.
- 4.10 Management Changes, Technical Employees. No change in the executive management of the Company as listed on Schedule 4.10(a) shall have occurred from the date hereof, and not less than 80% of the Company's employees listed on Schedule 4.10(b) shall have indicated to the Buyer, in form reasonably satisfactory to the Buyer, their intention to continue their employment with the Company or the Buyer on substantially their current terms following the Merger.

- 4.11 Employment and Noncompetition Agreements. The individuals designated in Schedule 4.11 shall have entered into (i) employment agreements with the Buyer in form satisfactory to the Buyer, in replacement of the employment agreements, if any, currently in effect between each of them and the Company and (ii) proprietary information and noncompetition agreements with the Buyer substantially in the form attached as Exhibit C.
- 4.12 Option Holder Approval. At least 75% of the Company employees who hold options to acquire Shares as of the date hereof under the Company's 1986 Stock Option Plan shall have agreed to accept options to acquire shares of the Buyer's common stock (issued under the Buyer's 1995 Stock Option / Stock Issuance Plan) in substitution for such Company stock options held by such employees as of the date hereof.
- 4.13 Nonemployee Option Consents. The Company shall have obtained the consent (i) of those option holders, if any, described in Section 3.1 of the Plan of Merger to the conversion of such options into the right to receive the Spread Amount (as defined in the Plan of Merger) and (ii) of all option holders under the 1995 Stock Option Plan for Nonemployee Directors to the cancellation of all options under such plan.
- 4.14 Financial Statement Compliance. The Buyer shall have received confirmation from its outside auditors, KPMG Peat Marwick LLP, reasonably satisfactory to the Buyer, that the Financial Statements are satisfactory in form for inclusion in the Buyer's SEC filings, both for reporting the Merger and on a going forward basis.
- 4.15 Corporate and Other Proceedings. All corporate and other proceedings on the part of the Company in connection with the transactions to be consummated at the Closing, and all documents and instruments incident to such transactions, shall be reasonably satisfactory in substance and form to the Buyer.
- 4.16 Cancellation of Company 401(k) Plan. The Company shall have canceled its 401(k) Plan effective as of prior to the Closing. The Company has provided the Buyer with a copy of the most recent determination letter issued by the Internal Revenue Service with regard to the Plan. The parties acknowledge and agree that the Company will cancel its 401(k) Plan as of December 31, 1999 and pay any matching funds due to the accounts of the 401(k) Plan participants (in accordance with the amounts accrued in the Financial Statements) at that time. The Company's 401(k) Plan shall remain in effect to the extent necessary to permit the rollover of all funds held by the Plan's participants.
- 4.17 Officer and Director Resignations. All officers and directors of the Company shall have submitted their resignations in writing, effective as of the Effective Time.
- 4.18 Advisory Opinion. The Company shall have received a written advisory opinion from the United States Department of Commerce, in form satisfactory to the Buyer in its reasonable discretion, stating that the Saber Simulator and the Verias Simulator, and all models provided in the Company's model library, have an ECCN of 3D991. As a result of such ECCN classification, the Company did not ship either simulator product into any jurisdiction where an export license was required.

4.19 DARPA Letter. The Company shall have received a written letter of assurance from the United States Defense Advanced Research Projects Agency ("DARPA"), in form satisfactory to the Buyer in its reasonable discretion, stating that (i) any past transfers to foreign firms or institutions by the Company of any versions of the Verias Simulator in executable code format did not constitute a breach of the terms of the DARPA Agreement and based on such transfers DARPA will not assert any rights to a royalty on commercial products which included the Verias Simulator, (ii) DARPA will not discriminate in awarding funding to the Buyer in the future as the result of any past transfers to foreign firms or institutions by the Company of any versions of the Verias Simulator in executable code format and (iii) no other monetary fines, reimbursement of funds or rights to royalties will be assessed or demanded by DARPA against the Company based on facts disclosed to DARPA.

Section 5 Conditions to Company's Obligations at Closing

The obligations of the Company hereunder to be performed at the Closing are subject to the satisfaction at or prior to the Closing of the following conditions, except for any condition the Company may waive in accordance with Section 7.3.

- 5.1 Representations and Warranties. The representations and warranties of the Buyer contained in Section 3 shall have been true in all material respects on the date of this Agreement and shall be true in all material respects at and as of immediately prior to the Closing with the same effect as though made at and as of immediately prior to the Closing.
- 5.2 Closing Certificate. The Buyer shall have delivered to the Company a certificate dated the date of the Closing that the conditions specified in Sections 5.1 and 5.3 are satisfied. Such certificate shall be deemed a representation and warranty of the Buyer under Section 3 for all purposes of this Agreement.
- 5.3 Performance. The Buyer shall have performed and complied in all material respects with all covenants required herein to be performed or complied with by the Buyer on or before the Closing.
 - 5.4 Stockholder Approval. The Stockholder Approval shall have been given.
- 5.5 Third-Party Action. All Third-Party Action required in order to consummate the Closing on the terms hereof, other than any the absence of which in the aggregate would not have a material effect on the transactions contemplated hereby, shall have been taken, and the Hart-Scott waiting period shall have expired.
- 5.6 Opinion of Counsel. The Company shall have received from McCutchen, Doyle, Brown & Enersen LLP, counsel to the Buyer, an opinion dated the date of the Closing, in form and substance substantially as set forth in Exhibit D.
- 5.7 Transactional Litigation. No action, suit or proceeding before any Governmental Agency shall have been commenced, and no investigation by any Governmental Agency shall have been commenced or overtly threatened, against the Company, the Buyer, Buyer Subsidiary or any of their respective principals, officers, directors or stockholders seeking to restrain, prevent or change

the transactions contemplated hereby or questioning the validity or legality of any of such transactions or seeking damages in connection with any of such transactions.

5.8 Corporate and Other Proceedings. All corporate and other proceedings on the part of the Buyer and Buyer Subsidiary in connection with the transactions to be consummated at the Closing, and all documents and instruments incident to such transactions, shall be reasonably satisfactory in substance and form to the Company.

Section 6 Covenants of Company, Buyer Subsidiary and Buyer

- Non-Disclosure. Each party agrees not to divulge or communicate, or use for any purpose other than evaluating this transaction or exercising rights as a party hereto, any information or materials concerning this Agreement, the negotiation between the parties hereto and the transactions contemplated hereby, except to the extent that such information (w) is or hereafter becomes lawfully obtainable from other sources, (x) is required to be disclosed to a Governmental Agency having jurisdiction over the party or its Affiliates, (y) is otherwise required by law to be disclosed or (z) is disclosed following a waiver in writing from the other parties. Promptly after the date hereof and after the Effective Time, the Buyer and the Company will issue a mutually agreeable press release concerning the transactions contemplated hereby. The parties hereto will consult and cooperate with each other and agree upon the terms and substance of all press releases, announcements and public statements with respect to this Agreement and the Merger, provided, however, that such consultation and cooperation shall not interfere with any obligation of either party hereto to disclose any information as required by applicable law. Any press release or other announcement by any party with respect to the Merger will be subject to the consent and approval of the other party, which consent or approval will not be unreasonably withheld. The parties also hereby ratify and confirm the confidentiality letter dated October 19, 1999, which shall continue in effect.
- warranties of the Company set forth in this Agreement or any instrument delivered pursuant to this Agreement will expire at and shall not survive the Effective Time, and the Surviving Corporation (as defined in the Plan of Merger) shall have no liability with respect to any such representation or warranty and shall not be subject to any contribution, indemnity or similar claims with respect thereto by any Person. However, nothing in this Section 6.2 will relieve any Person from liability for his, her or its knowing personal fraud.

6.3 Termination of this Agreement; Termination Fees.

(a) If any condition of the Closing stated in Section 4 is not satisfied on or before March 31, 2000, then, provided the Buyer is not in material default hereunder, the Buyer may at any time thereafter terminate any further obligations under this Agreement by giving written notice thereof to the Company. If any condition of the Closing stated in Section 5 is not satisfied on or before such date, then, provided the Company is not in material default hereunder, the Company may at any time thereafter terminate any further obligations under this Agreement by giving written notice thereof to the Buyer. This Agreement may be so terminated, or terminated by mutual agreement of the parties upon the authorization of their respective boards of directors, notwithstanding approval of this Agreement by the stockholders of any or all parties. Furthermore, this Agreement may be

21

PATENT REEL: 011820 FRAME: 0323 terminated by either party if any Governmental Agency shall have issued an injunction or taken any other action (which injunction or other action the parties hereto shall use their reasonable business efforts to lift), which permanently restrains, enjoins or otherwise prohibits the Merger, and such injunction shall have become final and non-appealable.

- (b) In addition, the Board of Directors of the Company may terminate this Agreement pursuant to Section 6.6(e) in the circumstances there specified if, simultaneously with such termination, the Company pays the Buyer \$1,000,000 by wire transfer of immediately available funds.
- (c) In the event the Board of Directors of the Company withdraws or modifies its approval or recommendation of the Merger or approves or recommends a Superior Proposal pursuant to Section 6.6(e), or the Merger is voted down by the Company's stockholders, or this Agreement is terminated by the Buyer prior to the Effective Time after the date specified in Section 6.3(a) for failure to satisfy the conditions specified in Section 4 herein (except for the conditions specified in Sections 4.6, 4.8 (if the suit or proceeding primarily alleges a Buyer legal deficiency), 4.10, 4.11, 4.12, 4.14, 4.15, 4.18 or 4.19), or this Agreement is terminated by the Company prior to the Effective Time after the date specified in Section 6.3(a) for failure to satisfy the conditions specified in Sections 5.4, 5.5 or 5.7 (if the suit or proceeding primarily alleges a Company legal deficiency), or if this Agreement is terminated by the Buyer due to a material breach hereof by the Company, and in any such case a Third-Party Transaction is announced within 12 months after such withdrawal, modification, approval, recommendation, vote or termination which is thereafter consummated (or is consummated within such 12-month period, irrespective of any announcement), the Company will, simultaneously with such consummation, pay the Buyer \$1,000,000 by wire transfer of immediately available funds.
- (d) In the event the Board of Directors of the Company withdraws or modifies its approval or recommendation of the Merger or approves or recommends a Superior Proposal pursuant to Section 6.6(e), or the Merger is voted down by the Company's stockholders, the Buyer may, at any time thereafter, at its sole option by notice given to the Company, terminate any obligation on its or the Buyer Subsidiary's part to consummate the Merger. Any such termination will not affect the Buyer's rights under Section 6.3(c), or, in the case of breach by the Company, be in lieu of or adversely affect in any way any right or remedy otherwise available to the Buyer.
- (e) Any termination pursuant to this Section 6.3 will not, however, terminate or otherwise affect the obligations of the parties under Sections 6.1, 7.1 or 7.2.
- 6.4 Reasonable Business Efforts, No Inconsistent Action. Each party will use its reasonable business efforts to cause the conditions over which it has control to be satisfied on or before the Closing. No party will take any action which will foreseeably result in the nonsatisfaction of any condition stated in Section 4 or 5 on or before the Closing.
- 6.5 Access. Between the date of this Agreement and the Closing or any earlier termination of this Agreement in accordance with its terms, the Company will (i) give the Buyer and its authorized representatives access to its books, records, Properties, officers, attorneys and accountants and permit the Buyer to make inspections and copies of such books and records, and (ii) furnish the Buyer with such financial information and operating data and other information with

respect to its business and Properties, and to discuss with the Buyer and its authorized representative its affairs, all as the Buyer may from time to time reasonably request for the purposes of this Agreement, during normal office hours. Any on-site visit shall be subject to reasonable advance notice and to being accompanied by an officer or designated employee of the Company. No information furnished to the Buyer pursuant to this Section 6.5 or otherwise known to the Buyer shall affect any representation, warranty or condition in this Agreement.

6.6 No Solicitation or Negotiation.

- (a) Until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, neither the Company nor any Subsidiary nor any representative of the Company or any Subsidiary shall, directly or indirectly, take any action to (i) encourage, solicit or initiate the submission of any Acquisition Proposal or any inquiries with respect thereto, (ii) enter into any agreement for or relating to a Third-Party Transaction, or (iii) participate in any way in discussions or negotiations with, or furnish any non-public information to, any Person in connection with any Acquisition Proposal. Notwithstanding any other provision of this Section 6.6(a), the Company may, prior to the Stockholder Approval, in response to an unsolicited bona fide written offer or proposal made by a third party which is reasonably likely to lead to a Superior Proposal, provide non-public information to or have discussions or negotiations with such third party, if and only to the extent that the Board of Directors has determined in good faith, after receiving the advice of its outside counsel, that such action is necessary in order for the Board of Directors to comply with its fiduciary duties to the Company's stockholders under applicable law. The Company will immediately communicate to the Buyer the receipt of any third party solicitation, proposal or bona fide inquiry that the Company, any Subsidiary or any representative of the Company or any Subsidiary may receive in respect of any such transaction, or of any request for such information, including in each case a copy thereof and all other particulars thereof, and keep the Buyer fully apprised of all developments therein on a current basis, and consider in good faith any counterproposals which the Buyer, in its sole discretion, elects to make.
- (b) "Acquisition Proposal" means any proposed Acquisition Transaction. Transaction" means any (i) merger, consolidation or similar transaction involving the Company, (ii) sale, lease or other disposition directly or indirectly by merger, consolidation, share exchange or otherwise of any assets of the Company or its subsidiaries representing 15% or more of the consolidated assets of the Company and its subsidiaries, (iii) issue, sale or other disposition of (including by way of merger, consolidation, share exchange or any similar transaction) securities (or options, rights or warrants to purchase, or securities convertible into, such securities) representing 15% or more of the votes attached to the outstanding securities of the Company, (iv) transaction in which any person shall acquire Beneficial Ownership or the right to acquire Beneficial Ownership, or any Group shall have been formed which has Beneficial Ownership or has the right to acquire Beneficial Ownership, of 15% or more of the outstanding shares of common stock of the Company, (v) recapitalization, restructuring, liquidation, dissolution or other similar type of transaction with respect to the Company or any of its subsidiaries, or (vi) transaction which is similar in form, substance or purpose to any of the foregoing transactions. "Third-Party Transaction" shall mean an Acquisition Transaction with a party unrelated to the Buyer. "Beneficial Ownership" and "Group" shall have the meanings stated in Regulation 13D-G under the Securities Exchange Act of 1934, as amended (the "Exchange Act").

- (c) The Company will take all action necessary in accordance with applicable law and its articles of incorporation and by-laws to convene a meeting of its stockholders as promptly as practicable to consider and vote upon the Merger and to secure the Stockholder Approval, including without limitation the preparation of a proxy statement (the "Proxy Statement," which term shall include all amendments and supplement thereto). The Proxy Statement shall comply as to form in all material respects with the applicable requirements of the Exchange Act and the rules and regulations promulgated thereunder, and shall not, at the time of (i) first mailing thereof or (ii) such stockholders' meeting, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they were made, not misleading, except that the Company shall not be responsible for information supplied by the Buyer or any affiliates or representatives of the Buyer for inclusion in the Proxy Statement. The Company (i) shall promptly prepare and file with the SEC, use all reasonable efforts to have cleared by the SEC and thereafter mail to its stockholders as promptly as practicable the Proxy Statement and all other proxy materials for such meeting, (ii) shall notify the Buyer of the receipt of any comments of the SEC with respect to the Proxy Statement and of any requests by the SEC for any amendment or supplement thereto or for additional information and shall promptly provide the Buyer copies of all correspondence between the Company or any representative of the Company and the SEC and (iii) shall give the Buyer and its counsel the opportunity to review the Proxy Statement prior to its being filed with the SEC and shall give the Buyer and its counsel the opportunity to review all amendments and supplements to the Proxy Statement and all responses to requests for additional information and replies to comments prior to their being filed with, or sent to, the SEC.
- (d) The Board of Directors of the Company shall recommend and declare advisable to its stockholders such approval and the Company shall take all lawful action to solicit, and use all best efforts to obtain, approval of its stockholders, and neither the Board of Directors of the Company nor any committee of such Board of Directors shall, except as otherwise provided in Section 6.6(e) below, (i) withdraw or modify the approval or recommendation by such Board of Directors or such committee of the Merger, (ii) approve or recommend any Acquisition Proposal other than the Merger, or (iii) cause the Company to enter into any letter of intent, agreement in principle, acquisition agreement or other similar agreement with respect to any Acquisition Proposal other than the Merger.
- (e) Notwithstanding Section 6.6(d) hereof, the Board of Directors of the Company may, prior to the Stockholder Approval, withdraw or modify its approval or recommendation of the Merger, approve or recommend a Superior Proposal or terminate this Agreement in order to simultaneously enter into a binding agreement with respect to a Third-Party Transaction that constitutes a Superior Proposal, but in each case subject to its compliance with Section 6.3(b) or (c), as applicable, if and only to the extent that both (i) the Buyer has been given at least 5 days written notice of the Company's intent to do so and (ii) the Board of Directors of the Company has determined in good faith, after receiving the advice of its outside counsel, that such action is necessary in order for the Board of Directors to comply with its fiduciary duties to stockholders under applicable law. A "Superior Proposal" means any bona fide written Acquisition Proposal, the terms of which the Board of Directors of the Company determines in its good faith judgment, based on the advice of its financial advisor, to be more favorable to the Company's stockholders than the Merger and to be already financed or readily financeable.

- (f) Nothing in this Section 6.6 shall prohibit the Company from taking and disclosing to its stockholders a position as contemplated by SEC Rule 14e-2(a), provided that neither the Company nor its Board of Directors nor any committee of its Board of Directors shall approve or recommend any Third-Party Proposal except as permitted by Section 6.6(e).
- 6.7 Interim Financial Information. The Company will supply to the Buyer unaudited consolidated monthly financial statements within 30 business days of the end of each month ending between the date of the Interim Balance Sheet and the Closing or any earlier termination of this Agreement in accordance with its terms, prepared on a basis consistent with the unaudited consolidated financial statements for the preceding months. For purposes of these statements, employee bonuses and similar expenses may be accrued based on actual results for the year to date and budgeted results for the balance of the year.
- 6.8 Interim Conduct of Business. From the date of this Agreement until the Closing or any earlier termination of this Agreement in accordance with its terms, unless approved by the Buyer in writing, the Company will operate its business consistently with past practice and in the ordinary course of business, and will not:
 - (a) merge or consolidate with or agree to merge or consolidate with, or sell or agree to sell all or substantially all of its Property to, or purchase or agree to purchase all or substantially all of the Property of, or otherwise acquire, any other Person or a division thereof, except as provided in this Agreement;
 - (b) amend its articles of incorporation or by-laws;
 - (c) make any changes in its accounting methods, principles or practices, except as required by GAAP;
 - (d) sell, consume or otherwise dispose of any Property, except in the ordinary course of business consistent with past practices;
 - (e) authorize for issuance, issue, sell or deliver any additional shares of its capital stock of any class or any securities or obligations convertible into shares of its capital stock or issue or grant any option, warrant or other right to purchase any shares of its capital stock of any class, other than, in each case, the issuance of Common Stock pursuant to the exercise of the options listed in Section 2.1(a);
 - (f) except as provided in the Plan of Merger, accelerate, amend or change the period of exercisability of vesting of options or other Stock Rights granted under its stock option plans or authorize cash payments in exchange for any options or other Stock Rights granted under any of such plans;
 - (g) declare any dividend on, make any distribution with respect to, or redeem or repurchase, its capital stock except under existing repurchase agreements or obligations as set forth in Schedule 2.1;
 - (h) modify, amend or terminate any Benefit Plans, except as otherwise set forth herein or as required under Legal Requirements or any Disclosable Contract;

- (i) enter into any material Contract, or violate, amend or otherwise modify or waive any of the terms of any of its material Contracts other than amendments or modifications in the ordinary course of business consistent with past practice:
- (j) transfer or license to any Person or otherwise extend, amend or modify any rights to the Company Intellectual Property other than the grant of non-exclusive licenses in the ordinary course of business consistent with past practice;
- (k) enter into or amend any Contracts pursuant to which any other Person is granted exclusive marketing, manufacturing or other exclusive rights of any type or scope with respect to any of its products or technology;
- (l) incur or commit to incur any indebtedness for borrowed money or guarantee any such indebtedness or issue or sell any debt securities or guarantee any debt securities of others;
- (m) enter into any operating leases requiring cumulative annual payments in excess of \$25,000;
- (n) pay, discharge or satisfy in an amount in excess of \$25,000 in any one case or \$50,000 in the aggregate, any claim, liability or obligation (absolute, accrued, asserted or unasserted, contingent or otherwise) arising other than in the ordinary course of business, other than the payment, discharge or satisfaction of liabilities reflected or reserved against in the Financial Statements:
- (o) incur or commit to incur any capital expenditures in excess of \$50,000 in the aggregate;
- (p) materially reduce the amount of any material insurance coverage provided by existing insurance policies;
- (q) terminate or waive any right of substantial value, other than in the ordinary course of business;
- (r) take any of the following actions: (i) increase or agree to increase the compensation payable or to become payable to its officers or employees, except for increases in salary or wages of non-officer employees in the ordinary course of business and in accordance with past practices, (ii) grant any additional severance or termination pay to, or enter into any employment or severance agreements with, any officer or employee, (iii) enter into any collective bargaining agreement, or (iv) establish, adopt, enter into or amend in any material respect any bonus, profit sharing, thrift, compensation, stock option, restricted stock, pension, retirement, deferred compensation, employment, termination, severance or other plan, trust, fund, policy or arrangement for the benefit of any directors, officers or employees, except as otherwise provided in the Plan of Merger;
- (s) commence a lawsuit or arbitration proceeding other than (i) for the routine collection of bills, or (ii) in such cases where it in good faith determines that failure to commence suit would result in the material impairment of a valuable aspect of the Business, provided that it consults with the Buyer prior to the filing of such suit;

- (t) make any material Tax election other than in the ordinary course of business and consistent with past practice, change any material Tax election, adopt any Tax accounting method, file any Tax return (other than any estimated Tax returns, immaterial information returns, payroll Tax returns or sales Tax returns) or any amendment to a Tax return, enter into any closing agreement, settle any Tax claim or assessment or consent to any Tax claim or assessment provided that the Buyer shall not unreasonably withhold or delay approval of any of the foregoing actions;
- (u) revalue any of its assets, including without limitation writing down the value of inventory or writing off notes or accounts receivable other than in the ordinary course of business; or
 - (v) authorize or enter into an agreement to do any of the foregoing.
- 6.9 Section 338 Election; Tax Status. The parties agree that the Buyer may make an election under Section 338(a) of the Code with respect to the Merger. Each party has reviewed the income and other Tax aspects of the structure of the Merger with its own professional advisers, and no party or representative of a party shall have any obligation or responsibility to any other party with respect thereto.

6.10 Option to Purchase.

- (a) The Company hereby irrevocably grants the Buyer the right (the "Option"), at the Buyer's option, to purchase from the Company up to 968,640 (subject to adjustment as provided in this Section 6.10) Shares in the aggregate at the exercise price of \$2.48 per share, subject to adjustment as provided in this Section 6.10 (as so adjusted, the "Exercise Price"), as specified by the Buyer in its notice or notices of exercise from time to time, but only after the occurrence of a Triggering Event. A "Triggering Event" means the first to occur of the following events:
 - (i) the Company terminates this Agreement pursuant to Section 6.6(e), or
 - (ii) the Board of Directors of the Company withdraws or modifies its approval or recommendation of the Merger or approves or recommends a Superior Proposal pursuant to Section 6.6(e), or the Merger is voted down by the Company's stockholders, or this Agreement is terminated by the Buyer prior to the Effective Time after the date specified in Section 6.3(a) for failure to satisfy the conditions specified in Section 4 herein (except for the conditions specified in Sections 4.6, 4.8 (if the suit or proceeding primarily alleges a Buyer legal deficiency), 4.10, 4.11, 4.12, 4.14, 4.15, 4.18 or 4.19), or this Agreement is terminated by the Company prior to the Effective Time after the date specified in Section 6.3(a) for failure to satisfy the conditions specified in Sections 5.4, 5.5 or 5.7 (if the suit or proceeding primarily alleges a Company legal deficiency), or if this Agreement is terminated by the Buyer due to a material breach hereof by the Company, and in any such case a Third-Party Transaction is announced within 12 months after such withdrawal, modification, approval, recommendation, vote or termination which is thereafter consummated (or is consummated within such 12-month period, irrespective of any announcement). Such consummation shall constitute the Triggering Event within the scope of this Section 6.10(a)(ii).

- (iii) any Person other than an Affiliate of the Buyer commences a tender offer (within the meaning of SEC Rule 14d-2) for 15% of more of the Company's outstanding Shares.
- (b) The Company will give the Buyer notice of any proposed Triggering Event under Section 6.10(a)(ii) or (iii) to which it is a party at least 20 days prior to the proposed consummation thereof, and will give the Buyer notice of any other such Triggering Event immediately upon receiving knowledge thereof. The Option may be exercised, at any time and from time to time, commencing immediately prior to the Triggering Event and continuing thereafter until 5:00 p.m., California time, on the earlier of (i) the Effective Time or (ii) the 5th anniversary of the date of this Agreement, if a business day, or on the next succeeding business day if it is not. To exercise the Option, the Buyer shall deliver written notice to the Company at its address listed in Section 7.4 specifying the number of Shares as to which the Option is being exercised, accompanied by the Buyer's check or wire transfer in payment of the aggregate Exercise Price. Alternatively, the Buyer may pay the Exercise Price by surrender of this Option with respect to a number of shares whose aggregate Spread Value equals the aggregate Exercise Price. "Spread Value" means (x) the excess, if any, of the market value of a Share over the Exercise Price times (y) the number of Shares so to be surrendered. For this purpose, "market value" shall mean the average closing price of Shares for the 3 most recent trading days ending with the 2d trading day prior to any such cashless exercise, if Shares are then traded on a national securities exchange or the Nasdaq National Market, and will mean the per-Share value of the transaction associated with the Triggering Event in all other cases.
- (c) Upon any exercise of the Option as set forth above, the Buyer shall immediately be the record owner of all Shares subject to such exercise for all purposes, without any other action being necessary. Within 3 trading days after each exercise of the Option, the Company shall deliver certificates for the Shares so purchased to the Buyer, but the delivery of such certificates shall not be required in order for the Buyer to exercise any rights as a holder of the Shares to be represented thereby. Until its exercise, the Option does not confer on the Buyer any rights of a stockholder of the Company.
- (d) The Company need not issue any fractional shares in connection with any exercise of the Option. Instead, the Buyer may purchase a whole Share from the Company at the Exercise Price.
- (e) During the term of the Option, the Company shall reserve sufficient authorized but unissued shares of Common Stock or other securities for the full exercise of the rights represented by the Option. To the extent required for the lawful exercise of the Option, the Company will promptly make, at the Buyer's expense, all filings with Governmental Agencies, including without limitation a premerger notification under the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, requested by the Buyer in connection with an intended exercise of the option.
- (f) The Exercise Price and the number of Shares that the Company must issue upon exercise of the Option are subject to adjustment from time to time as follows:
 - (1) If the Company at any time or from time to time after the date hereof: (1) declares or pays, without consideration, any dividend on Shares payable in Shares; (2) creates any right to acquire Shares for no consideration; or (3) subdivides the outstanding Shares (by stock split, reclassification or otherwise), the Company shall increase the number of Shares that the Buyer

may purchase upon exercising the Option and decrease the Exercise Price in proportion to the increase in the number of outstanding Shares that results from any such action.

- (2) If the Company combines or consolidates the outstanding Shares, by reclassification or otherwise, into a lesser number of Shares, the Company shall decrease the number of Shares that the Buyer may purchase upon exercising the Option and increase the Exercise Price in proportion to the decrease in the number of outstanding Shares that results from any such combination or consolidation.
- (3) If Shares change into shares of any other class or classes of security or into any other Property for any reason other than a subdivision or combination of Shares provided for in Section 6.10(f)(1) or (2), including without limitation any reorganization, reclassification, merger or consolidation and any sale of substantially all of the Company's properties and assets, the Company shall make lawful provision for giving the Buyer the right, by exercising the Option, to purchase the kind and amount of securities or other Property receivable upon any such change by the owner of the number of Shares subject to this Option immediately before the change.
- (4) If the Company spins off any Subsidiary by distributing to the Company's shareholders as a dividend or otherwise any stock or other securities of the Subsidiary, the Company shall reserve until the end of the term of the Option enough such shares or other securities for delivery to the Buyer upon any exercise of the rights represented by the Option to the same extent as if the Buyer owned of record all Common Stock or other securities subject to the Option on the record date for the distribution of the Subsidiary's shares or other securities.

Upon each adjustment or readjustment required by this Section 6.10(f), the Company shall promptly compute such adjustment or readjustment and furnish to the Buyer a certificate setting forth such adjustment or readjustment and showing in detail the facts giving rise to the adjustment or readjustment. Upon the Buyer's written request, the Company also shall furnish to the Buyer a similar certificate setting forth (1) such adjustments and readjustments, (2) the Option Price in effect on the date of the certificate, and (3) the number of Shares and the amount of any other property that the Buyer would receive upon exercising the Option.

- (g) The Buyer may not transfer, sell or make any other disposition of the Option (other than in connection with a succession to or transfer of its business as a whole), or grant any Lien respecting any of its rights under the Option, without the Company's prior consent.
- (h) By accepting the Option, the Buyer agrees that the Option and the Shares or other securities issuable upon exercise of the Option may be offered or sold, only in compliance with the Act and all applicable state securities laws. The Buyer acknowledges that if the Option is exercised, the shares issued upon such exercise will be "restricted securities" within the meaning of Rule 144 promulgated under the Act. The Buyer hereby agrees to comply with this Section 6.10(h) with respect to any resale or other disposition of such securities. The Buyer may offer or sell any such securities only in accordance with (1) an effective registration statement under the Act; (2) SEC Rule 144; or (3) another exemption from the registration requirements of the Act and all applicable state securities laws demonstrated, to the Company's reasonable satisfaction, by an opinion of securities counsel reasonably

acceptable to the Company. The Company may make a notation on its records, and on the certificates for any Shares or other securities issued upon the exercise of the Option, to implement the restrictions set forth in this Section 6.10(h). The Buyer represents and warrants that it is acquiring the Option, and will acquire the Shares subject thereto, for its own account and not on behalf of any other Person, and that it is an "accredited investor," as that term is defined in SEC Regulation D. By accepting the Option, the Buyer acknowledges that the Company is granting the Option to the Buyer in reliance on the Buyer's foregoing representations and warranties and the terms of this Section 6.10(h). Before allowing any transferee of any of the Buyer's rights under the Option to exercise this Option, the Company may, in its sole discretion, require the transferee to execute and deliver representations, warranties and acknowledgments to the Company substantially similar to those set forth in this Section 6.10(h).

- (i) In the event the Company proposes to consummate a transaction (however structured, including without limitation a sale of stock or a merger) to which it is a party and which involves an acquisition of shares or equity interest in the Company by a third Person, and immediately following such transaction such Person together with its parents and subsidiaries (collectively, the "Acquiring Person") owns a majority of both the voting power and economic interest represented by the outstanding stock of the Company on a fully-diluted basis, then the Company (x) shall give the Buyer at least 20 days written notice of the proposed consummation, including a summary of the transaction and a copy of all transaction documents (which summary and documents shall be kept current by further notices to the Buyer) and (y) if it has complied with clause (x), may at its option, exercised by notice to the Buyer not later than 5 days prior to such consummation, redeem the Option (in whole and not in part) simultaneously with such consummation for an amount per share, paid by wire transfer of same-day funds to the account designated by the Buyer, equal to the excess, if any, of the Redemption Price over the Exercise Price then in effect. The "Redemption Price" is:
 - (i) if the equity interest the Acquiring Person is acquiring in such consummation, together with any equity interest acquired by such Acquiring Person within 180 days prior to such consummation constitutes a majority of both the voting power and economic interest represented by the outstanding stock of the Company on a fully-diluted basis, the highest price per share paid (or to be paid upon such consummation) by the Acquiring Person in the course of acquiring such majority ownership, and
 - (ii) otherwise, the fair market value of a Share (or the other Property then subject to the Option) as agreed to by the Company and the Buyer or, in the absence of such agreement, as determined by an investment bank of national reputation selected by the Company from a list of 3 such banks proposed by the Buyer (on its own initiative or within 15 days after written request by the Company), which bank shall not have engaged in a transaction with, or advised with respect to a transaction, either the Company or the Buyer within the prior 3 years. If this clause (ii) is applicable, the Company will not consummate such transaction unless and until the fair market value has been agreed or determined pursuant to this clause (ii).
- (j) Any Shares purchased by the Buyer upon exercise of the Option shall be entitled to the registration rights set forth in $Exhibit\ E$.

- 6.11 SEC Reports. From and after the date of this Agreement until the earlier of the termination of this Agreement pursuant to its terms or the Effective Time, the Company will timely file all reports required to be filed by it under the Exchange Act.
- 6.12 Notice of Certain Events. The Company shall notify the Buyer, and the Buyer shall promptly notify the Company, of:
 - (i) receipt of any notice or other communication from any Person alleging that the consent of such Person is or may be required in connection with the transactions contemplated by this Agreement;
 - (ii) receipt of any notice or other communication from any Governmental Entity in connection with the transactions contemplated by this Agreement;
 - (iii)receipt of notice that any action, suit, claim, investigation or proceeding has been commenced or, to the knowledge of the Company, threatened, against or involving the Company, any Subsidiary or the Buyer, as applicable, which, if pending on the date of this Agreement, would have been required to have been disclosed pursuant to Section 2.18 or which relates to the transactions contemplated by this Agreement;
 - (iv) the occurrence or non-occurrence of any event the occurrence or non-occurrence of which would be likely to cause any representation or warranty of it (and, in the case of the Buyer, of Buyer Subsidiary) contained in this Agreement to be untrue or inaccurate; and
 - (v) any failure of the Company, the Buyer or Buyer Subsidiary, as the case may be, to comply with or satisfy any covenant, condition or agreement to be complied with or satisfied by it hereunder.

The delivery of any notice pursuant to this Section 6.12 shall not limit or otherwise affect the remedies available to the party receiving such notice.

- 6.13 Takeover Statutes. If any Takeover Statute is, becomes or may become applicable to the Merger or any of the transactions contemplated hereby, each of the Buyer, Buyer Subsidiary and the Company, and their respective Boards of Directors, shall grant such approvals and take such lawful actions as are necessary to ensure that the Merger and such transactions may be consummated as promptly as practicable on the terms contemplated hereby, and to the extent permitted by law otherwise act to eliminate the effects of such statute and any regulations promulgated thereunder on the Merger and such transactions or, if they cannot be eliminated, to minimize them.
- 6.14 Pay-Off. The Company will, upon the request of the Buyer, cooperate with the Buyer in arranging the pay-off or refinancing of any indebtedness of the Company set forth in Schedule 2.6(b) which the Buyer, in its business discretion, desires to pay off or refinance in connection with the consummation of the transactions contemplated hereby.
- 6.15 Other Filings. As promptly as practicable, the Company, the Buyer and Buyer Subsidiary each shall properly prepare and file any other filings required under the Exchange Act or any other federal or state law relating to the Merger and the transactions contemplated hereby (including filings, if any, required under Hart-Scott)(collectively, the "Other Filings"). Each of the

Company, the Buyer and Buyer Subsidiary shall promptly notify the other of the receipt of any comments on, or any request for amendments or supplements to, any of the Other Filings by any Governmental Agency or official, and each of the Company, the Buyer and Buyer Subsidiary shall supply the other with copies of all correspondence between it, on the one hand, and members of any Governmental Agency or any other appropriate governmental official, on the other hand, with respect to any of the Other Filings. The Company, the Buyer and Buyer Subsidiary each shall use its respective reasonable best efforts to obtain and furnish the information required to be included in any of the Other Filings. The Company, the Buyer and Buyer Subsidiary hereby covenant and agree to use their commercially reasonable best efforts to secure termination of any waiting periods under Hart-Scott and obtain the approval of the Federal Trade Commission or any other Governmental Agency for the transactions contemplated herein.

6.16 Officers and Directors Indemnification. The Company shall, and from and after the Effective Time, the Surviving Corporation shall, indemnify, defend and hold harmless the present and former directors, officers, employees and agents of the Company or any Subsidiaries (the "Indemnified Parties") against all losses, claims, damages, costs, expenses (including reasonable attorney's fees and expenses), liabilities or judgments or amounts that are paid in settlement with the approval of the indemnifying party of or in connection with any threatened or actual claim, action, suit, proceeding or investigation based in whole or in part on or arising in whole or in part out of or pertaining to the fact that such person is or was a director or officer of the Company or any of the Subsidiaries whether pertaining to any matter existing at or prior to the Effective Time and whether asserted or claimed prior to, or at or after, the Effective Time ("Indemnified Liabilities"), including all Indemnified Liabilities based in whole or in part on, or arising in whole or in part out of, or pertaining to this Agreement or the transactions contemplated hereby, in each case to the fullest extent a corporation may indemnify its own directors or officers, as the case may be, in compliance with applicable law, under the OBCA as the same exists or may hereafter be amended (but, in the case of any such amendment, only to the extent that such amendment permits broader rights than such law permitted prior to such amendment and only to the extent such amendment is not retroactively applicable). Without limiting the foregoing, in the event any such claim, action, suit, proceeding or investigation is brought against any Indemnified Parties (whether arising before or after the Effective Time), (i) the Indemnified Parties may retain counsel satisfactory to them and the Surviving Corporation, and the Company or the Surviving Corporation shall pay all reasonable fees and expenses of such counsel for the Indemnified Parties promptly as statements therefor are received and otherwise advance to such Indemnified Parties upon request reimbursement of reasonable documented expenses incurred, in either case to the fullest extent and in the manner permitted by the OBCA; and (ii) the Company or the Surviving Corporation, and the Indemnified Party, will use all reasonable efforts to assist in the vigorous defense of any such matter, provided that neither the Company nor the Surviving Corporation shall be liable for any settlement effected without its prior written consent. Any Indemnified Party wishing to claim indemnification under this Section 6.16, upon learning of any such claim, action, suit, proceeding or investigation, shall notify the Company (or after the Effective Time, the Surviving Corporation) (but the failure to so notify shall not relieve a party from any liability which it may have under this Section 6.16 except to the extent such failure materially prejudices such party). The Indemnified Parties as a group may retain only one law firm to represent them with respect to each such matter unless there is, under applicable standards of professional conduct, a conflict on any significant issue between the positions of any two or more Indemnified Parties. The Company, Buyer and Buyer Subsidiary agree that all rights to indemnification, including provisions relating to advances or expenses incurred in defense of any action or suit, existing in favor of the Indemnified Parties with respect to matters occurring through the Effective Time, shall survive the Merger and shall continue in full force and effect for a period of not less than three years from the Effective Time; provided, however, that all rights to indemnification in respect of any Indemnified Liabilities asserted or made within such period shall continue until the disposition of such Indemnified Liabilities. This Section 6.16 is for the irrevocable benefit of, and to grant third party rights to, the Indemnified Parties and shall be binding on all successors and assigns of Buyer, Buyer Subsidiary, the Company and the Surviving Corporation. Each of the Indemnified Parties shall be entitled to enforce the covenants contained in this Section 6.16.

Section 7 Miscellaneous

7.1 No Brokers, Finders.

- (a) Company. The Company has not engaged any agent, broker, finder or investment or commercial banker in connection with the negotiation, execution or performance of this Agreement or the transactions contemplated hereby, other than Broadview Int'l LLC, for whose fees and expenses the Company will be solely responsible and whose fees and expenses will not exceed \$600,000 if the Merger is consummated. The Company shall indemnify, defend and hold the Buyer harmless against and in respect of any claim for brokerage fees or other commissions incurred or owing due to any such engagement or alleged engagement, including without limitation, any fees and expenses of counsel incurred by the Buyer in connection with enforcing this Section 7.1(a).
- (b) Buyer. The Buyer has not engaged any agent, broker, finder or investment or commercial banker in connection with the negotiation, execution or performance of this Agreement or the transactions contemplated hereby. The Buyer shall indemnify, defend and hold the Company and its stockholders harmless against and in respect of any claim for brokerage fees or other commissions incurred or owing due to any such engagement or alleged engagement, including without limitation, any fees and expenses of counsel incurred by the Company or its stockholders in connection with enforcing this Section 7.1(b).
- 7.2 Expenses. Whether or not the transactions contemplated by this Agreement are consummated, the Company and the Buyer shall each pay their own fees and expenses incident to the negotiation, preparation, execution, delivery and performance hereof, including, without limitation, the fees and expenses of their respective counsel, accountants and other experts.
- 7.3 Complete Agreement; Waiver and Modification; No Third Party Beneficiaries. This Agreement constitutes the entire agreement between the parties pertaining to the subject matter hereof and supersedes all prior and contemporaneous agreements and understandings of the parties other than the confidentiality letter dated October 19, 1999 between the Buyer and the Company, which shall continue in effect. There are no representations or warranties by any party except those expressly stated or provided for herein, any implied warranties being hereby expressly disclaimed. There are no covenants or conditions except those expressly stated herein. No amendment, supplement or termination of or to this Agreement, and no waiver of any of the provisions hereof, shall be binding on a party unless made in a writing signed by such party. This Agreement may be modified by mutual agreement of the parties as authorized by their respective boards of directors, notwithstanding approval hereof and thereof by the stockholders of the parties. Nothing in this

Agreement shall be construed to give any Person other than the express parties hereto any rights or remedies.

- 7.4 Notices. All notices, requests, demands, claims and other communications hereunder shall be in writing and shall be given by delivery (by mail or otherwise) or transmitted to the address or facsimile number listed below, and will be effective (in all cases) upon receipt. Without limiting the generality of the foregoing, a mail, express, messenger or other receipt signed by any Person at such address shall conclusively evidence delivery to and receipt at such address, and any printout showing successful facsimile transmission of the correct total pages to the correct facsimile number shall conclusively evidence transmission to and receipt at such facsimile number.
 - (a) If to the Buyer or Subsidiary:

46871 Bayside Parkway Fremont, CA 94538 attention: Sam Chang

facsimile: (510) 413-7793

with copies to:

McCutchen, Doyle, Brown & Enersen, LLP 3150 Porter Drive Palo Alto, CA 94304 attention: Bartley C. Deamer

(b) If to the Company:

9205 S.W. Gemini Drive Beaverton, OR 97008 attention: Gary P. Arnold

facsimile: (503) 641-3193

with copies to:

Ater Wynne LLP 222 SW Columbia, Suite 1800 Portland, OR 97201 attention: Stephen M. Going

Any party may change its address or facsimile number for purposes of this Section 7.4 by giving the other party written notice of the new address or facsimile number in accordance with this Section 7.4, provided it is a normal street address, or normal operating facsimile number, in the continental United States.

- 7.5 Law Governing. This Agreement shall be interpreted in accordance with and governed by the laws of the State of California, without regard to principles of conflicts of laws, except as to matters manditorily governed by the OBCA.
- 7.6 Headings; References; "Hereof;" Interpretation. The Section headings in this Agreement are provided for convenience only, and shall not be considered in the interpretation hereof. References herein to Sections, Exhibits or Schedules refer, unless otherwise specified, to the designated Section of or Exhibit or Schedule to this Agreement. Terms such as "herein," "hereto" and "hereof" refer to this Agreement as a whole. This Agreement has been negotiated at arm's length between parties sophisticated and knowledgeable in the matters addressed in this Agreement. Each of the parties has been represented by experienced and knowledgeable legal counsel. Accordingly, any rule of law or legal decision that would require interpretation of any ambiguities in this Agreement against the party that has drafted it is not applicable and is waived. The provisions of this Agreement shall be interpreted in a reasonable manner to effect the purpose of the parties and this Agreement.
- 7.7 Successors and Assigns. This Agreement shall inure to the benefit of and be binding upon the heirs, executors, administrators and successors of the parties hereto, but no right or liability or obligation arising hereunder may be assigned by any party hereto.
- 7.8 Counterparts, Separate Signature Pages. This Agreement may be executed in any number of counterparts, or using separate signature pages. Each such executed counterpart and each counterpart to which such signature pages are attached shall be deemed to be an original instrument, but all such counterparts together shall constitute one and the same instrument.
- 7.9 Severability. In the event any of the provisions of this Agreement shall be declared by a court or arbitrator to be void or unenforceable, then such provision shall be severed from this Agreement without affecting the validity and enforceability of any of the other provisions hereof, and the parties shall negotiate in good faith to replace such unenforceable or void provisions with a similar clause to achieve, to the extent permitted under law, the purpose and intent of the provisions declared void and unenforceable.

Section 8 Glossary

Acquiring Person - Section 6.10(i).

Acquisition Proposal - Section 6.6(b).

Acquisition Transaction - Section 6.6(b).

Act - the Securities Act of 1933, as amended.

Affiliate – a Person who controls, is controlled by or is under common control with another Person, or who directly or indirectly owns 10% or more of the voting power in such other Person, or of whose voting power such other Person (or a Person holding 10% or more of the voting power in such other Person) owns 10% or more. For purposes of this definition, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management

and policies of a Person, whether through the ownership of voting securities, by contract or otherwise.

Agreement - this Agreement of Merger, including the Exhibits and Schedules hereto.

Audited Statements - Section 2.6(a).

Beneficial Ownership - Section 6.6(b).

Benefit Plans - Section 2.14(b).

Business – introductory paragraphs.

Buyer - introductory paragraphs.

Buyer Subsidiary - introductory paragraphs.

Closing – Section 1.1.

COBRA - Section 2.14(f).

Code – the Internal Revenue Code of 1986, as amended.

Company – introductory paragraphs. In the case of any representation or warranty relating to events or circumstances occurring or existing prior to the merger of Analogy Acquisition Corporation into Symmetry, Inc., the term "Company" also includes Symmetry, Inc.

Company ERISA Plan - Section 2.14(d).

Company Intellectual Property - Section 2.21(a).

Company SEC Documents - Section 2.25.

Confidential Information - Section 2.21(g).

Contract – any agreement, written or oral, any license or authorization by another Person of a contractual nature or any promissory note or other instrument of a contractual nature, which is intended to be enforceable against the Person in question or against any Property of such Person. Any Person which is, or any of whose Property is, subject to enforcement of a Contract shall, for purposes of this Agreement, be deemed a party to it.

DARPA - Section 4.19.

DARPA Agreement – the Cooperative Agreement Under 10 U.S.C. 2358 between the United States of America U.S. Air Force, Air Force Materiel Command Rome Laboratory/PKPF and Analogy, Inc. (F30602-96-2-0309), with a term from September 5, 1996 through April 4, 1999.

Designated Contract - any of the following:

- (a) the lease to the Company referred to in *Schedule 2.11* for 8705 SW Nimbus Avenue, Beaverton, Oregon, and for 9205 S.W. Gemini Drive, Beaverton, Oregon, and
 - (b) the following agreements:
- (i) Tauri OEM Software Agreement between FTL Systems, Inc. and the Company, dated May 21, 1999;
- (ii) Source Code License and Binary Code Distribution License Agreement between Phase III Logic, Inc. and the Company, dated June 7, 1991, as amended including Addendum No. 3 effective November 3, 1993;
- (iii) Cooperative Agreement under 10 U.S.C. 2358 between the United States of America U.S. Air Force, air Force Materiel Command Rome Laboratory/PKPF and the Company, effective as of September 5, 1996;
- (iv) Cross-License and Support Agreement between Cadence Design Systems, Inc. and the Company, dated June 20, 1994;
- (v) Connections Partnership Membership Agreement between Cadence Design Systems, Inc. and Symmetry Design Systems, Inc., dated August 1, 1996;
- (vi) Mentor Graphics OpenDoor Loan and Support Agreement for OpenDoor Integration, Agreement No. OD-1004-LS-8, between Mentor Graphics Corporation and Analogy, Inc., dated January 12, 1999;
- (vii) Joint Marketing Agreement between Viewlogic Systems, Inc. and the Company, dated September 13, 1993;
- (viii) Remarketing and Distribution License Agreement between Mathworks, Inc. and the Company, dated January 16, 1994;
- (ix) Participation Agreement between Model Technology Incorporated and the Company, dated May 22, 1998;
- (x) Open Access Agreement between GEC Marconi Avionics (Holdings) Ltd. and Analogy UK, dated April 1, 1997, as amended March 31, 1999 and June 30, 1999;
- (xi) Open Access Agreement between GEC Alsthom ERC and Analogy UK, dated December 19, 1997;
- (xii) Open Access Agreement between GEC Alsthom Traction Ltd. and Analogy UK, dated December 19, 1997;
- (xiii) Open Access Agreement between GEC Marconi Radar & Defense Systems Ltd., Dynamics Division, and Analogy UK, dated January 1998;

- (xiv) Open Access Agreement between GEC Marconi Ltd. (MES) and Analogy UK, dated September 24, 1998 and as renegotiated October 25, 1999;
- (xv) Open Access Agreement between Lucas Aerospace and Analogy UK, dated December 19, 1997, as amended March 31, 1999;
- (xvi) Open Access Agreement between Peugeot Sport (PSA) and Analogy SARL, dated March 20, 1998; and
- (xvii) Distribution Agreement between C-Itoh & Co. and the Company, dated August 16, 1998.

Disclosable Contract - Section 2.15.

Disclosable Leases - Section 2.11.

EAR – the Export Administration Regulations promulgated by the United States Department of Commerce and set forth in parts 730-774, inclusive, of Title 15 of the Code of Federal Regulations.

ECCN - Section 2.30.

Effective Time - Section 1.2.

Environmental Requirement – any Legal Requirement relating to pollution, waste, disposal, industrial hygiene, land use or the protection of human health, safety or welfare, plant life or animal life, natural resources, wetlands, endangered or threatened species or habitat, the environment or property, including without limitation those pertaining to reporting, licensing, permitting, controlling, investigating or remediating emissions, discharges, releases or threatened releases of Hazardous Materials, chemical substances, pollutants, contaminants or toxic substances, materials or wastes, whether solid, liquid or gaseous in nature, into the air, surface water, groundwater or land, or relating to the manufacture, generation, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Material, chemical substances, pollutants, contaminants or toxic substances, materials or wastes, whether solid, liquid or gaseous in nature.

Equity Interest – any common stock, preferred stock, partnership interest, limited liability company interest or ownership interest in any Person, and any right to acquire any of the foregoing, whether by exercise of an option, warrant or other right, by conversion, exchange or subscription or otherwise.

ERISA - the Employee Retirement Income Security Act of 1974, as amended, and any successor statute.

ERISA Affiliate – any company which, as of the relevant measuring date under ERISA, is or was a member of a controlled group of corporations or trades or businesses (as defined in Sections 414(b), (c), (m) or (o) of the Code) of which the Company or any Subsidiary or any predecessor of either is or was a member.

38

Exchange Act – Section 6.6(b).

Exercise Price - Section 6.10(a).

Financial Statements – Section 2.6(a).

GAAP – generally accepted accounting principles applied on a consistent basis, as set forth in authoritative pronouncements which are applicable to the circumstances as of the date in question. The requirement that such principles be applied on a "consistent basis" means that accounting principles observed in the period in question are comparable in all material respects to those applied in the preceding periods, except as change is permitted or required under or pursuant to such accounting principles.

Government Audit – Section 2.18(b).

Governmental Agency – any agency, court, department, board, commission, district or other public organ, whether federal, state, local or foreign.

Group – Section 6.6(b).

Hart-Scott – the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and regulations issued thereunder.

Hazardous Material – all or any of the following: (i) any substance the presence of which requires investigation or remediation under any applicable law or regulation; (ii) substances that are defined or listed in, or otherwise classified pursuant to, any applicable laws or regulations as "hazardous substances," "hazardous materials," "hazardous wastes," "toxic substances," or any other formulation intended to define, list or classify substances by reason of deleterious properties such as ignitability, corrosivity, reactivity, carcinogenicity, reproductive toxicity or "TCLP toxicity;" (iii) any petroleum products, explosives or radioactive materials; and (iv) asbestos in any form or electrical equipment which contains any oil or dielectric fluid containing levels of polychlorinated biphenyls in excess of fifty parts per million.

Indemnified Liabilities - Section 6.16.

Indemnified Parties - Section 6.16.

Intellectual Property - Section 2.21(a).

Interim Balance Sheet - Section 2.6(c).

Interim Statements - Section 2.6(a).

ISO - Section 2.14(h).

Last Fiscal Year-End - Section 2.6(a).

Legal Requirement – a statute, regulation, ordinance or similar legal requirement, whether federal, state, local or foreign, or any requirement of a Permit or other authorization issued by a Governmental Agency.

Lien – any lien, security interest, mortgage, deed of trust, pledge, hypothecation, capitalized lease or interest or right for security purposes.

Material Adverse Effect – a matter will be deemed to have a "Material Adverse Effect" if such matter would have a material adverse effect on the business, condition, assets, liabilities, operations, financial performance or prospects of the Company and its Subsidiaries taken as a whole.

Merger – introductory paragraphs.

OBCA - Section 2.27.

Option – Section 6.10(a).

Order – any judgment, injunction, order or similar mandatory direction of, or stipulation or agreement filed with, a Governmental Agency, court, judicial body, arbitrator or arbitral body.

Option - Section 6.10(a).

Other Filings - Section 6.15.

Person — an individual, or a corporation, partnership, limited liability company, trust, association or other entity of any nature, or a Governmental Agency.

Plan of Merger - Section 1.2.

Plan Termination Date - Section 2.29.

Programs – Section 2.21(a).

Property – any interest in any real, personal or mixed property, whether tangible or intangible.

Proxy Statement - Section 6.6(c).

Redemption Price - Section 6.10(i).

SEC - the Securities and Exchange Commission.

Shares - shares of the Common Stock, no par value, of the Company.

Spread Value - Section 6.10(b).

Stock Purchase Plan - Section 2.29.

Stock Right – any right (including without limitation any option or warrant or subscription right) to acquire any capital stock or any other Stock Right or any instrument convertible into or exchangeable for any capital stock or any other Stock Right.

Stockholder Approval - Section 2.3.

Subsidiary – any Person which would be included in consolidated financial statements of the Company prepared in accordance with GAAP, and any Person in which the Company holds 50% or more of the voting power or 50% or more of the equity interests, and any former Subsidiary with respect to any of whose obligations the Company or any current Subsidiary is liable. In the case of any representation or warranty relating to events or circumstances in the past, the term "Subsidiary" also includes any Person that at the relevant time was a Subsidiary, irrespective of such Person's current status as a Subsidiary.

Superior Proposal – Section 6.6(e).

Support Agreements – Section 2.21(h).

Takeover Statute - Section 2.26.

Tax – any federal, state, local or foreign tax, assessment, duty, fee and other governmental charge or imposition of any kind, whether measured by properties, assets, wages, payroll, purchases, value added, payments, sales, use, business, capital stock, surplus or income, and any addition, interest, penalty, deficiency imposed with respect to any Tax.

Third-Party Action – any consent, waiver, approval, license or other authorization of, or notice to, or filing with, any other Person, whether or not a Governmental Agency, and the expiration of any associated mandatory waiting period.

Third-Party Intellectual Property - Section 2.21(a).

Third-Party Right — any Lien on any Property of the Person in question, or any right (other than the rights of the Buyer hereunder) (i) to acquire, lease, use, dispose of, vote or exercise any right or power conferred by any Property of such Person, or (ii) restricting the Person's right to lease, use, dispose of, vote or exercise any right or power conferred by any Property of such Person.

Third-Party Transaction - Section 6.6(b)

Triggering Event - Section 2.10(a).

[remainder of page left blank intentionally]

IN WITNESS WHEREOF, the parties have executed this Agreement of Merger.

AVANT! CORPORATION

Name: Charman, President & CE

Title: GERALD C. Hen

By: Tualde. The

Name: Title:

ANALOGY INC.

By:
Name: Gary Alex
Title:

Exhibits

Company:

Buyer:

A Plan of Merger

Buyer Subsidiary:

B Opinion of Company's Counsel

C Form of Noncompetition Agreement

D Opinion of Buyer's Counsel

E Registration Rights

Schedules

Exhibit A

Plan of Merger

Exhibit B

Form of Opinion of Company's Counsel

Exhibit C

Form of Noncompetition Agreement

Exhibit C

PROPRIETARY INFORMATION AND INVENTIONS AGREEMENT

, 20	ሰሰ
, 20	U

Avant! Corporation 46871 Bayside Pkwy. Fremont, CA 94538

Gentlemen:

The following confirms an agreement between me and Avant! Corporation, a Delaware corporation (the "Company" which term includes the Company's subsidiaries, successors and assigns), which is a material part of the consideration for my employment by the Company:

- 1. "Proprietary Information" is information that was or is developed by, became or becomes known by, or was or is assigned or otherwise conveyed to the Company, and which has commercial value in the Company's business. Proprietary Information includes, without limitation, trade secrets, financial information, product plans, customer lists, marketing plans and strategies, forecasts and other business information, improvements, inventions, formulas, ideas, circuits, mask works, works of authorship, processes, computer programs, algorithms, techniques, schematics, know-how and data. I understand that my employment creates a relationship of confidence and trust between me and the Company with respect to Proprietary Information of the Company or its customers which may be learned by me during the period of my employment.
- 2. In consideration of my employment by the Company and the compensation received by me from the Company from time to time, I hereby agree as follows:
- (a) All Proprietary Information and all patents, copyrights, trade secret rights, rights with respect to mask works and other rights (including throughout, without limitation, any extensions, renewals, continuations or divisions of any of the foregoing) in connection therewith shall be the sole property of the Company. I hereby assign to the Company any rights I may have or acquire in such Proprietary Information. At all times, both during my employment by the Company and after its termination, I will keep in confidence and trust and will not use or disclose any Proprietary Information or anything relating to it without the written consent of the Company, except as may be necessary in the ordinary course of performing my duties to the Company.
- (b) In the event of the termination of my employment by me or by the Company for any reason, I shall return all documents, records, apparatus, equipment and other physical property, or any reproduction of such property, whether or not pertaining to Proprietary Information, furnished to me by the Company or produced by myself or others in connection with my employment, to the Company immediately as and when requested by the Company.

- (c) I will promptly disclose to the Company, or any persons designated by it, all "Inventions" which includes all improvements, inventions, formulas, ideas, circuits, mask works, techniques, schematics, know-how and data, whether or not patentable, made or conceived or reduced to practice or developed by me either alone or jointly with others, during the term of my employment. To the extent the Company does not have rights therein hereunder, such disclosure shall be received by the Company in confidence and does not extend the assignment made in Section (e) below.
- (d) During the term of my Company employment and for a period of one year after the termination of my Company employment, I will not, directly or indirectly, whether through a third party or otherwise, recruit, induce, invite, solicit or otherwise encourage any Company employee to leave the Company for any reason or to devote less than all of any such employee's efforts to the affairs of the Company, provided that the foregoing shall not affect any responsibility I may have as an employee of the Company with respect to the bonafide hiring and firing of Company personnel.
- (e) I agree that all Inventions which I make, conceive, reduce to practice or develop (in whole or in part, either alone or jointly with others) during my employment shall be the sole property of the Company to the maximum extent permitted by Section 2870 of the California Labor Code, a copy of which is attached herewith, and to the extent, permitted by law shall be "works made for hire". The Company shall be the sole owner of all patents, copyrights, trade secret rights, rights with respect to mask works and other intellectual property or other rights in connection therewith. I hereby assign to the Company any rights I may have or acquire in such Inventions. I agree to perform, during and after my employment, all acts deemed necessary or desirable by the Company to permit and assist it, at the Company's expense, in obtaining and enforcing patents, copyrights, trade secret rights, rights with respect to mask works or other rights on such Inventions and/or any other inventions I have or may at any time assign to the Company in any and all countries. Such acts may include, but are not limited to, execution of documents and assistance or cooperation in legal proceedings. I hereby irrevocably designate and appoint the Company and its duly authorized officers and agents and attorneys-in-fact to act for and in my behalf and instead of me, to execute and file any applications or related filings and to do all other lawfully permitted acts to further the prosecution and issuance of patents, copyrights, trade secret rights with respect to mask works or other rights thereon with the same legal force and effect as if executed by me.
- (f) I attach hereto a complete list of all Inventions or improvements to which I claim ownership and that I desire to remove from the operation of this Agreement, and I covenant that such list is complete. If no such list is attached to this Agreement, I represent that I have no such Inventions and improvements at the time of signing this Agreement.
- (g) I represent that my performance of all the terms of this Agreement will not breach any agreement or obligation to keep in confidence proprietary information acquired by me in confidence or in trust prior to my employment by the Company. My signature below represents that I have not and will not bring with me any proprietary information of other parties including, but not limited to, source code. I understand that Avant! does not require, request or condone unauthorized copying or use of computer software. I understand and accept that illicit or unauthorized copying of software, and the use of such software, whatever the sources, are expressly forbidden by Avant!

- 3. In consideration of the foregoing, the Company agrees that it will not request as part of my employment that I divulge or make use of confidential information of any of my former employers that has commercial value to the business of the former employer who developed such information.
- 4. I will not, for a period of twelve (12) months from the date of this Agreement, directly or indirectly manage, operate, control, be employed by, provide services for or be connected in any manner with, whether as a consultant, contractor, advisor, principal, partner, stockholder, officer, employee or agent, or in any relationship or capacity, any business or entity that is in the business of providing solutions to very deep-submicron integrated circuit design in competition with the products or services being created, developed, manufactured, marketed, distributed, hosted or sold by the Company, unless this Agreement be amended, modified, or superseded by written instrument executed by both parties. The provisions of this Section 4 are in addition to, and not in replacement of, any conditions set forth in any employment agreement entered into between me and the Company.
- 5. This Agreement shall be effective as of the first day of my employment by the Company, and shall be binding upon me, my heirs, executors, assigns, and administrators and shall inure to the benefit of the Company, it subsidiaries, successors and assigns.

Dated:	, 2000		
		Employee Signature	
		Print Name	-

EXHIBIT A

Avant! Corporation 46871 Bayside Pkwy. Fremont, CA 94538

1.	The following is a complete list of all inventions or improvements relevant to the subject matter of my employment by Avant! Corporation (the "Company") that have been made or conceived or first reduced to practice by me alone or jointly with others prior to my employment by the Company that I desire to remove from the operation of the Company's Proprietary Information and Inventions Agreement.		
		No inventions or improvements.	
		See below:	
		Additional sheets attached.	
2.	I propose to l employer:	bring to my employment the following materials and documents of a former	
		No materials or documents.	
		See below.	
		Employee	

Exhibit D

Form of Opinion of Buyer's Counsel

Exhibit E

Registration Rights

EXHIBIT E

REGISTRATION RIGHTS

The following registration rights terms are those referenced by Section 6.10(j) of the Agreement of Merger dated December 2, 1999 (the "Merger Agreement") between Avant! Corporation, a Delaware corporation (the "Buyer"), AC Acquisition Corp., an Oregon corporation and wholly owned subsidiary of the Buyer, and Analogy, Inc., an Oregon corporation (the "Company"). All terms used herein but not defined shall have the meanings ascribed to them in the Merger Agreement.

- Amendment. These terms and the rights and obligations included herein may hereafter be 1. amended by the written consent of the Company and the Buyer.
- 2. Registration Rights. The Company covenants and agrees as follows:
- 2.1 **Definitions.** As used herein, the following terms shall have the following respective meanings:
- (a) "Securities Act" shall mean the Securities Act of 1933, as amended, or any similar federal statute and the rules and regulations of the SEC thereunder, all as the same may be in effect at that time;
- The terms "Register", "Registered" and "Registration" refer to a (b) registration effected by preparing and filing a registration statement or similar document in compliance with the Securities Act and the declaration or ordering of effectiveness of such registration statement or document;
- "Registrable Securities" shall mean (i) any and all Common Stock of the Company issuable or issued upon exercise of the Option and (ii) any Common Stock of the Company issued as (or issuable upon the conversion or exercise of any warrant, right or other security which is issued as) a dividend or other distribution with respect to, or in exchange for or in replacement of, such Common Stock of the Company, excluding in all cases, however, any (x) Registrable Securities sold by a person in a transaction in which his rights under this Section 2 are not assigned and (y) shares that have been sold through a broker, dealer or underwriter in a public distribution or a public securities transaction, whether in a registered offering, pursuant to Rule 144 or otherwise. In addition, shares shall cease to be Registrable Securities hereunder at such time as all Registrable Securities held by a Holder can be sold in any three-month period without registration in compliance with Rule 144 of the Securities Act, including Rule 144(k);
- "Holder" shall mean any person who holds outstanding Registrable Securities which have not been sold to the public, but only if such person is the Buyer or an assignee or transferee thereof in accordance with Section 2.11 hereof;

1

PATENT REEL: 011820 FRAME: 0354

VWAP.TPN

- (e) "Registration Expenses" shall mean all expenses incurred by the Company in complying with Sections 2.2 and 2.3 hereof, including, without limitation, all registration, qualification and filing fees, printing expenses, escrow fees, fees and disbursements of counsel for the Company, blue sky fees and expenses, and the expense of any special audits incident to or required by any such registration (but excluding the compensation of regular employees of the Company which shall be paid in any event by the Company);
- (f) "Selling Expenses" shall mean all underwriting discounts, selling commissions and stock transfer taxes applicable to the securities registered by any Holder and all fees and disbursements of counsel for any Holder; and
- (g) "Form S-3" means such form under the Securities Act as in effect on the date hereof or any registration form under the Securities Act subsequently adopted by the SEC which permits inclusion or incorporation of substantial information by reference to other documents filed by the Company with the SEC.

2.2 Request for Registration.

- (a) If the Company shall receive at any time before the sixth (6th) anniversary of the date of the Merger Agreement a written request from a Holder that the Company file a registration statement under the Securities Act covering Registrable Securities with an anticipated aggregate offering price, net of underwriting discounts and commissions, of at least \$1,000,000, then the Company shall, within ten (10) days of the receipt thereof, give written notice of such request to all Holders and shall, as soon as practicable and subject to the limitations of Section 2.2(b), use its best efforts to effect the registration under the Securities Act of all Registrable Securities which the Holders request to be registered as soon as practicable.
- If the Holders intend to distribute the Registrable Securities covered by their (b) request by means of an underwriting, they shall so advise the Company as a part of their request made pursuant to this Section 2.2 and the Company shall include such information in the written notice referred to in Section 2.2(a). In such event, the right of any Holder to include his Registrable Securities in such registration shall be conditioned upon such Holder's participation in such underwriting and the inclusion of such Holder's Registrable Securities in the underwriting (unless otherwise mutually agreed by a majority in interest of the Holders and the Company) to the extent provided herein. All Holders proposing to distribute their securities through such underwriting shall (together with the Company as provided in Section 2.4(e)) enter into an underwriting agreement in customary form with the underwriter or underwriters selected for such underwriting by a majority in interest of the Holders and reasonably acceptable to the Company. Notwithstanding any other provision of this Section 2.2, if the underwriter advises the Holders and the Company in writing that marketing factors require a limitation of the number of shares to be underwritten, then the number of shares of Registrable Securities that may be included in the underwriting shall be allocated among all Holders thereof in proportion (as nearly as practicable) to the amount of Registrable Securities of the Company owned by each Holder.
- (c) The Company is obligated to effect only two (2) such registrations pursuant to this Section 2.2 The term "effect" in the previous sentence shall mean that, in addition to the

2

obligations set forth in Section 2.4, the applicable registration statement shall have been declared effective by the SEC.

- (d) Notwithstanding the foregoing, if the Company shall furnish to the Holders a certificate signed by the President of the Company stating that in the good faith judgment of the Board of Directors of the Company, it would be seriously detrimental to the Company and its shareholders for such registration statement to be filed and it is therefore essential to defer the filing of such registration statement, the Company shall have the right to defer such filing for a period of not more than ninety (90) days after receipt of the request for registration of the Holder; provided, however, that the Company may not utilize this right more than once in any twelve (12) month period.
- (e) The Company shall not be required to effect a Registration Statement requested pursuant to this Section 2.2 during the period starting forty-five (45) days prior to the Company's good faith estimated date of filing of, and ending on the date ninety (90) days following the effective date of, a Registration Statement initiated by the Company provided the Company is at all times diligently pursuing such registration at all times during such period.
- 2.3 Company Registration. If (but without any obligation to do so) the Company proposes to register (including for this purpose a registration effected by the Company for shareholders other than the Holders) any of its stock or other securities under the Securities Act in connection with the public offering of such securities solely for cash (other than a registration relating solely to the sale of securities to employees of the Company pursuant to a stock option, stock purchase or similar plan, or a registration relating to a Rule 145 transaction or a registration on any form which does not include substantially the same information as would be required to be included in a registration statement covering the sale of the Registrable Securities) the Company shall, each such time, promptly give each Holder written notice of such registration. Upon the written request of each Holder given within twenty (20) days after the giving of such notice, the Company shall, subject to the provisions of Section 2.7, cause to be registered under the Securities Act all of the Registrable Securities that each such Holder has requested to be registered.
- 2.4 **Obligations of the Company.** Whenever required under this Section 2 to effect the registration of any Registrable Securities, the Company shall, as expeditiously and as reasonably possible:
- (a) Prepare and file with the SEC a registration statement with respect to such Registrable Securities and use its best efforts to cause such registration statement to become effective and, upon the request of the Holders of a majority of the Registrable Securities registered thereunder, keep such registration statement effective for up to one hundred twenty (120) days.
- (b) Prepare and file with the SEC such amendments and supplements to such registration statement and the prospectus used in connection with such registration statement as may be necessary to comply with the provisions of the Securities Act with respect to the disposition of all securities covered by such registration statement.
- (c) Furnish to the Holders such numbers of copies of a prospectus, including a preliminary prospectus, in conformity with the requirements of the Securities Act, and such other

documents as they may reasonably request in order to facilitate the disposition of Registrable Securities owned by them.

- (d) Use its best efforts to register and qualify the securities covered by such registration statement under such other securities or Blue Sky laws of such jurisdictions as shall be reasonably requested by the Holders, provided that the Company shall not be required in connection therewith or as a condition thereto to qualify to do business or to file a general consent to service of process in any such states or jurisdiction.
- (e) In the event of any underwritten public offering, enter into and perform its obligations under an underwriting agreement, in usual and customary form, with the managing underwriter of such offering. Each Holder participating in such underwriting shall also enter into and perform its obligations under such an agreement.
- 2.5 Furnish Information. It shall be a condition precedent to the obligations of the Company to take any action pursuant to this Section 2 that the selling Holders shall furnish to the Company such information regarding themselves, the Registrable Securities held by them, and the intended method of disposition of such securities as shall be required to effect the registration of their Registrable Securities.
- 2.6 Expenses of Registration. All Registration Expenses incurred in connection with any registration, filing, qualification or compliance pursuant to this Section 2 shall be borne by the Company; provided, however, that the Company shall not be required to pay for any expenses of any registration proceeding begun pursuant to Section 2.2 if the registration request is subsequently withdrawn at the request of the Holders of a majority of the Registrable Securities to be registered, unless the Holders of a majority of the Registrable Securities agree to forfeit their right to one demand registration pursuant to Section 2.2; provided further, however that if at the time of such withdrawal, the Holders have learned of a material adverse change in the condition, business, or prospects of the Company from that known to the Holders at the time of their request, then the Holders shall not be required to pay any of such expenses and shall retain their rights pursuant to Section 2.2. Unless otherwise stated, all Selling Expenses relating to securities registered by the Holders shall be borne by the holders of such securities pro rata on the basis of the number of shares so registered.
- underwriting of shares, the Company shall not be required under Section 2.3 to include any of the Holders' Registrable Securities in such underwriting unless they accept the terms of the underwriting as agreed upon between the Company and the underwriters selected by it, and then only in such quantity as will not, in the opinion of the underwriters, jeopardize the success of the offering by the Company or the Company's shareholders demanding such registration. If the total amount of Registrable Securities that all selling Holders of the Company request to be included in such offering exceeds (when combined with the securities being offered by the Company or its shareholders demanding such registration) the amount of securities that the underwriters reasonably believe compatible with the success of the offering, then the Company shall be required to include in the offering only that number of such Registrable Securities which the underwriters believe will not offering only that number of such Registrable Securities so included to be apportioned pro rata among the selling Holders according to the total amount of Registrable Securities owned by each selling

4

Holder or in such other proportions as shall mutually be agreed to by such selling Holders), but in no event shall the amount of securities of the selling Holders included in an offering by the Company of its shares be reduced below twenty percent (20%) of the total amount of securities included in such offering.

- 2.8 **Delay of Registration.** No Holder shall have any right to obtain or seek an injunction restraining or otherwise delaying any such registration as the result of any controversy that might arise with respect to the interpretation or implementation of this Section 2.
- 2.9 Indemnification. In the event any Registrable Securities are included in a registration statement under this Section 2:
- (a) To the extent permitted by law, the Company will indemnify and hold harmless each Holder, the officers and directors of each Holder, any underwriter (as defined in the Securities Act) for such Holder and each person, if any who controls such Holder or underwriter within the meaning of the Securities Act or Exchange Act, against any losses, claims, damages, or liabilities (joint or several) to which they may become subject under the Securities Act, the Exchange Act or other federal or state law, insofar as such losses, claims, damages, or liabilities (or actions in respect thereof) arise out of or are based upon any of the following statements, omissions or violations (collectively a "Violation"): (i) any untrue statement or alleged untrue statement of a material fact contained in such registration statement including any preliminary prospectus or final prospectus contained therein or any amendments or supplements thereto; (ii) the omission or alleged omission to state therein a material fact required to be stated therein, or necessary to make the statements therein not misleading; or (iii) any violation or alleged violation by the Company of the Securities Act, the Exchange Act, any state securities law or any rule or regulation promulgated under the Securities Act, the Exchange Act or any state securities law; and the Company will reimburse each such Holder, officer or director, underwriter or controlling person for any legal or other expenses reasonably incurred by them in connection with investigating or defending any such loss, claim, damage, liability, or action, provided, however, that the indemnity agreement contained in this Section 2.9(a) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability, or action if such settlement is effected without the consent of the Company (which consent shall not be unreasonably withheld), nor shall the Company be liable in any such case for any such loss, claim, damage, liability, or action to the extent that it arises out of or is based upon a Violation which occurs in reliance upon and in conformity with written information furnished expressly for use in connection with such registration by any such Holder, underwriter or controlling person. The foregoing indemnity agreement of the Company is subject to the condition that, insofar as it relates to any Violation made in a preliminary prospectus but eliminated or remedied in the amended prospectus on file with the SEC at the time the registration statement in question becomes effective or the amended prospectus filed with the SEC pursuant to SEC Rule 424(b) (the "Final Prospectus"), such indemnity agreement shall not inure to the benefit of any person if a copy of the Final Prospectus was furnished to the indemnified party and was not furnished to the person asserting the loss, liability, claim or damage at or prior to the time such action is required by the Act.
- (b) To the extent permitted by law, each selling Holder will indemnify and hold harmless the Company, each of its directors, each of its officers who have signed the registration statement, each person, if any, who controls the Company within the meaning of the Securities Act, any underwriter (within the meaning of the Securities Act) for the Company or such other Holders,

5

VWAP.TPN

any person who controls such underwriter, and any other Holder selling securities in such registration statement or any of its directors or officers or any person who controls such Holder, against any losses, claims, damages, or liabilities (joint or several) to which the Company or any such director, officer, controlling person, or underwriter or controlling person, or other such Holder or director, officer or controlling person may become subject, under the Securities Act, the Exchange Act or other federal or state law insofar as such losses, claims damages, or liabilities (or actions in respect thereto) arise out of or are based upon any Violation, in each case to the extent (and only to the extent) that such Violation occurs in reliance upon and in conformity with written information furnished by such Holder expressly for use in connection with such registration, and each such Holder will reimburse any legal or other expenses reasonably incurred by the Company or any such director, officer, controlling person, underwriter or controlling person, other Holder, officer, director, or controlling person in connection with investigating or defending any such loss, claim damage. liability, or action, provided, however, that the indemnity agreement contained in this Section 2.9(b) shall not apply to amounts paid in settlement of any such loss, claim, damage, liability or action if such settlement is effected without the consent of the Holder, which consent shall not be unreasonably withheld, and, provided further that in no event shall any selling Holder's liability hereunder exceed the gross proceeds actually received by such Holder in respect of the sale of such Holder's shares in such offering.

- Promptly after receipt by an indemnified party under this Section 2.9 of (c) notice of the commencement of any action (including any governmental action), such indemnified party will, if a claim in respect thereof is to be made against any indemnifying party under this Section 2.9, notify the indemnifying party in writing of the commencement thereof, and the indemnifying party shall have the right to participate in, and, to the extent the indemnifying party so desires, jointly with any other indemnifying party similarly noticed, to assume the defense thereof with counsel mutually satisfactory to the parties; provided, however, that an indemnified party shall have the right to retain its own counsel, with the fees and expenses to be paid by the indemnifying party, if representation of such indemnified party by the counsel retained by the indemnifying party would be inappropriate due to actual or potential differing interests between such indemnified party and any other party represented by such counsel in such proceeding. The failure to notify an indemnifying party within a reasonable time of the commencement of any such action, if prejudicial to its ability to defend such action, shall relieve such indemnifying party of any liability to the indemnified party under this Section 2.9, but the omission so to notify the indemnifying party will not relieve it of any liability that it may have to any indemnified party otherwise than under this Section 2.9.
- 2.10 Reports Under Exchange Act. With a view to making available to the Holders the benefits of Rule 144 promulgated under the Securities Act and any other rule or regulation of the SEC that may at any time permit a Holder to sell securities of the Company to the public without registration or pursuant to a registration on Form S-3, the Company agrees to:
- (a) make and keep public information available as those terms are understood and defined in SEC Rule 144, at all times that Registrable Securities remain outstanding;
- (b) take such action as is necessary to enable the Holders to utilize Form S-3 for the sale of their Registrable Securities;

- (c) file with the SEC in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act; and
- (d) furnish to any Holder, so long as the Holder owns any Registrable Securities, forthwith upon request: (i) a written statement by the Company that it has complied with the reporting requirements of SEC Rule 144 (at any time after ninety (90) days after the effective date of the first registration statement filed by the Company), the Securities Act and the Exchange Act (at any time after it has become subject to such reporting requirements), or that it qualifies as a registrant whose securities may be resold pursuant to Form S-3 (at any time after it so qualifies); (ii) a copy of the most recent annual or quarterly report of the Company and such other reports and documents so filed by the Company; and (iii) such other information as may be reasonably requested in availing any Holder of any rule or regulation of the SEC which permits the selling of any such securities without registration or pursuant to such form.
- 2.11 Assignment of Registration Rights. The rights to cause the Company to register Registrable Securities pursuant to this Section 2 may be assigned by a Holder to a transferee or assignee, provided the Company is, within a reasonable time after such transfer, furnished with written notice of the name and address of such transferee or assignee and the securities with respect to which such registration rights are being assigned; and provided, further, that such assignment shall be effective only if immediately following such transfer the further disposition of such securities by the transferee or assignee is restricted under the Securities Act.
- 2.12 Limitations on Subsequent Registration Rights. From and after the date of the Merger Agreement, the Company shall not, without the prior written consent of the Holders of a majority of the outstanding Registrable Securities, enter into any agreement with any holder or prospective holder of any securities of the Company which would allow such holder or prospective holder: (a) to include such securities in any registration filed under Section 2.2 hereof, unless under the terms of such agreement such holder or prospective holder may include such securities in any such registration only to the extent that the inclusion of his securities will not reduce the amount of the Registrable Securities of the Holders which is included; or (b) to make a demand registration which could result in such registration statement being declared effective prior to the earlier of the date set forth in Section 2.2(a) or within one hundred twenty (120) days of the effective date of any registration effected pursuant to Section 2.2.

7

VWAP.TPN

AMENDMENT NO. 1 TO AGREEMENT OF MERGER

This Amendment No. 1 to Agreement of Merger (this "Amendment No. 1") is made as of March 15, 2000 by and among Avant! Corporation, a Delaware corporation (the "Buyer"), AC Acquisition Corp., an Oregon corporation ("Buyer Subsidiary") and Analogy, Inc., an Oregon corporation (the "Company").

WHEREAS, the Buyer, Buyer Subsidiary and the Company are parties to the Agreement of Merger dated as of December 2, 1999 (the "Merger Agreement"); and

WHEREAS, the Buyer, Buyer Subsidiary and the Company desire to amend the Merger Agreement in the manner contemplated hereby;

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements contained hereinafter set forth, the parties hereto agree as follows:

ARTICLE I

DEFINITIONS; ETC.

- 1.1 <u>Definitions</u>. Capitalized terms used but not defined herein shall have the respective meanings ascribed to them in the Merger Agreement.
- 1.2 Effect on the Merger Agreement. Except as expressly provided in this Amendment No. 1, all of the terms, conditions, restrictions and other provisions contained in the Merger Agreement shall remain in full force and effect. All references to the "Agreement" herein and in the Merger Agreement shall refer to the Merger Agreement as amended by this Amendment No. 1.

ARTICLE II

AMENDMENTS

- 2.1 <u>Amendment of Section 4.16</u>. Section 4.16 of the Merger Agreement is hereby amended in its entirety to read as follows:
- "4.16 Cancellation of Company 401(k) Plan. The Company shall have taken all required actions, if any, in order to cancel its 401(k) Plan effective as of a time mutually agreeable between the Company and the Buyer. The Company has provided the Buyer with a copy of the most recent determination letter issued by the Internal Revenue Service with regard to the Plan. The parties acknowledge and agree that the Company will cancel its 401(k) Plan as soon as practicable after the Closing and pay any matching funds due to the accounts of the 401(k) Plan participants (in accordance with the amounts accrued in the Financial Statements) at that time. The Company's 401(k) Plan shall remain in effect to the extent necessary to permit the rollover of all funds held by the Plan's participants."

ARTICLE III

CERTAIN REPRESENTATIONS

3.1 Each party hereto hereby represents and warrants to each other party that (i) it has full power and authority to execute and deliver this instrument and to consummate the transactions contemplated hereby and (ii) this instrument has been duly executed and delivered by such party and, assuming this instrument constitutes a valid and binding obligation of each other party, this instrument constitutes a valid and binding agreement of such party, enforceable against such party in accordance with its terms.

ARTICLE IV

GENERAL PROVISIONS

- 4.1 <u>Interpretation</u>. The headings contained in this Amendment No. 1 are for reference purposes only and shall not affect in any way the meaning or interpretation of this Amendment No. 1.
- 4.2 <u>Counterparts</u>. This Amendment No. 1 may be executed in one or more counterparts, each of which will be deemed to be an original copy of this Amendment No. 1 and all of which, when taken together, will be deemed to constitute one and the same agreement. Delivery of a signed counterpart by facsimile transmission will constitute a party's due execution and delivery of this Amendment No. 1.

[END OF TEXT. SIGNATURE PAGES FOLLOW.]

2

[Signature Page to Amendment No. 1 to Merger Agreement]

IN WITNESS WHEREOF, each of the undersigned has caused this Amendment No. 1 to Margar Agreement to be executed as of the date first written above.

	BUYER:
	AVANTI CORPORATION
X	By: G.C. St. Name: Title:
	BUYER SUBSIDIARY:
	AC ACQUISITION CORP.
X	By: E. C. R. Name: Title:
	COMPANY:
•	Analogy, Inc.
•	The same

Name: Title:

[Signature Page to Amendment No. 1 to Merger Agreement]

IN WITNESS WHEREOF, each of the undersigned has caused this Amendment No. 1 to Merger Agreement to be executed as of the date first written above.

BUYER:
AVANT! CORPORATION
By:
Name:
Title:
BUYER SUBSIDIARY:
AC ACQUISITION CORP.
By:
Name:
Title:
COMPANY:
ANALOGY, INC.
By:
Name:
Title:





Models developed pursuant to contract modeling arrangements are retained by the contracting party. Models have also been provided to the Company by universities and certain customers. These outside parties also retain use rights. In each of these cases, the Company has complete and unlimited rights to commercially distribute such models.

Symmetry's Intellectual Property:

- One registered trademark: MODPEX
- software source code associated with the Company's MODPEX software product,
 SymSpice, TFT models, and products under development
- software source code for the model library known as SymLib

With respect to the subset of Intangible Property that is the core technology of Symmetry, namely its proprietary modeling process, all of the model library known as SymLib, the source code for the TFT models included within SymSpice source code and the source code for the modeling software known as MODPEX (collectively, the "Symmetry Technology"), Symmetry is the sole and exclusive owner of all right, title and interest in and to the Symmetry Technology, free and clear of all liens, security interests, charges, encumbrances, or other adverse claims.

60488/4/TVW/052859-0047

ANALOGY, INC. PATENT STATUS (Updated October 27, 1999)

A. ANALOGY HOLDS THE FOLLOWING PATENTS:

- U.S. Patent No. 4,868,770, dated September 19, 1989 and Canadian Patent No. 1 323930. Simulation Results Enhancement Method and System. (Parameter Extraction) A method and system for extracting information from a simulation that was not explicitly obtained during simulation from a simulations results database and a modeling subsystem (such as a HDL).
- U.S. Patent No. 4,985,860, dated January 15, 1991 and Canadian Patent No. 13 19989 dated July 6, 1993. Mixed-Mode Simulator Interface. (Calaveras

 Algorithm). A system used to synchronize analog and digital simulation algorithms at non-regular intervals to perform simulation of a single, mixed-mode system.
- 3. U.S. Patent No. 5,046,024, dated September 3, 1991. <u>Vector Calculator Apparatus for Graphic Waveform Manipulation</u>. (Waveform Calculator) A system that simulates the appearance and operation of a calculator providing a user interface to graphic waveform data base software.
- 4. U.S. Patent No. 5,199,103, dated March 30, 1993. <u>Vector- Calculator Apparatus for- Graphic Waveform Manipulation</u>. (Waveform Calculator II) These claims were separated, amended, and filed on April 23, 1991 as a continuation application of the original Vector Calculator Apparatus for Graphic Waveform Manipulation. These claims were ultimately allowed. The claims allow the waveform calculator to work without requiring the graphical display.
- U.S. Patent No. 5,404,319, dated April 4, 1995. <u>Translation of Behavioral Modeling Properties into an Analog Hardware Description Language</u>. This technology is the basis of Analogy's graphical modeling and analog model synthesis products.
- 6. U.S. Patent No. 5,404,319, dated August 20, 1996. Analysis Mechanism for the System Performance Simulator. This technology covers aspects of Design for Reliability (DFR) tools such as stress, sensitivity, FMEA, and others.

B. PENDING ANALOGY PATENT APPLICATIONS/MATTERS:

- 1. An application entitled <u>Component-Based Analog and Mixed Signal Model Development</u> was filed in February, 1996. This comprises much of the base Model Architect technology developed under the NIST/ATP grant by the Software Productivity R&D group. This patent has been split into the following:
 - 1467-2 U.S. Case A Notice of allowance is expected following the last-filed Amendment 1467-5 U.S. Divisional Prior to issuance of the -2 case, divisional 5 has been filed, 1467-7/8 U.S. Additional Cases Prior to issuance of the -2 case, file continuance applications may be filed entitled: (a) Component-Based Analog and Mixed Signal Simulation Model With Mixed Signal Finite State Machine; and (b) Component-Based Analog and Mixed-Signal Simulation-Model Experiment Editor. The 1467-2 patent should issue in 1999. Based on a reduced set of claims, an EPO application has been filed designating at UK, France and Germany.
- 2.

 1467-13/14/15 Provisional patent applications regarding aspects of the Verias™ Simulator were filed on June 18, 1999. The provisionals are entitled: (1) Integral Simulation Time For Mixed-Signal Simulation; (2) Dynamic Association of Equations to Unknowns During Simulation of Systems

 Described by Hardware Description Languages; and (3) Classification of the Unknowns in a System of Simultaneous Equations Described by a Hardware Description Language.

CONFIDENTIAL AND PROPRIETARY TO ANALOGY. INC.

SCHEDULE 2.21(b)

TRADEMARK	JURISDICTION
AHDL®	Germany
AHDL®	UK
ANALOGY®	U.S.
ANALOGY®	Japan
ANALOGY® and Device	ÚK
ANALOGY®	France
ANALOGY® and Design	Germany
ANALOGY®	East Germany
ANALOGY®	Singapore
ANALOGY®	Sweden
ANALOGY HDL®	U.S.
ANALOGY HDL®	Germany
ANALOGY HDL®	Japan
ANALOGY HDL®	UK
CALAVERAS®	Japan
CALAVERAS®	U.S.
DESIGNSTAR®	U.S.
FRAMEWAY®	France
FRAMEWAY®	Germany
FRAMEWAY®	UK
FRAMEWAY®	U.S.
HYPERMODEL®	US
HYPERMODEL®	UK
HYPERMODEL®	Germany
HYPERMODEL®	East Germany
INSPECS®	U.S
INSPECS®	France
INSPECS®	Germany
INSPECS®	UK
MAST®	US
POWEREXPRESS®	France
POWEREXPRESS®	Germany
POWEREXPRESS®	U.S.
POWEREXPRESS®	UK
RISING STAR®	U.S.
SABER® *	Worldwide license
WAVECALC®	France
WAVECALC®	Germany
WAVECALC®	UK
WAVECALC®	U.S.

^{*}SABER is licensed to Analogy, Inc. By American Airlines pursuant to a negotiated license agreement.

PENDING TRADEMARKS	
TESTIFY™	EU Community TM
MODELARCHITECT™	EU Community TM
MICROMODEL™	EU Community TM
CUT THE RISK	U.S.
GET IT RIGHT	
MAKE IT REAL™	
HYDRAULICEXPRESS™	U.S.
FASTMAST™	U.S
MODELARCHITECT™	U.S.
MODELEXPRESS™	U.S.
MICROMODEL™	U.S.
TELECOMEXPRESS™	U.S.
TESTIFY™	U.S.
THEDHL™	U.S.
VARIAS VTK	U.S.

Detroduce B. HI(D) - COPITIZITE INCEDITATIONS

COPYRIGHT DESCRIPTION REGISTRATION NUMBER

DATE OF ISSUANCE

Saber Tutorial Manual, User's Guide & Reference Manual

Beta Version Release 1.1 Reg. No. TX 2,293,702 Reg. No. TX 2,326,664 3/21/88 4/21/88

Saber Tutorial Manual, User's Guide, Reference Manual,

MAST User's Guide & Analogy Library

Release 2.0

Reg. No. TX 2,293,706

3/21/88

Saber Tutorial Manual, User's Guide & Analogy Library

Release 2.0

Reg. No. TX 2,394,720

6/14/88

Saber Tutorial Manual, Guide to Writing Templates,

Template Library Manual, Installiation and System-specific information

Release 2.1

Reg. No. TX 2,451,079

7/27/88

Saber Tutorial Manuals, Vols. R1, R2, R3, U1, U2, C1

Release 3.0

Reg. No. TX 3,002,185

2/12/91

SABER

Release 3.0

Reg. No. TX 3,021,944

2/11/91

SCHEDULE B - PENDING COPYRIGHT APPLICATIONS

FIRST DATE

COPYRIGHT DESCRIPTION APPLICATION DATE OF NUMBER FILING

DATE OF CREATION

OF PUBLIC
DISTRIBUTION

None

SCHEDULE C - UNREGISTERED COPYRIGHTS (Where No Copyright Application is Pending)

SABER Release 3.4 and newer versions of the software.

h:\pblc\frm\fctrdcs\ipagrmt.doc - R\$2995

Schedule 2.21(c)(i).

ANALOGY:

VALUE-ADDED RESELLERS:

Analogy Value-Added Reseller (VAR) Agreement; 6/15/97; Analogy, Inc. and Exco, Inc.

Analogy Sales Representation Agreement; 3/1/96; Analogy, Inc. and Comtel Corporation DBA Comtel EDA Co.

Analogy, Inc. VAR/OEM Agreement; 4/1/98 Analogy, Inc. and Microcasm Technologies, Inc.

DISTRIBUTORS:

INDIA

Analogy, Inc. Distributor Agreement; 10/1/94

Analogy, Inc. and CMR Design Automation (P) Ltd.

ISRAEL

Analogy UK Ltd. Sales Representation Agreement; 6/1/97 Analogy UK Ltd. And IES Electronic Agencies (1986) Ltd.

ITALY

Distributor Agreement; Eff. Date unknown - approx. 12/92 Analogy UK Ltd. And ACSIS S.R.L.

JAPAN

Analogy, Inc. International Product Distribution and Software License Agreement; 7/11/88; Analogy, Inc. and C ITOH & Co. LTD.

KOREA

Analogy, Inc. Distributor Agreement; 9/25/96;

Analogy, Inc. and JasonTech, Inc.

NETHERLANDS

Analogy UK Ltd. Sales Representation Agreement; 6/23/95;

Analogy UK Ltd. And The CAE Company (TCC)

60488/4/TVW/052859-0047

PEOPLES REPUBLIC OF CHINA

Analogy, Inc. Distributor Agreement; 7/1/98

Analogy, Inc. and Beijing Goldprize Technology co. Ltd.

Analogy, Inc. Distributor Agreement; 10/1/98;

Analogy, Inc. and HWA Create (China) Co, Ltd.

Analogy, Inc. Distributor Agreement; 11/1/98;

Analogy, Inc. and Synthesis Systems Design Ltd.

Analogy, Inc. Distributor Agreement; 6/17/99;

Analogy, Inc. and Beijing WangTong Hi-Tech Corp.

SINGAPORE

Analogy, Inc. Distributor Agreement; 8/1/99;

Analogy, Inc. and SPS-DA Private Limited

SPAIN

Analogy France SARL Sales Representative Agreement; 5/1/96;

Analogy France SARL and CVA 2000 S.L.

TAIWAN

*Analogy, Inc. Distributor Agreement; 9/1/99;

Analogy, Inc. and Cache Technology Inc.

*This is a newly-signed agreement that is provided in hard copy attached to this schedule 2.21(c)(i)

Analogy UK Ltd. Distributor Agreement; 3/18/94;

Analogy, Inc. and Dimension Computer Technology Co, Ltd.

Analogy, Inc. Distributor Agreement; 9/1/97;

Analogy, Inc. and TimeWave International Corp.

SYMMETRY:

Remarketing and Distribution License Agreement; 9/1/97; Symmetry Design Systems, Inc. and Interactive Image Technologies, Inc. (IIT)

Symmetry Design Systems, Inc. Manufacturer's Representative Agreement; 6/1/99;

Symmetry Design Systems, Inc. and Interface Technology

60488/4/TVW/052859-0047

JAPAN

Software Distribution Agreement; 11/1/92; Symmetry Design Systems, Inc. and Kanematsu Electronic Ltd. (Distributor) and Kanematsu USA Inc. (Exporter)

Symmetry Design Systems, Inc. Product Distribution Agreement; 12/1/94; Symmetry Design Systems, Inc. and QuadTool Marketing, Inc.

Veribest Incorporated Joint Reference-Selling Agreement; 1/12/98; Symmetry Design Systems, Inc. and VERIBEST, Inc.

CHINA and HONG KONG

Analogy, Inc. Distributor Agreement; 6/99;

Analogy, Inc. and Beijing WangTong Hi-Tech Corp.

60488/4/TVW/052859-0047

Schedule 2.21(c)(ii).

Source Code License and Binary Code Distribution License Agreement between Phase III Logic, Inc. and Analogy, Inc., dated June 7, 1991, as amended including Addendum No. 3 effective November 3, 1993.

Tauri OEM Software Agreement between FTL Systems, Inc. and Analogy, Inc., dated May 21, 1999.

Remarketing and Distribution License Agreement between Mathworks, Inc. and Analogy, Inc., dated January 16, 1994.

"Freeware:"

Alfalfa Software Inc. Message Catalog System Baker & Newton EDIF lib Cornell University Tel-DP Free Software Foundation CVS Free Software Foundation **EMACS** Free Software Foundation Glib Free Software Foundation Gtar Free Software Foundation Perl TK table Hobbs Lehenbauer & Deikhans Tel X Anchor/Tel Lucent Technologies **Iwidgets** Lucent Technologies IMG lib Niitmans et al University of California, et al. Tel TK University of California flex University of California yacc

"Commercially Available" software

Microquill Software Publishing
Rogue Wave Software
Globetrotter Software, Inc.
Installshield Software Corporation

Smartheap
LA pack
FlexIm
Installshield

Master License Agreement between the Company and Texas Instruments Incorporated, effective May 7, 1996 (the "TI Agreement").

60488/4/TVW/052859-0047

2.21(f)—

Non-development personnel:
Giles Lewis—UK Probationary AE
Fredrick Johansson—Sweden AE

Schedule	2.21(g)(i).
----------	-------------

None.

60488/4/TVW/052859-0047

Schedule 2.21(g)(ii).

Standard Confidential Information Nondisclosure Agreement-see attached. Standard Confidential Information Exchange Agreement-see attached.

60488/4/TVW/052859-0047



Confidential Information Nondisclosure Agreement

	3
Agreement between: Analogy, Inc. (Analogy) And 9205 S.W. Gemini Drive Beaverton, Oregon 97008	Effective Date:(RECEIVER)
Subject Matter:	
Permitted Use:	
Receiver may receive certain data and other information of a confidential or proprietary nature from Analogy related to the Subject Matter. Analogy considers this information confidential but is willing to provide such information on a confidential basis under the following terms and conditions: 1. Definition. "Confidential Information" means any unpublished information owned or controlled by Analogy relating to technical, manufacturing, marketing, sales, or financial operations of Analogy and that is not generally disclosed by Analogy to the public, including, without limitation, proprietary processes and designs, trade secrets, know-how, inventions (whether or not patentable), concepts, drawings and data, research methods and results, proprietary software, product specifications, business, marketing and future product plans, pricing, cost and profit information, customer lists, supplier identities, whether disclosed orally, in writing, or by inspection. 2. Nondisclosure and Use. Receiver agrees: (a) to use Confidential Information only for the Permitted Use, (b) to retain in confidence all such Confidential Information, and (c) not to disclose any such Confidential Information to anyone, except employees of Receiver who are authorized to receive Confidential Information. Receiver shall, upon request by Analogy, provide Analogy with a list of all persons who have had access to Confidential Information from Receiver.	through no wrongful act of the receiving party; (d) is already known by the receiving party as evidenced by documentation bearing a date prior to the date of disclosure; or (e) is approved for release in writing by an authorized representative of the disclosing party. 4. Term. The obligations of Receiver under this Agreement shall continue for five (5) years after initial disclosure to Receiver. 5. Return of Confidential Information. Upon request by Analogy or on completion of the business dealings relating to the Confidential Information. Receiver shall promptly return to Analogy, or if requested by Analogy destroy all copies of the Confidential Information and provide written confirmation of this by an officer of Receiver. 6. Governing Law. This Agreement shall be governed by the laws of the State of Oregon. 7. Entire Agreement. This Agreement contains anytice agreement between the parties, and
ANALOGY, INC:	ACKNOWLEDGED AND AGREED RECEIVER: BY:
(Authorized Representative)	(Authorized Representative)
IVAME. (Type or Print)	(Type or Print)
TITLE	

Analogy, Inc. Confidential Information Exchange Agreement
Analogy, Inc. 9205 S.W. Gemint Drive, Beaverton, OR 97008. Phone (503) 626-9700
Analogy is a registered trademark at Analogy, Inc.

DATE: ______

40894



Confidential Information Exchange Agreement

Agreement between:		Effective Date:	
ANALOGY, INC. (ANALOGY) 9205 S.W. GEMINI DRIVE	AND		(COMPANY)
BEAVERTON, OREGON 97008			
Analogy Subject Matter:			
Permitted Use:			
Company Subject Matter:			
Permitted Use:			

Company and Analogy enter into this Agreement to establish a confidential relationship and to exchange confidential information which shall be protected by the receiving party from a disclosure or use that is not authorized by the disclosing party in writing.

Both parties, for their mutual benefit, are desirous that Company shall disclose to Analogy information that is related to the Company Subject Matter; that Analogy shall disclose to Company information that is related to the Analogy Subject Matter; and that such information shall be used only for the relevant Permitted Use.

The parties agree as follows:

- 1. Information that is disclosed by Analogy to Company or by Company to Analogy shall be disclosed free from any restrictions on its subsequent disclosure or use, unless such information is disclosed in confidence as confidential information or is the subject of patent or copyright protection.
- 2. Information shall be Confidential Information if it is disclosed in tangible form and marked as being "Confidential," "Proprietary," or by any other appropriate legend clearly indicating the confidential nature of the information. Information, it first disclosed in oral, visual or other intangible form, shall be Confidential Information if (a) it is identified by the disclosing party at the time of disclosure as being disclosed in confidence, and (b) it is reduced to tangible form and marked in accordance with this Section 2 and delivered to the receiving party within 30 days after the date of first disclosure.
 - 3. Confidential Information shall not be disclosed by the receiving party to others, except to employees of the receiving party having a need-to-know in order to carry out the purpose of this Agreement. Both parties represent that they protect their own confidential information from unauthorized use or disclosure, and further that they have agreements with their employees to so protect their own confidential information and the confidential information of others. The receiving party shall protect Confidential Information received from

disclosing party with reasonable care that is at least the same degree of care that it regularly employs to safeguard its own confidential information from an unauthorized use or disclosure.

- 4. The obligations of each party with respect to all Confidential Information received under this Agreement shall survive termination set forth in Section 14 and shall remain in effect for a period of five (5) years from the date of first receipt of such Confidential Information. The expiration of the above five (5) years shall not affect any rights of the parties with respect to patents, trademarks or copyrights.
- 5. The protection provided in this Agreement shall not apply to any portion of Confidential Information that: (a) is rightfully received by the receiving party from a third party who has a right to disclose such information; (b) is independently developed by the receiving party without use of the confidential information disclosed hereunder; (c) is or becomes publicly available through no wrongful act of the receiving party; (d) is already known by the receiving party as evidenced by documentation bearing a date prior to the date of disclosure; or (e) is approved for release in writing by an authorized representative of the disclosing party.
- 6. This Agreement shall not restrict the right of either party to disclose Confidential Information pursuant to a judicial order, but only to the extent so ordered, provided, however, that the party receiving such order shall notify the other party of such order in sufficient time to permit such other party to intervene in response to such order, and provided that the confidential or proprietary markings remain on the information disclosed.
 - 7. Both parties understand that the other party may have already developed or received from third parties information or material similar to that received from the other party hereunder, or in the future may internally develop or receive from third parties information or material similar to that received from the other party hereunder Accordingly, provided Confidential Information of the other party is not used in violation of this

Analogy, Inc. Confidential Information Exchange Agreement
Analogy, Inc. 9205 S W. Gemini Drive, Beaverton, OR 97008 Phone (503) 626-9700

Agreement, this Agreement shall not be construed as a representation or inference that either party has not or will not develop information, material, technology or products, for itself or for others, that is similar to material or information disclosed by the other party hereunder or that competes with products of the other party.

- 8. Except by explicit licensing agreements, no license to either party under patents, trademarks or copyrights is granted or implied by the disclosure of Confidential Information; however, a maximum of two copies of Confidential Information may be made by the receiving party without the written permission of the disclosing party.
- All Confidential Information and copies thereof shall remain the property of the disclosing All Confidential Information shall be destroyed or returned if requested by the disclosing party. Upon breach of any of the obligations of Section 3 by the receiving party and upon demand by the disclosing party, Confidential Information. and all copies thereof, shall be returned to the disclosing party within thirty days of such demand. Alternatively, Confidential Information, and all copies thereof, may be destroyed by the receiving party. Such destruction shall be proved by an affidavit signed by an officer to that effect, submitted to the disclosing party within thirty days of such destruction. However, the receiving party may retain in the files of its legal counsel, for archival purposes only, one copy of the Confidential Information. The rights and obligations of both parties under this Agreement with respect to such

returned or destroyed confidential information shall be determined by Section 4.

- 10. If one party breaches this Agreement, the fallure of the other party to enforce any right under this Agreement shall not be deemed a waiver of any right hereunder. The rights and remedies of the parties as set forth in this Agreement are not exclusive and are in addition to any other rights and remedies provided by law; additionally, the invalidity in whole or in part of any condition of this Agreement shall not affect the validity of any other condition hereof.
- This Agreement is deemed to be made under and shall be construed in accordance with the laws of the State of Oregon, exclusive of the conflict of laws provisions thereof, and constitutes the entire understanding between the parties hereto with respect to the subject matter of this Agreement.
- 12. This Agreement supersedes all previous understandings between the parties with respect to the subject matter of this Agreement.
- 13. No amendment or modification of this Agreement shall be valid or binding on the parties unless made in writing and signed on behalf of each of the parties by their respective duly authorized officers or representatives.
- 14. With the exception of the obligations set forth in Section 4, this Agreement shall terminate one year from the Effective Date, except that either party, upon thirty days written notice to the other party, may terminate this Agreement with respect to disclosures made thereafter.

ANALO	OGY, INC:	COMP	
BY:	(Authorized Representative)	BY:	(Authorized Representative)
NAME:	(Type or Print)	NAME:	(Type or Print)
TITLE		TITLE	
DATE		DATE	

CONDANY.

2.21(g)(iii)--

- Chrysler Corporation (Eff. Date 10/18/94)
 Chrysler Letter of Ethics Agreement
 Signed under protest that the letter was insufficient and only a general overview of products was recommended for disclosure.
- 2. General Motors Tech Center (May 15, 1997)

 Drafted for obligations of Analogy employee to not disclose confidential information of Chrysler obtained from being on-site and developing software to test GM models. No term on protection of confidential information.
- 3. Hitachi America, Ltd. (Eff. Date 3/1/99)
 Hitachi Nondisclosure and Ownership Agreement.
 Period of protection of Confidential Information is until info falls into one of the exceptions from the obligation to protect as confidential.
- 4. IBM Corporation (Eff. Date May 7, 1992)
 Amendment to SLA very narrowly defines what will comprise confidential information and what may be done with it. Allows exception for retained memory.
- 5. Intel Corporation (Eff. Date 6/17/99)
 Intel Confidential Information Transmittal Record & Agreement
 Disclosing party agrees not to assert claims against receiving party regarding
 confidential information for 60 months (5 years).
- 6. Microchip Technology Incorporated (Eff. Date 8/13/99)
 Confidentiality Agreement. Period to protect confidential information is until 8/13/2002.
- 7. Model Technology Incorporated (Eff. Date 12/16/97)
 Confidential Disclosure Agreement two year term.
 Unusual clause Analogy must provide MTI with notice one year before Analogy announces a stand-alone digital VHDL and or Verilog product.
- 8. Motorola Corporation Austin (Eff. Date 11/5/93)
 Nondisclosure Agreement regarding Motorola's MOS process technologies known as SSIM and collaboration on an interface with Saber. Analogy shall provide on-going assistance to Motorola on an as-needed basis to ensure the continued operation of the Motorola MOS SSIM model in Saber as long as Saber is a major design tool used within Motorola.
 - 9. Motorola, Inc. (Eff. Date 3/28/95)

Design Rule Exchange Agreement Subject matter: Motorola's SMARTMOS 2.5 Design Rules and related SPICE models. Period of protection is five years but the agreement is evergreen.

- National Semiconductor Corporation (Eff. Date 3/25/92)
 National Semiconductor Nondisclosure Agreement
 Obligation of protection of confidential information expires in ten years
 (3/25/2002). Subject: IMOB33 MOSFET model equations and all associated code and documentation
- Proprietary Information Disclosure Agreement for TPS Developers

 Agreement terminates upon either the completion of the termination of the TPS

 Developer Program. Subject: Test program sets for depot level testing of B-2

 Radar Hardware. No term of protection of confidential information perpetual?
- Sun Microsystems, Inc. (Eff. Date 9/15/97)
 Confidential Disclosure Agreement Disclosures to Sun
 Sun obligation to protect Analogy confidential information is seven years from date of disclosure.

Schedule 2.21(h).

See WorldwideMaintenanceReport1-see attached.

Also included:

Ford (on-site AE)
Delco (on-site AE)
EDS Software License Agreement
Intel Software License Agreement
Rockwell Collins Volume Purchase Agreement
Open Access Agreements (Schedule 2.15)

Long-term development order reflected on Purchase Order dated October 17, 1996 from Raytheon E-Systems.

60488/4/TVW/052859-0047

WorldWideMaintenanceReport 1 x1	v,
HUWIDEMAINTENANCER	₹ =
HUWIDEMAINTENANCER	ğ
III Wide Maint	ğ
III Wide Maint	Sec.
MadwideM	ξ
3	٤
Ĭ	\$
>	Ĭ
	>

Company Name	Contact Name	Conlact Phone	Address	Address2	Address3	City	State	Countr	Countr Posts	MaintExpDf
ארן שייטן	Lan Ferbain	91442 438405	3COM Cenire	Boundary Way		Heme: Henyslova HERT	340 MERT	68	HP2 3YU	December 31, 1999)
Author J Crown rook OFNMARK	Frede Blaabjery	- 45 96 35 92 5시	Dept of Electrical Energy Conversion	Pontoppidansiræde 101		DK 9220 Aalborg	Ģ,	š		March 31, 2002
Authing University DENMAHK	Strg Munk Nedsen	(459) 635-9242	Dept. of Electrical Energy Conversion	Portloppidanstrade 101		CASI OK-9220 Aalburg Casi	Ţ.	č		March 31, 2002
ABB Coxporate Ros. Hordelberg ABB Coxporate Research Ltd. ABB Curporate Rosearch SWEDEN	Georg Gutermuth Marco Sulur Kyell Pristrand	06221 596477 0041 56 486 8016 (2) 132-3000	CH CRC	Speyerer Str. 4 Im Segethol :	Positach 10 13 32	Hardelberg Baden-Daltwii S.721 78		9. C. D.	0 69003 CH-5405	December 31, 1999 August 31, 2000 March 31, 2001
ABB Industrie Turg	Adrian With Mka Mkkola	+41 56 299 3226 011 35 810 222 000	Industrieerinebe Marine Division	Plone 2 P O Box 185		Västerås Turgi FIN-00381		5 .	CH-5300	February 28, 2000 Appl 10, 2000
ABB Industry (VSD) FINLAND	Haina	(35) 810-2211	Vanable Speed Drives	Atomitie 5, P.O. Box 211		Helsinki FIN-00381		ũ		December 31 1999
Acutex Hittle Industries	Burzanowska Paul Christensen	(231) 894-3213	2001 Peach Street			Helsink: Whiehall	3	NS	49461	September 36, 2000
Adnan Incorporaled	Ron Wayden	(256) 963-8556	901 Explorer Boulevard	4 South		Huntsville	¥	S	35806	June 30, 2000
Actioniz	Shawn West	(412) 655-6705	1501 Lebanon Church Road			Pattaburgh	ď	Sn	2807 15236- 1491	December 31 1999
ADtranz Hermigsdurf	Jorg-Torsten Maar) 03302 69 2148	03302 69 2148		Am Rathenaupark		Hennigsdort		90	D-16761	October 31, 2000
ADiranz Schweiz Adiranz Sweden	Markus Jorg Jorge Man	+41 56 299 2254 -46 21 130299	Sleigstraffc			Turgi S-721 72		7 %	CH-5300	March 31: 2000 January 31, 2000
A Accepted Energy Voorhees, NJ	Invan Bhulla	(6096276) 100-2333		1007 Laurel Oak Road		Vasierās Voomees	3	ns	8043	Seolember 30, 2000
Arkertest ATE/SB-0011	*ASUKO		336-1,0azu-Owa	Mena-mura		Oura-Gun	GUNM		370.07	March 31, 2000
Advantest EL1(Instrument) SG-0057	YASUKO		1-5, shin-tone	Otone-cho		kıta-sanama	SAIT	٩	349-11	March 31, 2000
Aerospace Corporation	Brien Lenertz	(310) 336-1026	COSULE: El Segueldo Biva.		M4/988	El Segundo	₹		90245	August 31 2000
AEROSPATIALE MATRA AIRBUS	Enc CREQUY	(56) 193-7141	Eladricement de l'odouse	WB1 EUSY/SQIP	316 route de Bayonne	TOULOUSE CEDEX 03	MPY	Œ	31060	June 30, 2000
AEROSPATIALE MATRA AIRBUS	Jean-François	(56) 19 3-8698	Etablissement de Toulouse	ABTE/SY/SOMP	316, route de Bayonne	TOULOUSE	₩by	æ	31060	June 30, 2000
A.EROSPATIAL E MATRA AIRBUS	Luc LEROY	(56) 193-8518	Elabitsoment de Toutouse	A/BTE/SY/SQIP	316, route de Bayonne	TOULOUSE	ΝΡΥ	F.	31060	March 31 2000
AEROSPATIALE MATRA CCR	Julian BOULLIE	(14) 697-3008	Centre Corrmun de Racherche 5720 Smetana Dave	12, rue Pasteur	BP 76	SURESNES				December 31, 2000
Ag Criem Equipment Co., Inc.	Donald Neison	(707) 792-5708				Mirmetonka	₹ (September 30, 2000
Atcatel (DSC Communications)	Parick CATTEUX	(16) 963-1380	Etablissement de Marcoussis	Route de Nozay		MARCOUSSIS	5	Si er	94954	December 31, 1999
ALCATEL Submanne Networks	Herve MASSON	(16) 449-4521	0.00 O	Route de Villejust		NOZAY CEDEX	Õ			December 31 1999
Atenia Marcuni Systems, Stanmore UK		0181 420 3152		Warren Lane		Slanmore	MSEX	_	HA7 4LY	March 31, 2001
Aligad Muslim Univ.	G Hagupathy	(603228) 553-3341	162 Pembloke Road			4,000	;			March 31, 2000
Alegic Microsystems, inc. Caracha inc.	Patnck Herbert	(440) 646-3585	Rockwell Automation	I Allen Bradlay Drive		Mayfeld Helohis	Į č		3301	October 31, 2000
rieights Orf		(414) 382-2715	1201 S Second St							epienese, or con-
Alled Bradley Co. Inc. Miredonas Alled Signal Aerospace	Douglas Gulekinst (816) 997-4936	(816) 997-4936			2000 East 95th Sirest	Kansas City	<u> </u>	S. S.	53204 A 64131- D	August 31 2006 December 31 1999
AllestStandl Aeruspace	Frank Smith	1810) 997-4592			2000 East 95th Street	Kansas Catr	OM	e a	_	Open it seemen
	4	4520, 459,1000	1110) North Oracle Rd			-				בכפון וכן ואמן
AlledSignal Aerospace	Boone Tudsick	(HO56086) 000-2123		3333 Unit; Drive		Tuscon	¥ 8	SD SD		December 31 1999
Aliedalijas verospace (Crans.)								_	יס הקר אפן מר אפן	Septomber 30 გსეც

o D P

ır.
÷.
_
=
ኣ
Δ
•
Œ
•
۶
è
č
9
Ē
ã
ゞ
₹
Ũ
۶
₹
ĕ
ō
2
_

Company Name	Contact Name	Contact Phone	Address	Address 2	Address3	CHY	St at	Count	County Postal	NaintExpOt
AmedSignal Aerost.a. c. Ontana	Rob Fluman	14157986, 868-2903		3333 Unity Drive		Mississauga	Š	Š	15L 3SE	September 30, 2000
CANADA Alegon Fransonian morandoolis IN Ales (Loboka Alps) Car Electronics Div	Steve Dao YASUKO	(317) 242:4072	4/00 West 10th Street CAR ELECTRONICS DIVISION	FURUKAWA PLANT	6 3-36, Nakazało	Indianapolis Funkawa-city	N.Y.G	US 9.	46222 989 61	January 31, 2000 March 31, 2001
58-0325 Afris Automotive Aubum Hills Mi	Enc Yuzon	(248) 391 5054	Branch Plant	1500 Atlantic Blvd		Auburn Hills	Ē	NS	48326	March 31, 2000
AI STOM	Fredence	1381 455-6320	THANSPORT S.A	3. avenue des Trois Chênes		BELFORT CEDEX FNCM	EX FNCM	Œ	90002	March 31, 2000
	PIGNILLON DIGNOL DANLEY	(56) 253 4621	TRANSPORT	Rue du Docteur Guinier	8₽ 4	SEMEAC	¥₽	Œ	00959	November 30, 1999
ALSTON Belowin Fabroard	Raphadi MAFFE!	(327) 144-5811		Hue Cambier Dupret, 50-	BP 4211	MARCINELLE	N N	96	9031	December 31, 1999
ALSTON BEINGER TOWNEY.	Bill Garlick	01785-274660	PO Box 30	Confield Road		Stafford	STFF	89 6	ST17 4LN	March 31 2001
Alexan RTC Stafford UK	Calin Oates	01785 274652	PO Box 30	25 out 444 Determine		Stafford	T G	9 9	S137 4CN	March 31 2001
ALSTOM TAD SA	Dirk UHDE	(14) 945-7051				SAIN OUEN	<u>\$</u>	ŗ.	93404	December 31 1998
Aistom Traction, Preston UK Altair Engineering Inc., Troy, MI	Robin Howie Keith Drittler	01772 553280 (248) 614-2400	Chainel Way 1757 Mapielawn			Preston Troy	N I	GB US	PR1 8X1 48084: 4004	March 31, 2001 Aprv 30, 2000
Amencan Electronic Components Inc	James Rustman	(219264) 111-6179		P O Box 280	1010 North Main Street Elkhart	81 Elkhari	Z	sn	46515	December 31, 1999
100	Delega Thelea	(406) 582 8512	4699 Chaparrel Way			Bozeman	T.W	SO	59715	February 28 2000
Analog interfaces Bozeman, Mil	Town No. and	01705 492412	36 Brookside Road			Haven	HAMP		PO9 1.JR	March 31, 2000
Aporto Fire Detectors UK	Adrian Brown	01222 - 500700	Cypress Dave	St Mellons		Cardiff		6 B	CF3 0EG	December 31, 1999
Ascom Teleconnium and is On	Ben Mika	(630505) 040-4107	1620 Bond St			Naperville	=	Sn	60583	December 31, 1999
ASIC Designs, Inc.	Dick van Hees	+31 40 2303370	De Run 1110	5503 LA Veldhoven	1		,	ž '		June 30, 2000
Authorn University Auburn, AL	Mark Nekns	(334) 844-1609			200 Brown Hall	Authorn	₹	S S	36849 5201	June 30, 2000
	Manfred Zuber	0841 69 89568				Ingolstadt		30	D-85045	October 31, 2000
AUDI AG	Patrick CAMPIN	(13) 480-7412		41, boulevard de la Republique		CHATOU	10F	æ	78401	July 31 2000
AXODYN	Siephane	(13) 043-0300		150, av Joseph Kessel		VOISINS LE	JQ.	æ	09682	March 31, 2000
Becntel Bollis Latis	Vince Baramonte	(412) 476 5803		814 Pritsburgh. McKeesport Blvd		West Mifflin	P A	SO	15122	September 30, 2000
Becntel, Plant Apparatus Division	James Rockot	8008-629 (214)	Plant Apparatus Div	500 Penn Center Blvd		Pinsburgh	ď V	SO	15235- 5713	March 31, 2000
						Beling		Š		June 30, 2000
Beying Machine & Equipment Besearch & Beying Power Electronics Research &		Zheng: Yuan Wang (85106) 205-5243	Research & Development Centre	No 31 Bei Huen Zhoug Lu De Sheng		Men Wa	BEI	Š	102088	December 31 2002
Development Centra Chily	Ken King	(972) 437-9195	17919 Waterview Pkwy			Dadas	¥	SO	75252	June 30, 2000
Bentand Intended GmbH	Michael Neisen	08458 3407 0	DAE	Literalizate 50-52		Garmensheim		90	D 85080	December 31, 2000
Bharat Dynamics Ltd. INDIA	Cidores and	91 BO 5275866				Se de mando		2 2		March 31, 2000
Bharat Electronics Ltd	G. Ragupathy	91 80 5276866	H &O 3			Bangalore		z		March 31, 2000
INDIA		01.80.5276866	SED					4		
Brignat Electronics Ltd iNDiA	G. Magupanny G. Racupathy	91 80 5276866	Baibhadrapur	Dist Paun	Gartinal	Kotowara	ULPR	<u> </u>	246 144	March 31 2000
Bharat Flectioniss Ltd. Kotawara Angrat Housy Electricals Ltd.	P V Hayagobal		Power Electronics	Vikes Nagar		Hyderabad		<u>z</u>		
Hyderabad, INDIA	Oliver Frohith	1189 182 33394		Potualing 130	Positach 40 L2 40	Monchen		DE	D 808C3	December 31 1999
OF MARK	Withelm Brand	089 382 33528	Caummarcial Airolant Ground	1405 Bulls Days Co	Positach 40 02 40	München		0E	ē.	December 31 1999
Boarg Commercial Arphane Seattle	Brian Mills	0000 000 (-03)		Za loor) fixed con-	ebzü East Marginal Way Siriffi	Seatter	ά	SO.	98108	December 31 1995
twace or to				Page 2						

v,
뜫
-
_
ェ
۰
Δ
40
σŦ
┶.
•
ŭ
_
œ
ت
•
=
č
Ŧ
٠
~
Φ
Ç.
-
<
=
×
٦.
Ξ.
<
_

										1
Company Name	Contact Name	Contact Phone	Address	Address2	Address3	Çi ı A	State	Countr	Poste Code	Mainterpor
Since 182 suncamo O sur	Chies Fester	(314) 214 4132	P.O. Bax 516			St Cours	Š	Š	53166. 0516	May 31 2000
October 1 march 2 marc	I yndali Stafford	(281) 244-4041	Detense & Space Group	Space Station	13100 Space Center	Houston	ĭ	Sr	77059	December 31 1999
Businel Deferration of Source Committee		130 11 652, 3167	Blou 18 238 7	20819 - 72nd Avenue	DWG	X	V.A	S.O.	98032	Decembar 31, 1999
Boeing Defense & Space Kent, WA	Scott Irain	2531 773-6367	M.S 3E-73	20435 - 72nd Avenue		X For	*	Sn	98032	
Boering Kerk Douglas Aerospace TC Wang	TC Wang	等地处理者也非非地位可以地位的			5301 Boisa Ave	Hu Nington Bea ch	5	S	92647	June 30, 2000
Huntington Beach CA	Dave Silva	(818) 586-1830		PO Box 7922	6633 Canoga Avenue	Canoga Park	5	U S	91309-	December 31 1999
Borg-Warner Automotive, Warren MI	Enc Sandstrom	(810) 497-4067	Air/Fluid Systems Corp. of Mi	11955 E. Nine Male Road		Warren	ž	S	48069	June 30, 2000
Base Corporation	Inomas Froeschie	Homas Froeschie australesassette The	# The Mountain			Framingham	ž	SO	1071	December 31, 1999
Brace Technologies, Inc., Sterling	Mohannad Murad	(810) 726-3944	Au Bag & Seal Bell Division	7000 19-Mile Ruad		Sterling Heights	Ē	S	48314	December 31, 1999
Heights Neights October Agreemes (Netance Systems	Phil Rudd	01983 202418	Newport Poad			Cowes	WGHT	38	PO31 8PF	March 31, 2300
Cowes UK			New Filton House	Filton		Bristol	AVO	95	BS99 7AR	November 30, 1999
British Aerospace, Filton UK British Aerospace, Warton UK	Andrew Has	01772-85	Warron Agrodrome	Warton		Preston Preston	2 Z	88 89 89	PR4 LAX PR4 LAX	August 31, 2000 November 10, 2000
British Aelospace, Warren UK	W427J						!	;		
Braisti Aerospace Wanon UK	Mike Falls Klaus Wintersberg	01772 852533 09561 21 1681	Waron Agronome	Warron Ketschendorfer Straße 36. Postfach 13 53 50	Positiach 13 53	Præsion Coburg	Š	GB 0.E	PR4 1AX D-96403	August 31, 2000 April 30, 2000
BVB College Hubbi	G. Ragupathy	91 80 5276866	BV Bhoomareddy College of Engineering		Hubh	Kamataka		Z		March 31 2000
Canna Hiratsulia Works (FL Project) SB YASUKO	B. YASUKO		6670 Tamura			Hiratsuka City	¥ Š	٩	254	March 31, 2000
0025/SB-0206	O VASUKO				7-5-1, hakusan	Tonde-city	BAR	ō'	302	December 31 1999
Caron Toride (Susono) SB-0034/04/2 YASUKO	YASUKO	0923 003 0030	Control of the Contro	Rm 5158 Markense But		Tonde-city	IBAP		302	December 31 1999
Carteton University	David Walkey	(412) 268-8443			5000 Forbes Avenue	Onewa	§ §	5 8	A15 586 15213.	September 30, 2000 August 31, 2000
Carregie Mellon (Inversity Instance)	orium I cood	(10) 858-4236	No. 2 Yue Tan North Small St.	P.O. Box 635		Besieng	BEIJ	5	3890	June 30, 2000
CASC Reliability & Safety Research Center		(309) 578-6194			908 Wesl Raven Road	Peone	=	us 6		June 30, 2000
Caterpitar, Inc.										
COOT Bangalore INDIA CEDEX	Arvind Verkayadn Ignado Jorge	34 913357214	71/1 Mallers Hoad	Alfonso XII 3 y S		Bangalore NADRID	MORD	S S	8 _	March 31, 2000 March 31, 2002
CEDT Gorahhpur Marathwada Univ	G Regulatify	91 80 5276866						ž	~	March 31, 2000
INDIA	Henri GODFROID	(38) 455-1073		3, avenue des Trois Chânes		BELFORT CEDEX FNCM	FNCM	E.	90018	November 30, 1999
	Raynald	(38) 455-3168		3, avenue des Trois		BELFORT CEDEX FNCM	FNCM	a.	90018	November 30, 1999
CEGELEC	KRZYZANOWSKI	.41 22 767 4918		Site de Mayrin		Geneva 23		5	CH 12:1	February 28, 2000
CEAN CENTRAL OF TECHNOLOGY	Jorina Luden	(31) 772-5725	PO Box 8873	Elektrovágen 3		\$ 412 96				December 31 1999
SWEDEN	to Statement Su	136106) 8:18-7207				Gullaborg Bering			19000.	March 31 2000
Chinese Academy Spirisary	Jae-Ho Cho	(43) 161-2425	Dept. of Electional Engineering	j. O		Checng _{JU} ·City	C160	E E		January 31 2001
Chungbue Malitrical University CLATEO	Leonardo Vitela	(524) 216-3898 (70) 107-3669	Hesearth DGATAFSEAC	Cata Metable 150 Cot Fovi 18 avenue Edouard Baltis RP 1425	essie	Overetaro				December 31 1999
S NF S	GINES PICART	156) 167 3332				CEDEX	÷	π -	31055	June 30 2070
				1. 3g. J						

~
ž
-
_
7-
×
v
•
•
Œ
_
9
v
~
ā
7
ō
\mathbf{x}
=
=
C.
⋖
~
۵
ń
Ŧ
>
2
≂
Ξ
ã
₹
_

Company Name	Confact Name	Contact Phone	Address	Address2	Address3	City	State	Countr Posts!	Postal Code	MaintExpDi
Curtican to Australia Computer Sciences Corporation Fort	Tony Nygard Angela Bigi	(6129) 428 6514 (817) 777-4359	14 Mars Road		6103 Western Place	Lane Cove For Worth	NSW XX	AU JS	2066 76107	June 38-200 September 36-2000
Wath 1X Controtte Dotteshed by Concords University	Nen Or. Vijay Saod	(408) 441-6154) (514) 848-2802	1530 Oakland Road Elec & Computer Engrg Dapt	Suite «A113 1455 de Maisonneuve Bivd W (H961)	P.O. Box 512473	San Jose Montreal	₹ 2	85 40	95131 H3G 1MB	November 30 1999 March 31, 2000
Conexant Systems, Inc. Continental Teves	Don Rogge Balu Balaguru	(570) 266-2105 (248) 340-3626	3536 JFK Parkway 3000 University Drive			Fort Collins Autourn Hills	8 \$	SO SO	40525 48326	December 31 1999 December 31 1999
Curlingalial Teves AG & Co ONG	Michael Ransmann 069 603 3485	069 603 3485		Guerickestr 7	Postfach 90 01 20	Franklun / M		90	88	October 31, 2000
3	Not Available	Not Available	School of Industrial	& Manufacturing Science		Bedford		68	MK43 DAL	March 31, 2002
Cranheld University On Cranheld Greaves Bombay, INDIA	Santosh Zope	3383C2 00			Kanjur East	Bombay			6703069	March 31, 2009
Crompton Greaves Nasik, INDIA	G Ragupathy	(617) 258-2144		555 Technology Square		Nager Carte		Z :	56.0	March 31, 2000
CS Draper Latis Cambridge MA	Conthia Brown	(812) 377-6743	Technical Center Rec #10-64	WS 50160	1900 McKinley Avenue	Columbus	ξZ		_	December 31, 1999
Currinins Engine Company, "". DaimlerChryster AG	Walter Unger	.497111757226		Marcadesetr 137	•	Stuttgart.			23	March 31, 2000
DaimlerChryster Corporation, Auburn	Jery Nasiadka	(248) 576-3502	M/S 484-02 15	800 Claysler Drive East		Aubum Hills	Ξ	NS 4	48326	December 31 1999
Hills, MI Dakim Diesel Locomotive Research	Chang'an Tian	(86411) 460-1010	49 Zhongchang Sir	Shahekou Disirici		Delian		<u>z</u>	≂	March 31, 2000
Institute Dantoss Drives DENMARK	Hans Jergen	45 7488 2222				DK-6300 Graastan	c	Ď		December 31 1999
Danioss Drives DENMARK	Saren Nelson	(457) 488-3898				DK-6300 Graaslen	_	Ă	_	December 31, 1999
DASA Daimler Chrysler Aerospace AG	Michael Mark	089 607 35180	Domier Salekkensystenie Girbhi. Will-Messerschmitt-Sir	Will-Messerschmiti-Sir	Postfach 80 11 69	Műnchen		DE	0.81663	March 31, 2001
DASSAULT AVIATION	Thierry GOUYON	(14) 711-4870		78, quai Marcel Dassaun	BP300	SAINT CLOUD	žÕ.	F.P.	92214 (December 31 2000
DASSAULT AVIATION	Christian CANCHE?	(55) 613-9249	DGT/DTA/ELEC	54, avenue Marcel Dassault	BP24	MERIGNAC	AQTA	E. S.	33702	Dacember 31, 1999
Datar Switchgear Ltd INDIA	G. Ragupathy	91 B0 5276866				INDIA		₹ :	-	March 31, 2000
DEAL Deliradun	G. Ragupamy	0151 549 3583	Moorgale Road	Kirkby		Liverood	ASM	<u>.</u>	4 1X7 PE /	March 31, 2000 December 31, 5000
Deico Electronics UK Deichi Automotive Systems	Klaus Kaufmann	(49202) 291 4502		LISE-Metner-Sir. 14		Wuppertail	5			September 30, 2000
Deutschland GmbH	Laura Penney	(937) 455-6382	1435 Cincinnati Sireet	P O Box 1245		Dayton	₹	US	45401	January 31, 2000
Delpte Delco Electronics	Yow P Lam	(65) 450-8285	Singapore Design Engrg Cir.		501 Ang Mo Kio Industral Pere 1				_	March 31, 2000
Detah Delco Electronics Systems.	Dave LaRusa	(765) 451-3046	One Corporate Center	Р О Вок 9005		Кокото	Z	SO.	÷	June 30, 2000
Kokomo iN	Discussion Accord	(716) 439-3257	Building 6	200 Upper Mountain Road		1 octood	ž	86 - S	3006	Cook of salasia
Delph Theimal Systems	Wike Jones	(765) 641-0426			2900 S Scatterfletc Rd	Anderson	Z			November 30, 1999
Defphi-E	Yana Tarasenko	(617225) 000-4221				,				March 31, 2000
Denso #2 Karya	Mark da Sih.i	101/10/ East 100		500-1, nanzan	Me Checker	Kanya City				February 28, 2000
Denso Foundation SB 0147	YASUKO		IC Engineding	Shrova	910	Negation-cay	Z S	90 dt		July 31, 2000
Denso IC SR 0389	YASUKO					Kative City			5	March 31, 2000
Densu SB 0350	YASUKO					hanya-city				August 31, 2000
Denso Theory Analysis & J.D. 1211	YASUKO					kanya-cıty		JP 448		December 31 1999
Denso Unix & Paris SB 0047	VASUKO	1729-600 (26.67 (2)	E9A2 St. Andrews Road	i walen	slikmohazuno cho	ABAND CITY	AICH			November 30 1999
DEHA Maiver	Yana Tarasenko			ı	-	Malvern	•	ge Š	WA14 JPS M	May 31 2000
DGACFG	Laurent MOREL	05.55 10.52.21	Contra d'Etudes de Gramat	Bêdes		SRAMAT	rapy r	FA 16:	ec 005 91	December 31 F039

a P

J
÷
-
Ξ
ž
Ð
Į
8
Ě
2
ē
₹
÷
Σ
ጁ
š
≦ .
Ĕ
ō
5

170-100 19694

Company Mane	Contact Name	Contact Phone	Addressi	Address 2	Address3	City	State		Countr Posts!	MaintExpOt
DGA/CEG	Philippe MONUAUX	561 510 5424	Centre d'Études de Gramai	Bedes		GAAWAT	Λď	× 4	Code	
DGA/FTA3	Michel ROCAMORA	C1 39 49 30 75	BP 36	Route de Lava!	Anyers	Monteut Juga		Œ	49460	December 31 1999
DGALPBA	Frank PAILLARD VASUKO	(232214) 200-3021	BP914			VERNON	ž	Œ	27207	0∈rember 31 1999
DHS intrained Technologies LP	Gary Slevers	19721 560 5962	Research West	13588 North Central		Dallas	×	ط SU	75374	December 31 2002
DTUPENMARK	Morten Nymand	(454) 525-5294	Building 451	CADIBSSWAY		DK 2800 Lynaby		l ž		2000 Oct.
Oynamics Research Curp	Mike Walsh	2000				West Newton	¥	ns	02465	March 31, 2000 March 31, 2000
Eastman Kodak Hochester, NY	Alan Karus	1716) 726-9861	901 Eimgrove Road			Rochester	×	Sn	2089 14653	Desember 11, 1900
Falon Automotive II	Jerry Craig	(630) 260:3158	191 E. North Avenue			Carol Stream	ليد		5108	Effect of the state
Eaton Culter-Hammer Prods.	Jeff Buchholy	(414) 449-6891	3060 W Hope Avenue			Milwaukae	3		2000	DOOS 83 Amount
Milwauke, Wi ECE	Sebastien WEBER (15) 606-1000	1151 606-1000		129 troulevard Davous		PARIS CEDEX 20 IDF	20 OF		75960	March 31, 2000
ECII INDIA	G. Ragupathy Jacques CHERON	91 80 5276866						ž		March 31 2000
FDF DEH		200 200 数数数数数。			avenue du General de Gaudle	CLAMART	Ą	Ŧ.	92141	Весембег 31 1999
EDF DFF	Jean DOMERIDDE 014/03/40 / 1	D C C C C C C C C C C C C C C C C C C C			1. avenue du General	CLAMART	Õ	Æ	92141	December 31 1999
EDF DEP	Stephane LORIOT	01 47 65 41 76			1. avenue du General	CEDEX CLAMABI	Ž		;	
Ach conformed to	Sebastien	(13) 492-6819	Division Promotion Ouvnère	17, full Alburd Thumas		CEDEX	ģ	ĭ.	32141	Весембе г 31-1999
EDF/GDF - Cerura do Formania con	FAUCHER	Coccas and	The state of the s	66	9	CEDEX	IQ.	E.	78100	November 30, 1999
EEV, Cheimsford UK	Jan Przybyła	01376 517101	Freebournes Road			Chelmstord	ESSX	0 89	CM1 20U	March 31, 2001
EEV Without UK		(708) 387-3769	Dr e3. Parts Gate, Engry Bldg	5401 East Avenue		Witham	ESSX	89		March 31, 2001
Clearing Data Svs Defor Seginary		(517) 757-4461			3900 Holland Dood	McCook	<u>ا</u> ك			November 30, 2000
Stand Sys. Saginaw, MI	•					A PLICA	Ī	T SO	_	May 31, 2000
Electronic Data Systems AC	Terry Smith	7995-/ 57 (018)	TOOL MORE AVERIE			Plat	Ī	os ₹	9494 48556 n	November 30, 1999
Flectrons Data Systems AC	Lary Cypher	(/16) 359 6441	Advanced Products Dev Group	5500 West Hensette		W Hennetta	×	y.	14500	
AOCHESTER, W. Henneria, NY	Mark Lutzke	(414) 768-2133		7929 S Howall Avenue					_	November 30, 1999
Electronic Data Systems Detoc						Oak Creek	š	us sa	53154	June 30, 2000
Electronic Data Systems Delco Pwitrain Vining Huang Electronic Data Systems Delco Pwitrain Vining Huang	i Yining Huang	(765) 451-0335			2705 South Gayer Road Kokomo	Kokomo	Z	US 46	46904	June 30, 2000
Electronic Data Systems Delphi Energy Daniel Martinez	Daniel Martinez	(915) 763-7320	P.O. Box 9007			E Paso	×	21. 20.	00007	
A Engine Mgml Systems Electronic Data Systems do Brasa Lida Magda Collado	Magda Collado	(5511) 828-4125	Centro Tecnologico	Portao 15 - Predio 3	Av Goras, 2769	ob overland only				December 31 1899
BRAZIL	Both Zakas	(248) 265-2639				Series Series		## ##	P09550	Jun 30 2000
Electronic Data Systems GM IVED	Delli 1.500	į			7000 Сінсадо Яовц	Warren	∑ ∑	US 48(90	December 31 1000
Electronic Data Systems GM Lansing	Beth Zakar	(248) 265 2639			7000 Chicago Road	Warron	3) i		
Automotive Division Electronic Data Systems GM Mid Lux	Beth Zakar	(218) 265-2639	M:C 485-303-807	4100 5 Sagmay St					Tenan	Остамбаг 31 1999
Car Group, Flink, Mi	Contract Cobar	(218) 265-2639	Bldg 2-7, Cube 370 05	30003 (25.0)	-	i mi	.¥	US 4855	ĸ.	December 31 1999
Electronic Data Systems OM Powerfram Engineering Watton Mi		and the second	Blds 2.3 Cube (70.06	DEGL BAYOUT A COOL	-	Warren	MI US			December 31 1999
Electronic Data Systems GM	Jaydish Patel	or the first	space announce force	JUNUS Van Dyke Road	,	Warren	ų. W	4000		
Powertan Engineering Watter in Figure Cata Systems SM	Beir, Zakai	243/265-2639	GMPT Vö Engrieding	902 E Handon	•					August 31 2002
Provention Floribil							S)	3 18550	-	December 31, 1999
				Pages						

'n
٣̈́
_
K
ŭ
0
Œ
9
×
à
7
ن
÷.
=
<u>~</u>
2
Ţ.
Ō
5
>
つ
Ę
2
3

			1	A chairman 2		į	9000	4000	o de la companya de l	940.044
Company Name	Contact Name					A	,			
MD Systems GM	Beith Zakai	(248) 255-2639	Bidy 3 MC 9004	1 General Motors Road		Miford	Ī	ns	33BC	December 31, 1999
Free Control of State	Dina McMuliin	(248) 635-5720	Bidy 3 MC 9054	1 General Motors Road		Millord	Ē	sn.	48360	August 31 2000
Fleckoric Data Systems Provertiant, Millord MI Construction of the Systems GMC	Frank Schauenu	966 0195	Building HANB Ruom 196		30500 Mound Road	Warren	ŝ	ns	18090-	March 31 2000
Research Little Warren, Mi	Beih Zahar	(248) 265-2639	MCD HQ Building 2:10	Roam 134-22	30001 Van Dyko Foad	Warren	₹	υS	48090	December 31 1999
Vibration Constitution Warrent Miss Systems MCD	Wade Juhnson	(810) 578-3483	MCC HQ Building 2-10	Room 134-22	30001 Van Dyke Foad	Warren	Ē	so	48090 4000	December 31, 2010
Warren, MI Warren, MI	Beth Zakar	(248) 265-2639	M/S 93-K	408 Dana St NE		Warren	P	Sn	44486	December 31 1999
Electronic Date Systems (IVED, Troy, MI Beth Zakas	Beth Zakas	(248) 265-2639	M/S 4222	750 Tower Drive		Tray	Š	so :	48039	December 31, 1999
Elementech HOLLAND	Munitin Karakus	20 70 70 75 15+	reliation and and	Bidboardseen 1		DENIEN DO C/+/				noo? of Jaguardas
EIM EBG EIJHIOON	Rudolf Febringer Henn FOCH	(56) 158-8256	Laboratore d'Electrolechnique	et Electronique industrielle 2, ruc Charles Camichel TOULOUSE CEDEX	e 2, rud Charles Camichel	TOULOUSE	MPY	ē Œ	31071	September 30, 2000 June 30, 2000
	Marc COUSINEAU (56) 158-8312	(56; 158 8312	Laboratoire d'Electrofechnique	et Electronique Industriells 2, rue Charles Camichel	s 2, rue Charles Camichel		¥ A	Ē.	31071	June 30, 2000
ENSEWINPL GREEN	Bernard DAVAT	(38) 359-5503	2, ave de la Forel de Haye			VANDOEUVRE	LARN	Œ	54516	December 31 1999
ENSEMINPL - GREEN	Prof. SAADATE	1361 391-2372	2, ave de la Foret de Haye			VANDOEUVRE	LAAN	Œ	54516	December 31 1999
AIGNI EDION (Time ACIDING CO.	Sakar Jan	119111) 855-3754	A-5, Sector-26			Norda		<u>z</u>	201 301	March 31, 2000
ERADC Tovandrum		91 80 5278866 (468) 721 6839	Energy Systems Division			S-164 81 Kisla		Z W		March 31, 2000 March 31, 2000
SWEDEN SWEDEN Engeren Telecom (Telus) SWEDEN	Of Walgren	(468) 719-9352	Teilusborgsvågen 83-87			S-126.25 Stockholm		SE		December 31, 1999
Losson Telecon BHAZII	Gilberto Tatibana	(55) 759-5208	Rua Ambrosio Molina 1090	12247.000 Sad Jose dos Campos		Sao Paulo		88		December 31 1999
531390 4 53	Anne JUSTEAU	(24) 090-5069	LARGE	CRTT Bd de l'Universite	B.P 406	SAINT NAZAIRE	POLL	Œ	44602	March 31, 2000
LOS OF PCX LIFERNIKOA	Estares OYARBIDE	ų		Loramends, 4 Apdo 23		4AGON	BSOU	ES 2	20500	June 30, 2000
ESAUCA	Jean-Paul DIDIER (14) 127-3700	1 (14) 127-3700		3, rue Jules Vernes		LEVALLOIS	Ďε	F. 9	92300	December 31, 1999
ETH Zurch ETH Zurch ET C. MAD ISTRIALES DE	Tobias Oetker Santiago de	+41 1 632 5286 (3498) 342-3344	inst i integrierte Schaltungen	Glonastraße 35 Paseo del Cauce, sin		Zürich VALLADOLID	פרע	CH C	CH-8092 47011	December 31, 1999 December 31, 1999
VALLADOLID Funec Gridh & Co KG	Andress Schulz	02902 764 1502		Max-Planck-Str. 5		Warstein-Belecke		0 3 0	D-59581	July 31, 2000
EUROCOFTER	Andre LEPERS	(44) 285-6817	Aeroport International	Marseale-Provence		MARIGNANE	PACD	E.	13725	June 30, 2000
Eurocopter Deutschland GmbH	Peter Almann	6689 0009 686		Willi-Messerschring-Str Tor 1, Geb 85	-	Munchen	_	OE 0	D R1663	March 31 2000
CHIM offered at the control	Christian Humer	. 43 1 333 30 82	FR Elektronik Service d'Elitetrohechmeter	Wexstr. 19-23					8	December 31, 2000
Faculte Polytechreque de Moris	Michel RENGLE		DEEC	ZBOO DBASIDO	Rua dos Bragas	MONS Porto	PATO P	BE 70 PT 40	7000 A	March 31, 2000 November 13, 1999
Faculty of Engineering Transfer Faculty Keren Serpicon Madde	CARVALHO Kee-Shik On	82 32 680 1246		Kyünggi-province 420 711 Korea			*.	ď.	٥	Dacember 31 1999
FASC CARRONS CARRACANOR, Smellby	Renard Wade	(704482) 958 2288	1100 Arrest Road		5	Snelby h	NC O	ŕ≀ sn	جن,ون ۱۸	March 31 2000
NG Fasci Mators Studio Control to a specificial SCO MG	Spadat Mussain Linti Kather	1174852311-1973 1741-3689-511	1500 West Jackson Street	heuenhofstr 181	0 4	Ozark N Ausnen	CW	US 65	65721 J.	July 31-2000 December 31-1999
FEV Modern autorian and an				स्त्र कि कि						

•
¥
_
=
ŏ
ā
•
Ē
3
۲
ro.
=
¥
Ξ
Ŧ
2
ī
ō
Ž.
2
≅.
ō
3
•

***** ****

Company Name	Contact Name	Contact Phone	Address	Address2	Address3	CJF.	State	Countr Postal		MeintExpDt
FIAT RESEARCH CENTER	Brund Oil ECCE	9011 902-3149	Strada formo 50			ORBASSANO	PE	1001		February 28 2000
	Makes Bround	(612) 949 5178			12001 Technology Dave Eden Prairie	re Eden Prante	Z	US 55344		December 31 1999
Fisher-Reservand inc	Dan a first	01268 402879	Small Vehicle Centre	Laundon	•	Rasidon	ESSX	3.8 SS15	SS15 6EE De	
	Use Mithies	11, 233 9211		Estrada Nacional 252, Km		PALMELA				
FORD ELECTRONICA PORTUGUESA Lord Flectronics Mig Corp. Markham		(505) 475 8510	7455 Birchmount Road			Markham	Ž O	CA L3R SC2		
ON COMPANY OF THE COM	Sleven Wemk	(719) 528-7858	MSO	9965 Federal Drive		Colorado Springs	8	US 80921-		August 31 2000
IL, Dearbon	Henry Huang	(313) 645-3768	20000 Retunda Dave			Dearborn	ž	J036 US 48121		June 30 2000
MI ALTER MEXICO	Roberto Pachecu	52142022 25	Alse. Dept PEO	Engrg. & Refrigeration	1270 Don Haskins Suite El Pasu	e El Paso	Ϋ́	5036 US 79906	10	December 31 1999
Ford Motor Company ALIEC MEXICO		•	3 C 3 C 7 C 7 C 7 C 7 C 7 C 7 C 7 C 7 C		.×.					
Ford Motor Company AVT, Dearborn,	Lawrence Krass	(313) 390-5247	Building 5, MC 5016		20000 Rollwids Drve	Dearborn	Ξ	US 48121- 2053	<u>.</u>	March 31, 2000
Mi Ford Motor Company Transportation &	John Czubay	(734) 268-6617			36200 Plymouth Road	Lvons	5	US 48150		December 31, 1999
Chiassis	Lill Fox	0221 90-32957	Apt MC/ME EEO	Educi-Ford-Str. 2-14		Kolin		DE D 50769		December 31, 1999
Ford Werke AG Koln	Heinrich Wolf	089 54759 031		Hansasır 27d		Munchan		DE D-80686		October 31, 2000
Fraunholer Géseuschan	Alan Fladmoe	(808) 934-4104	3055 Orchard Drive			San Jose		٥N	å	December 31, 2001
Fujitso America: m.c.	YASUKO	•			4-1-1 kamodanaka	kawasaki city				March 31, 2000
Courters Microelectronics UK	Andrew Kemp	01628 504781	NEWSTR TOOSE	Morrays Unive	:	Mardenhead			_	June 30, 2000
Control Perpoperal SB-0302	YASUKO			30 Man 0	VBShiro-cho	kato-gun	HVOG			March 31, 2000
c.7: Forechingszent Informatik	Thomas Strigi	0721 9654 154	tied of contract	ADS 100		Karlende			=	September 30 2000
CE Anglances, Louisville, KY	Mike McIntyre	(502) 452-5431		AT3:103		Louisville				June 30, 2000
SE CHAND	Ahmed Elasser	(516) 387-5745	Document Bouth 606	20 a 20 C a		Schenectady				December 31 1999
GE Fanue North America, Inc	Jim Bargeron	(604) 978-5044	1501 Boardte Blvd		/ O SW	Charlonesville				February 28, 2000
GE Industrial Control Systems	Mark Shepard		41 Woodford Ave				√ .			March 31 2000
GE Industrial Systems	Douglas Tighman							OS 8062	Š	December 31, 1999
HO	Gred Butler	(218) 266-6572	Nela Park, Bldg 335	Tech Division	1975 Noble Road	Oeveland	ਰ ਰ	US 44112		September 30 2000
GE LIGHTING CIEVERATE CO.	Jacky LE BOZEC	(13) 070-4859		283, rue de la Miniere	BP 34	BUC CEDEX	iDF.	FR 78530		December 31, 1999
On Machinal Systems - New Berlen, WI	Russ Hum	(414) 785-5216	MCS NB-926	16800 W Ryerson Road		New Berlin		US 53151		December 31, 1999
General Dynamics Electric Boat	John Sheginon					Gration	כו	US 6340	Jank	January 31 2000
Division		200 474.3895	Eastern Point Road						,	
General Dynamics Electric Boat	John Staller	and the transi				Groton	5	6340	9	September 30, 2000
Division George Divisance Electric Boat	John Walter	(860) 433-4465	Eastern Point Road			Grolon	CT L	US 6340	Sept	September 30, 2000
Division	1000 mg	(860) 433-4465	Eastern Point Road			Grapho	-	27.0	č	
General Dynamics Elisting Boat	MCK Emigin					i de la composição de l			e e	September 30, 2000
Division	Sleven Zdon	(413) 494-6154	Detense Systems		100 Plastics Avenue	Pittsfield	⊃ ¥¥	1201	Sept	September 30, 2000
Ceneral Dynamics, Control of the Parkheld MA		3030 004 1400	260 Hawts His Boad							
General Electronic Applications Scotts Joe Pernyeszi	s Joe Pernyeszi	(831) 438-050	2001 21 7000 1007			Scotts Valley	⊃ ∀ ∪	US 95066	July	July 31, 2000
Valley, CA	Ken Konoid	(310) 517-5947	Blog 237	3050 W LOMAS BIVE.		Totrance	SO AS	50506		November 30, 2000
General Molers Augustan		,								
Vehicles General Motors Brake and Bearing	Thomas Flann	(610) 685-5519	Braked Bearing Systems	3300 General Motors Road		Millord	SU IN	S 4838c	Octot	October 31 2000
Systems Melford MI	Craro rheks	(810) 986-5934	Engig Center	•	30200 Mound Rd	Waren	S⊃ F	S 48090	N N	May 31, 2000
General Micros Coly (Constitution)				1606 Tacharden					ì	
Operations View Community Service, Trus	Beth Zahan	(248) 265-2659		rago i dermunda La		Tro.	Mi US	18007	Dece	December 31 1999
Mt Connect Meters Powertrain Ypsilanie	Bein Zakar	12481 265-2639	Ecuse & Ward Rid			Ypsilanti	λ. 1.4S	S 18198	Jace.	December 31 1990
MI										

10
×
-
-
≒
×
1
α.
٠
U
Ċ
4
ē
ō
-
\sim
ā
÷.
~
2.
0
-
<
õ
÷.
ŏ
Š
<
_

Company Name	Contact Name	Contact Phone	Address	Address 2	Address3	Cħ Ż	State	Countr Postal y Code	Postel Code	Maint Exp Ot
Modes Powerten Yesten	But Goodwin	(734) 481 5104	Ecorse & Ward Ho			Ypsilanti	ž	Sr)	48196	March 31, 2000
.:	Heli Zakir	(248) 265-2639		2000 Centerpoint Pkwy		Portiac	Ē	JS 48J4	_	December 31-1999
MI INDICATRIES	Flobert	05 61 34 35 92	CENTRE DE TOULOUSE	152, avenue de Grande Bratanna		TOULOUSE	¥ dw	FF 33.	0.1	December 31 1999
GIAT INDUSTRIES	CANTEGREIL Pierre MAGNAN Siddharth	(3324) 821-9077 91- 11-436475 7		7 route de Guerry	BP 265 Niketan, 6 CGO	BOURGES New Deni	CNTR	g :	18023 6	Decembor 31 1999 March 31 2000
Covering Politics Company Booder Cranville Politics Company Booder Cr	Bhallacharya Bob Gray	(303) 998-4058	5675 Araphon Ave		Coracte a	Boulder	8	U.S 8030X	_	June 30, 2000
Grundias DENMARK	Hans Stougaard	(8) 668-1400	Poul Due Jensens Vej 7			DK-8850 Biermobro		¥a	•	March 31, 2000
Hahn Schickard Gesellschaft - IMIT	Matthias Slorz	0711 943 0		Withelin-Schickard-Str. 10		Villingen Schwenningen		DE	D-78052	September 30 2000
Helia KG Huuck & Co. KG Helsinki University FINLAND	Ewald Hessel Matti Linnavuo (Depti Elect &	62941 38 B572 [3] 590-4512	Olakaan 5A	Rinbecker Str 75 P.O. Box 3000		Uppstadt FIN-02150 Espou		90 =	0.59552	March 31 2000 March 31 2000
Harital Walt University UK	Comms) Ken Smith	(131) 451 3328	Dept. of Computing & Electrical End	Riccarton		Edingburgh		G8	EH14 4AS	March 31 2000
That's America Ltd. Farmington illits. Mi	George Sakalis	(248) 473-6119	34500 Grand River Ave			Farmington Hills	ā	NS 48:	48335	March 31 2000
Hisch, Automotive Products	Gordon DeBnncal	it (248) 473-8795	34500 Grand River Ave			Farmington Hills	₹	US 48:	48335 F	February 28 2000
Fairmington Hills MI History Central Research Lab SB-01866 YASUKO	YASUKO				1-280 Higashi Korgakuso	Kokabuny SN	TOKY	281 qt		March 31 2000
Hisachi Lati SB 0423 Hisachi Micro Systems UK	YASUKO Ashok Dhuna	01628 585000	Whitebrook Park	Lower Cockham Road		Mardenhead	BEAK	. 85 e	SL68YA S	September 31, 1999 September 30, 2000
HAACH Tuchig SB-0372	YASUKO Berend Evenbli	31 4834 45243	Aingdijk 390	2980 DA Ridderherk				Ьź	נטנ	December 31, 1999 December 31, 1999
Holec HOLLAND Honeywell H. B.C. Golden Valley, MN		(612) 954-4861 01698 481078	Newhouse Industrial Estate	1985 Dougles Drive North		Golden Valley	₹	US 55422	ď	December 31, 1999
Honeywell UK Huawei Telecommunications Co Ltd	Alastair watche Zhu Xianlong	(96755) 654-1126	R&D Depart.	Huawei Industrial City.	Longgang Industrial Zone, Longgang Det	Shenzhen				March 31 2004
Hughes Arradt Space & Communications Group El Segundo.	Doug Ourgley	(310) 416-4038	Bidg. S-12, M/5 W305	1950 E. Imperial Hwy		El Segundo	₹	US 90245		October 31, 2000
CA Hyunda Motor Company Ulsan	JEONG KWEON	(2) 420-6700	Electronic Test Team 1	700 Yongchung-dang	Jung-Ku	Ulsan	KTNIM	KA 681	681-791 M	March 31, 2000
t/O Sensors, Inc	Ben Janes	1512) 340-411B	7600B N. Capital of Yends Hwy Suite 100			Auskn	χ	US 78731.	<u>.</u>	September 30, 2000
IBM Corporation P.T.P. NC	Randhir Malik	(919) 254-1134			3039 Cornwallis Acad	Research Triangle NC		US 27709		0ecember 31 1999
IFA/LTH, LUND SWEDEN	Per Kartsson	464 6222	Box 116			S-221 00 Lund	V/ 3	₩ ≱	v z	Saptember 30, 2000
III New Delli, INDIA IKERLAN	C.M Drimie Raoul REYERO	(4) 377-1200	CENTRO DE INVESTIGACIONES TECNOLÒGICAS	J.M. Anzmendiarrieta, 2	P O Bur 146	MONDRAGON (GUIPLIZCOA)		ES 20500		April 30 2000
MEC BELGIUM	Guido		Kapeldreef 75	B.300" Leuven			ш.	96	J.	July 3. 2000
MALCH GRABH INSIDE for Mechatronik MS Inst. Mercelektronik Study.	Andre Jerneri Andre Jerneri Vinkus Nicke G. Ragupathy	02841 191 251 6711 685 5839 91 80 527c666		Eurotec-Ring 15 Almandring 30a		Moers Stuftgart Kampur	005	DE D 47445 DE D 79569 IN		December 31 1999 December 31 1999 Marzh 31 2000
hotan bisholi or name y				ກ ຄວາ						

WorldWideMaintenanceReport als

Company Name	Contact Name	Confact Phone	Address	Address2	Address3	City	State	Countr Postal	Postal	MaintExpDt
Arama Alexander of testing of the particular	Vivek Ageraal	91 22 578 2545				Powa. Bunha	à	<u>, </u>	Code 400.075	
	;		F.20 As Man Boxa	(÷	<u>:</u>		March 11 2000
inden tretanc et Technology Madras e Dio	VISACIV			ii campus		Madras		Ž	960009	March 31 2000
institute of Technishingy, Diffin	Visweswaran					Delhi		<u>Z</u>		March 31, 2000
INDIA Impan institute of Technology, Dethi	Therunavukkarasu	0091 080 994 1810				Ğ		3		:
INDIA	Annathera	1940 640 0767				<u>.</u>		:		March 31, 2000
Industrial Research (1d NEW	David Greager			Graceheld Research Centre	Gracefield Road	Lower Hint		NZ		November 30 1999
ZEALAND	Tony Sopranzi	(408) 725-3458	OptoBectronics Division	10960 North Tarolay		(
Inheen Technologes MUC	Georgica	089 234 24313		Balenstr. 73	Postlach 80 09 49	Cuperino Minchen	∀	US DE	95014 D-81609	July 31, 2000 September 30, 2000
	Albado	(34)944209680		Parone Terrange						2007
INGELECTRIC TEAM S.A.	GUERRERO			Edit 106		ZAMUDIO	BSOU	ES	48170	March 31, 2002
INNETS	Zoubir KHATIR	(14) 740-7334		2. avenue du Général Malaces		ARCUEIL CEDEX IDF	EX IDF	Œ	94114	March 31 2000
AiONI RIAS Salas de sens	G Ragupathy	91 80 5276866								
INSTITUT CHARLES FABRY	Rachid	04 91 05 46 33		Université de Provence	C/O IMT Technopole de MARSEILLE	de MARSEILLE	PACO	- 2 E	13451	March 31, 2000 March 31, 2000
	ALTERNATION AND A STREET AND A	no 84 22 90 01 ou 03 Parc	Parc Technologicus	2 Brembie fant br.	Château Gombert	CEDEX 20				000 S 1000
INSTITUTOR GENIE ENERGETIQUE	Marc HAMELIN	(38) 457-8206		2. Byenue Jean Moulin		BELFORT	FNCM	F.	00006	June 30 2000
INSTITUTION OF GENERAL CONTRACT	Zhano Yu	(860106) 252-1027		6 Bei Er Tian Zhong Girm Ling	: 6	BELFORT	FNCM		90000	June 30, 2000
Instable Flecting Engineering of Chinase	•			Cun	na Uan Usinci	Beikng		C.	100001	June 30, 2000
Academic Science Institute of Chinese Zou Xi institute Electric Engineering of Chinese Zou Xi	Zon Xi	(86106) 261 4988		6 Bet Er Tiao, Zhong Guan Hai Dan District	Hai Dain District	Beijing		3	100080	June 30, 2000
Academic Science	Hend Bun Suan		11 Science Park Rd II	5		1				
Institute of Microelectric actions	Cau Chuon How	710.5479	11 Science Park Rd II			Singapore				December 31, 2002
Institute of Microspectromes	Americo Vincente	(35173) 330-3083	Escola Superior de Tecnologia e	Campus de Santa		Singapore				December 31, 2002
INSTITUTO POLITICONICO DE	Leife		Gestao	Apolonia		BHAGANCA	BGNC	P1 5:	2300	March 31, 2000
Intel Corporation Hillsboro OR	Kent Cataban	(503) 613-7667	Portland Technology Devalopment	2501 NW 229th Ave		Hilaboro	9.	76 SO	, 42126	April 30, 2000
A residence of	Victoria Heii	(602) 554-8110	5000 W. Chandler Blvd			i				200
Intel Corporation Charles of		4000	Coc.			Chandler	7	US 85	85226. t	November 30, 1999
International Rectiler	Attonso (Al) Diy	(5704746) 761-4346	(510) /50-812/ (5704746) 751-4346 125 Crestwood Road			El Segundo	5	NS SA		March 31, 2000
Intersal Corp. Semiconductor Products Mountainton, PA						Mountaintop				August 31 2000
Invotronics Manufacturing.	Michael Ma	4163218822 320	365 Passmore Avenue			Scarborough	Š	CA	Z189 M1V 483	6000 1C 7F1
Scarborough, ON IPHT Institution	Urrich Dillingr	03641 206 312		Heimholtzweg 4	Postfach 100 239	Jena	·		,	200
Hochiechnologie e.V IRESTE Naries FHANCE	Denis JEULAND	(24) 068-3011	La Chantrene CP 3003			Nacies CEDEX 03 PD. 1			70//00	June 30 2000
	Raff Juchem	0561 7294 314	Solare	Konostor 59						June 30 2000
ISET institute V			Energievorsorgungstechnik			Kassel	٥	DE D-3	D-34119 A	August 31 2000
ISBO increas Systems Unit INDIA	S Ramarathnam VASUKO	91 471 361810	v athydofrav u,			Tavandrum	2		695 O13 MA	March 31, 2000
42454	Civardi Lorenzo	(3924) 388-8241	Palazzo Quadriloglio - CLTE CO1	Casteletto di Contrato			٩			June 30, 2000
ITALTEL MILANO			•	Milanese		SETTIMO	TH DBW I	50019		December 31 1999
ONA IMI TOUR OF LATE	Vagin Angelti				Cascina Castalano	MILANESE	•			
ITALIEL BUSH MILANO	Storem Lenge		SCIECTS AND)	Casona Castelletter	20019 Settime	: <u>t</u>		م م	December 31 1999
:	John Anight	01203 363080	Frigulteanny (entre	Abbey Ruad Whyles		Milanese			5	ORCHIDGE 31 1989
All vites the	Gregian			•		Covening	WW GB		CV3 4LF Ap	April 3C 2C00

Page 3

y)
ã
_
7
≂
×
ă
×
ų.
9
ō
⊆.
œ
Ç
•
Ξ
Ξ
<u></u>
≥
•
ŏ
Ē.
<
Ħ
Ξ
5
₹
3

			9DrADUOAA	WorldWideMaintenanceHeport1 xls						
Company Name	Contact Name (Contact Phone	Address1	Address2	Address3	City	State	Counir Postal y Code	Pastal Code	MaintExpOt
	Keth Baker	319) 292 8721		Gale 1. Hidgeway Avenue		Waterloo	<u>.</u>	St.	50704	December 31 1999
	Walgang Bodner	.43 1 81111 1873	Kia Tochnical Center	Wagensekgasse 1 781 I, Soha-Dong		Wien KweigMyung City	KYON	A A	A 1121 423 050	December 31, 1999 December 31, 1999
KIA Main's Thirling Committee Commit		(518) 395-5218		P. O. Box 1072		Schenectady	ž	s o	12301	January 31 2000
Kinglis Alumic Power Laboratory	Tom Van Wed	(518) 395-4733		P Q Box 1072		Schenectady	ž	So	12 3 01	January 31, 2000
KOLLMOHGEN ARTUS	Claude BARBIER Sang-Hoon Chang	(24) 133-6377 (8234) 361-8531	Chemin du Champ des Martyrs 374-1 Woulam-dong			Avnite CEDEX Uwang-city Kyonggi-do	POLL	A E E	49241 437 050	September 30, 2000 January 31, 2000
Kostal GmbH & Co. KG Krit (Electionics) SWEDEN KIH (Electionics) SWEDEN KIH (Power Engineenny) SWEDEN	Thomas Dr. Abel Haniu Tenhindin Anders Bergqvist	02351 162203 752-1240	ELE/ESD Lab Electronic Power Engineering	An der Bekmere, 10 Box 229	Positach 1969	Ludenscheid S-164 40 Kista S-100 44 Stockholm		0 SE SE	0.58513	February 28, 2000 March 31, 2000 November 30, 2000
KTH (Power Finginaeanny) SWEDEN	Ghran Engdahl		Electronic Power Engineering			S-100 44 Stockholm		SE		November 30, 2000
KTH (Power Engineering) SWEDEN	Remy Kolessar	(11) 468-7906	Electronic Power Engineering			S-100 44 Stockholm	;			June 30, 2000
1.3 Communication Systems, Camben	Richard Hendi	(609) 338-2845	Mrs Ade: Cae			Camber	ž		8102	December 31, 1999
NJ LAAS-CNRS	Bruno JAMMES	(56) 133-6991		7 avenue du Colonel Ruizhe		TOULOUSE CEDEX 4	≻ å		31077	March 31 2000
SHNOTA	Charles	(56) 133-6475		7. avenue du Colonel Roche		TOULOUSE CEDEX 4	≻	er E	31077	March 31 2000
(4) (4) (4) (4) (4) (4) (4) (4) (4) (4)	BOURDEAN David HANIA	01 49 18 34 42		17.rue de Clichy		SAINT-OUEN CEDEX	IDF	g.	93584	December 31, 1999
LABINAL MASORE : MESORE	Vyjendra	91 22 8521401	KIADB Industral Area		5200 Auto Club Deve	Mysore	ž	₹ 55 S2	48126.	March 31, 2000 June 30, 2000
	Grag Dickerhool	161985-1658		Plateau du Moulon		GIF SUR YVETTE 10F	و			September 30, 2000
(GEP	Claude						6			
Lourneed Monin Astronautics	But Jackson	(303) 977-3592 (303) 971-2433			12257 State Hwy 121 12257 State Hwy 121	Litteron	3 8	s so	B0127 B0127	Decamber 31, 1999 Decamber 31, 1999
Lockinged Martin Astronautics	Larron D (Luren)	(303) 971-5532			12257 State Hwy 121	Littleton	5 S		60127	December 31, 1999
LINADERE Martin Astronautics	Webb Debbie Geers	(303) 971-3338			12257 State Highway 121	Littleton	ပ္ပ	e sn	80127	December 31 1999
costneed Martin Astronautics	Stave Kwon	(303) 977-4904	•		12257 State Highway 121	Littleton	03	as SO	80127	December 31, 1999
Locationed Martin Comm & Power Cir	Thomas Yeaple	(215) 497-2037	Convinuescations & Power Center 100 Campus Drive	r 100 Campus Drive	M/S 2508	Newtown	4	E SO	18840	December 31 2000
Luckheed Martin Imaging Systems	Hon Ward	(78 1) 863-3581	JA:S 188	2 Forbes Road		Lewington	₩ ¥	SU SY	7393	September 30 2000
Luckneed, Bldg 151	Maynard Galyan Sam Burkit	(408) 756-8531 (303) 971-6128	8kg 151	1111 Lockheed Way	12257 State Hwy 121	Sunnyvale Littleton	& 3	SO SO	94089	June 30 2000 December 31 1999
Localised recommendation CO	Burnel IChung	(408) 433-6643	3115 Alfroil Sirect	MS J201		Santa Clare	- •	se sn	95054	February 28 2000
LSTEagle Corporation	Ping, Wan	(970) 206-5972			2001 Danheid Coun	Fort Collins	3	oa so	H0526 [December 31 1999
CS) of Osterapor	State Burnham	(719) 531 7023			20ul Canlield Coun	Fon Collins	3	.e 8∪	3 92506	December 31 1999

s
S
=
5
Ž
ř
ŧ
2
ō
ũ
چ
₹
€.
Ŏ
≥
ā
ā
>

					•					
Company Name	Contact Name	Contact Priorit	- 230 -20 V	A odress 2	Address3	Ž.	State	Coco	Countr Fostal	MaintExpOt
18) Forth Comparation	Sleve Oortend	9792235100 9050			2001 Danheid Court	Fon Collins	9	\$0.	90\$26	December 51 1499
Lucas Aerospace Cargo Systems	Alan Leingang	(714) 671-4591	610 Neptune Avenue	PO Bus 2207		Brea	5	U.S	92822	March 31 2004
Lucus Agrespace Autera OM	forn Blackford	(330) 995-1000	777 Lenia Drive			Aurora	ĕ	S,	44202-	March 31 2004
Lucas Aerospace Fordhouses UK	Ditvid Brown	(190) 264-4844	Stafford Road	Fordhouses		Wolverhampton	3	68	WV 10 7EF	8026 WVIQ 7EH March 31 2004
Lucus Aerospace, Fordhouses UK	David Tromes	01902 644644 x 3514	Stafford Road	Fordhouses		Wolvemampton	N N	85	WV10 7EH	WV10 7EH March 31, 2004
Lucas Aerospace, Mernel UK	Chins Adams	01442 242233 x 3144	Maylands Avenue			Hemel Hempsiead HERT	d HERT	89	HP2 4SP	March 31, 2004
Locas Aeruspace, Mernel UK	John Huni	01442 242233	Maylands Avenue			Hemel Hempslead HERI	d HERT	95	HP2 4SP	March 31, 2004
Lucas Aerospace, Shahmore Lane UK Liveus Aerospace, York Road UK	Martin Kirkman Mark Wood	0121 707 7111 0121 627 6600 x 3683	Shafimore Lane York Road	Hall Green		Birmingham Birmingham	3	89	B15 BSW B28 BLN	March 31, 2004 March 31, 2004
Lucas APD. Strattord Road UK Liicas Braking, Fen End UK	Maicolm Sparey Tony Ainsworth	(121) 627-4393 01203 533310	Stratford Hoad Oldwich Lane East	Sokhull Fen End		Birmingham Near Kernilworth	WM WCKS	62 63 63	B90 46W CVB 102	January 31, 2006 June 30, 2000
Lucas TVS Madras Maglev Ins of South West Jiotong Univ MAGNETI MARELLI	V. Narayanan Zhang Kulon Fabion PagaNELLE	91 44 62 58211 (8628) 766-0338 (14) 729 7714		19 rue Lavoisier	BP 214	Medras Chengdu NANTERRE		3 5 E	610031 92000	March 31, 2000 March 31, 2003 December 31, 1999
Majardalens Hogskola SWEDEN	Oif Bjurman	(2) 110.3102	Inst for Elektroleknik	Вох 883		5-721 23 V&steras		SE		November 30 1999
Manchester Metropolitan University UK	Fred Swift	(161) 247-1664	Dept of Electrical & Electronic Eno	Chester Street		Manchester		68	M1 5GD	June 30 2000
Marcon Aeruspace Systems, Inc.	Rich Morgese	(973) 633-4418	164 Tolowa Rd			Wayne	₹	o.s	07474.	March 31 2000
Wayne NJ Marcuni Annrics CSSD, Million Reynes Mark Humphrey	. Mark Humphrey	01908 483124	Foxhunter Drive	Linford Wood		Milon Keynes	BUCK	85	0975 WK14 6LA	March 31, 2001
UK Marcon Avience RCSD, Crewe Tuli UK Kevin Modre Marcon Avienits RCSD, Ponsmouth Sieve Shekle	K Kevin Moore Sieve Siickley	0131 343 4690 01705 22	Crewe Toll Browns Lane	Ferry Road The Arport		Edinburgh Portsmouth	HAMP	89 99 99	EH5 2XS PO3 5PO	March 31, 2000 March 31, 2001
UK Marconi Avionics PCSD, Silverknowes	Kenneth Weir	0131 343 5054	Ferry Road	Silverionwes		Edinburgh		68	EH4 4A0	March 31 2002
UK MACODI AVIOTIUS RCSD. Stammore UK MACODI AVIOTIUS SCSD. Basildon UK MACODI AVIOTIUS SCSD. Basildon UK MACODI AVIOTIUS ROCHESTER UK	Chas Gregory Simon Collins Robert Cooper	0181 420 3654 01268 885297 01634 844400	The Grove Chinstopher Martin Road Avionics Systems Division	Warren Lane Airport Works		Stanmore Basildon Rochester	MSEX ESSX KNT	8 8 8	HA7 4LY (SS14 3EL 1	December 31, 1999 March 31, 2002 March 31, 2001
Marcon Avience South Gyle UK	Steven Redly	K465/ 0131 314 B289	Avionics Systems Division	1 South Gyle Crescent		Edinburgh		85	~	March 31, 2001
Marcola Communications, Chemstord	David Theadom		New Street			Cheimsford	ESSX	69	CMIIPLA	March 31 2000
LIK Marconii Research, Great Baddow UK — Clive Mille Matsushala Dunkoh Lighting System SB-YASUKO	Ciiva Miller B. YASUKO	91245 242532	West Harnurghold Road	Great Baddow	1048 oaza kadoma	Cheimsford tadoma city	ESSX OSAK .	0.80 0.80 0.80	CM2 8HN 1.	March 31 2002 August 31 2000
02P3 Maisushira Derikoh R&D SB 0228 Maisushira Electric Corp. (MEC: R&D	YASUKO YASUKO		ALD		1048, oaza kadoma	na city IOKAKYOU	OSAK .	ar ar	57: N	Novenibar 30, 2000 March 31, 2000
SB 0376 Mayo Founfalion Represier, MN	Don Glascock	(507) 264-1850			200 First St. SW	CITY	- AM	s sn	\$5905	April 30 2000
Mazda rokohania R&O Center SE	YASUKO			2.5 monyachu	ห ล กลgan.aku	Yokohama	YANA J	Z dı	(127	Jecember 31 1999
SHC										

PATENT

Page 11

REEL: 011820 FRAME: 0393

ž
1100
eFe Fe
SAAC
ă Z
de⊠
Š
Š

TZ0-6T0 -\$684

			MadeWicher	WordsWideMaintenanceReport Lists						ع : :
		Phone	Address	Address2	Address3	City	State	Count	Code	
	Contact Name					1. 4. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	Š	s	18075	April 13 20cm
	64 S (5) E	(248354) 666 6161	Suite 207	24450 Evergreen Road		a contribution	į			
Methodis Europaines					347 Madison Avenue	New York	, Z	v.s	71001	October 31, 2000
Party Commuter Rusticad New P	Paul Cheng	(914) 271-1747		neal annave 081	curodudy Montpotent	SAINT	RNAL	æ	36,134	December 31 1999
	Isabelle STAUR	04 76 18 33 17		Kuntzmann	Saint Marin	CEDEX		S)	61032	Dacember 31 1999
	Norman Heyen	(815) 286 3159	Box 520	Bruttovagen (in Mest spinds in	S-175 26 Jartilla		SE		December 31 1999
Micro Switch Freepont, 10. Milet Semicondulint SWEDEN	Ame Lindstrom	(4685) 502 -515			1-1-4, takezono	tsukuba city	IBAR .	ط ہ	305	March 31 2000
!	YASUKO		Industrial Electronics		2-5, tsukaguchi bosmachi	Amagasakı	нхов	2 5	§	Maich 31, 2000
Missionshi Electric CDL SB 0029/0157	YASONO		POWER DEVICE DIVISION	POWER MODEL	1.1.1 IMAJUKU	NISHI-KU	FUKO	ر	819.01	March 31, 2000
MITSUBISHIFIECTRIC FUKUOKA SB. YASUKO	YASUKO			DEPARTMENT	HIGASHI 1. Nahimachi	nazawa	AICH	ā	492	April 30 2000
0175/0199 Masubish Electric Indiana Works SB:	YASUKO				8-1-1 tsukaguchi	amagasaki city	HYOG	٩	199	March 31, 2000
O012/0035	YASUKO				normach	Danapleativo CRV	KYOT	로	219	November 33, 1999
Misubism Caracia Kvolu SB-0216	YASUKO		Nagoya Works	1-14 Yada-Minami 5-	Higashi Ku	Nagoya	ACH	ē,	461	March 31 2000
Masubishi Eindric Nagoya Works SB-	YASUKO		ı	Chome	345 Kammachya	Kamakura-criy	KANA	g :	247	March 31 2000
6910 85 0030374 747 9500	YASUKO	41 90 5276866				Japur	3	2 S	82008	December 31 1999
Masubismi recisional Engo, College	G. Ragupaliny		5005 East McDowell			Phoenix	1 2	sa	85008	June 30, 2000
Motoreia Inc	Xeite Vanda		5005 East McDowell	Storioid		Hitchin	HERT	99	\$G5 4AY	December 31, 1999
Motorola Pro	ian Daniel	01462 831169		Schatzbogen 7	Postlach 62 09 60	München		.	0.81829	December 31, 1999
Motorcha Aleta Storiou Co.	Quentin Gum	069 92103 231		Schatzbogen 7	Postfach 82 09 60	Monther		ž	3	
Motorola Grabh	Gotzenberger				2-9-1 Akedon Izum	Sendar City	MIYG	đ,	96132	December 31 1999
2003	YASUKO	121 138 137		Bozeny Nembove 1720		Reznov p A	ì	7 5	CZ-75661	June 30, 2000
Motorola Nippon Se Ots	Jaroffer Skricks	(285) 455-5119		2733 S. Albright Road		Kokomo	Z	n X	70404	December 31, 1999
Molorola Semicraductors	Kin Gauen	(50.)	Sincon Harbour Center	2 Dai King Street	Taipo industnal Estate	. N o L . R		•		
Motorola Sumiconductors Hong Kong			Cantra Electronique de Toulouse		Av du General	TOULOUSE	ΜP	a.	31023	December 31, 1999
Ltd	Christophe OGER				Eisenhower.Le Mital	CEDEX	ΜPγ	æ	31023	December 31, 1999
	Senoger Jacques	9006-011 (99)	Centre Electronque de Lourduse	L	Eleanhower, Le Mirail	CEDEX	ı	1		0000
MUTOROL A Semiconductors 5.A.	ROLLAND	9628-611 (95)	Centre Electronique de Toutouse		Av du General	TOULOUSE CEDEX	Σ	E	57043	Uncommuner 31 1525
MOTOROLA Semiconductors S.A.	Suzarna BIGANZOLI	739-2391			1500 Gateway Blvd	Boynton Beach	Ľ.	S	33426 8292	33425 B292 December 31, 1999
Motorida, hr. Buynton Beach, FL.	Mary Macke	200 (196)	Pand olists a control			Northwille	₹	S	48187	May 31, 2000
	Jell Fleiter	(248) 347-7919	41700 6 Mile Road			Northville	Ī	5	46167	July 31, 2000
Motorola, Inc. Northwile, M.	Vibing Dong	(248) 347:7834	M/S 53	4000 Commercial Ave		Nanthbrook	ي	Š	1840	Uncommuner 51, 1999
Motorola inc. AIEG Northbrook IL	Robert Hou		1750 Belle Meade Court			Lawrencevife	V	S	30043	April 30 2000
Group Energy Systems Group	p Mark Zanhowshi			DE MOORE SCHOOL		Cvandler	7	S	85224	February 28 2000
MOJORCIA: INC. CONT.	CO. SOUR EVOLUK		Time of Solutions apparetion	2100 Filliot Boad		1 empe	7	SO	952B4	January 31 2000
Motorola, Inc. SPS/NCS G/PCD/15U			Design Systems Engineering	2100 E Ellion Fld		Tempe	ZY	Š	85284	December 31 1999
Molorofa, Inc. SPS/S17AD1 Application application SE		(602) 413-5743	Dosein Systems Engineening	2100 E Elica Rd		Tempe	Ş	S, ë	85284	December 31 1959 March 31 2000
Mararala, the SPS/ST-UDSL/DSE	Roy Ellis Yana Taraseriko		12			Nagoya	PICH	4		March 31 2070
Morald Napova uskule of Leginkology SB	YASIJKO									
0.382				PAGE 12						
				1						

PATENT

REEL: 011820 FRAME: 0394

v
ż
_
Ę
ጸ
ŭ
Œ
ance H
×
ξ
5
Ξ
₹.
5
š
ŝ
3
ADI SA
ō
3

170-510 18584

Company Name	Contact Name	Contact Phone	Address1	Address2	Address 3	City	State	Countr	Countr Postal	MaintExpOt
y SR 0229 ingical mist Corporation in Hip Performance	YASUKO Kathy Girh Wang Xiaofirity Chili-Miri Yac	(86025) 342 9900 (8863) 577-6085	School of EE No 24 Cayldaneng 7 R&D	huroucho VI Science Park	chigusaku Nanyang Ave	Nagoya Nanjing Hsinchu	AICH	7 3 8 5 ¥	2263 210003	March 17, 2000 January 31, 2000 March 31, 2000 May 31, 2002
	Bruce Tang Enca Lim	RR 6.1 480 2726	Dept of Erectnical Engirmering 750E Chai Chee Rd #07-03/04	VLSI Circuil Design	Chai Chee Industrial	TAIWAN		SG .	1646	August 31 2002 March 31 2000
	Lian Yong	6698 777 (68)	750E Chai Chee Rd #07-03/04	VLSI Circut Design	Park Cher Chec Industrial			SG	1646	December 31, 2002
Cravie	Russel Crow	(812) 854-6397			300 Hwy 361	Crane	ž	S	47522. 5000	March 31, 2000
Navat Surface Warlare Center Indian	Pablo Lopez	(301) 743 4000	101 Strauss Avenue			Indian Head	QW	Sn	20640.	August 31, 2000
Head, MD Nevisiar int Transportation Corp. Ft	Lynn Frauhger	(219) 461-1302	2911 Meyer Road			Ft Wayne	ž	ns	5036 46803	September 30, 2000
Wayne IN SB-0018/0266	YASUKO			1753 Shimonumabe	Nakahara-Ku	KAWASAKI	KANA	٩	211	March 31 2000
NEC Consumer LSI #3 SB-0064	YASUKO	0211 6503 366		Oberrather Str. 4	nakaharaku	Kawasaki-Shi	KANAX		211	March 31, 2000
NEC Electronics (Europe) GmbH	YASUKO		NEC ichinolsubo Building	499-1 ichinotsubo	nakahavaku	Kawasalu-Sp	¥ Z	7 g	0.40472	March 31 2000
NEC Engra/Denso SB-0318 NET Fund Rsch Lab SB-0163	YASUKO		2rd ASSP Development Dans	ALEC KARANA BIAN	34, mryukopaoka	Isukuba chy	BAR		305	
NEC 1C Micon System / ASSP #2 SB-	YASUKO			יייבר ויייבר איניים מות מיייבר	403-53 Kosugi cho 1- chome, Nakahara-Ku	Kawasaki-Shi	KANA	٩	211	March 31 2000
0251:0270 NEC 1C Micon System / ASSP #3 58	VASUKO		3rd ASSP Devalopment. Dept	NEC Kosug Bldg	403-53 Kosugi-cho 1.	Kawasaki-Shi	KANA	ď	211	March 31 2000
0271/0303	YASUKU		Primers Division		1-10 Nisshancho	Fuchu City	TOKY	٩	183	March 31 Sayon
NEC Printer SB-6041	YASUKO		or Electronics development	1753 Shmonumabe	Nakahara-Ku	KAWASAKI	KANA		211	
NEC Semiconductor Solution SB-0370	YASUKO		System ASIC Division	1743 Chamon amoba	:	Kawasaki	KANA			March 31, 2000
NEC System ASIC #2 #3 #4 SB	YASUKO				Makahara Ku	Kawasak	KANA	طر د	211	March 31, 2000
0020/0237 NEC System LSI C&C SB 0021,0267	YASUKO		System Micro #5	1753 Shimonumabo 1753 Shimonumaba	Nakahara-Ku Nakahara Ku	Kawasahi Kawasahi	KANA	4 4	211	March 31, 2000
NEC System Micro #2 /#5 50:										7007
OCENTRAL SHORES NEC ULSI Bipolar SH 0253 VASUKO	YASUKO		ULSi Systems Development			Sagamhara Sagamhara	KANA	라 라	- ~	March 31 2000 March 31 2000
	Heinz Hagmann	-41 81 7553 372		Werdenbergstr. 4		Buchs			CH-9470	February 28, 2000
New Jedinakuri Duking Newpor News Shipbuilding	Enc Nguyen	(757) 688-6034	4101 Washington Avenue			Newport News	\$			April 30 2000
Newport News Shipbuilding	Jacob Vernulst	9166-989 (151)	4101 Washington Avenue			Newport News	«	OS 2.2.	_1	March 31, 2000
Newbort News Shipbuilding	Monamed	(703) 416-2896	4101 Washington Avenue			Newport News	\$	us S	2776 23607. J	June 30, 2000
Nippon Selki SB-0385	YASUKO					NAGAOKAKYDU	S IS	ة مِ		March 31 2000
Name of Yorkosuka SB 0361	YASUKO				1 natsujimacho	Yokosuka city	XANA.	JF 237		March 31, 2000
Nissan/Zama Works SB-0425 Nissan/Zama Works Ouly FinLAND	YASUKO Sauli Hariatinisis	(35881,550-1556	Elektroniskkalie 10			Zama Oulu	, •	F. F.	D FIN 90570 A	December 31 1999 August 31 2000
. NO preson leton	Yan Fertiu	(613) 753 3937	185 Curkstown Road	Corner of Moodie Dave		Nepean	N _C		KZH 8V4 M	March 31 2000
Northeastern University Buston, MA	Prad Lehman	ora y	Dept of Electric Engineering			Baston	₩ AM		2115 Fe	February 28, 2000
Numbern Auctiong University	Zeng Guanding	} ;				Beying City Beying City	.,		100044 No	November 30, 2002
Northern Junitarig Disversity Northerp Cerporation Hawitherine CA		3101 332 6977			One Nuthrop Avenue	Hawfloris	່ •	15 900		March 31, 2000
				of account						

V.
_
Ş
ē
e P
č
ě
ξ
ž
80
₹
핕
š
_

Company Namo	Contact Name	Contact Phone	Address 1	Address2	Address3	City	State	Countr	ir Postal	MaintExpDt
transco eminan	Lines Husery	(457) 951 5000	2000 West NASA Blvd			Melboume	ď	s o	32902	August 3" 2000
Nontinup Sturman Balbmore, MD	Menael Trumpse	Michael Trumpson (410) 765-3025	Elec Sensors & Sys Div		7323 Aviation Blvd	Battenore	Q	S	21240	March 31 2000
Northern Sourcean Sykesyale MD	Douglas Clark	111C) 552 2909	7301 Sytosville Road			Sykesville	Q ¥	SO	21784	February 28 2000
NPOL INDIA	G Hagupainy	91.80.5276866	Gov of India Minstry of Defense	Research & Development Org.	Naval Physics & Oceanographic Lab	Thrikkakara Kochi 2 t	Ē	Z	- - -	March 31 2000
NS1T (Netap Subhas hist of Tech)	G Regupathy	91 BO 5276866		•	ne a meta de la company	- N		ž		March 31 2000
NTNU Trondneim NOHWAY	Roy Nielsen	(477) 359-4240	Dept of Electrical Power Engry			N-7034 Trondheim	Ę	2		
NTT OPT-Electromics SB 0242	YASUKO			162, shvakata shrane	loukamura	nakagun	BAR		319.11	March 31, 2000
OKI Flecting GaAs SB-0008	O VOCE L		•		asakawavho	Hachoy	ž	<u>a</u> ,	103	March 31 2000
OKI Frechis Integrated Design SB (2001 YASUKO	31 YASUKO		inegraled Design	1-3 nakase makuhan techno-center	minamaku	chibe ony	CHB	<i>و</i>	2 61 :01	March 31, 2000
Opel AG	Hadraud Minz	06142 766151		Bahrnhofstraße		Russelsheim		UE	D 65428	October 31 2000
Opel AG	Thomas Oebi	(331) 428 6071		dannoisinals.	2026s C. 100.00	Russelshern	:	DE	D-65428	October 31 2000
Orbital Sciences Corporation	YASUKO				DAIG KAHIRIN DIAG	Lermaniown	≥	3 ē	20874	March 31, 2000
Osram GmbH	Rene Twardzik	06669 33 539		Ohmstr. 50		Traunreut		3	D-83301	March 31, 2000
Osram Sylvania Devalopment, Inc. Ons Elevator Company	Fred Whitney Joe Merola	(978) 750-1727 (860) 676-6184	1 Farm Springs Road			Beverly Farmington	≸ 5	s o	1915 06032	September 30 2000 March 31 2000
	On the state of	.8141 865.5567	EE Deot & APL	2090 FF Week		,			2526	
Pennsylvana State University PELIGEOT SPORT	Pierre GAUTHIER			3 rue Latécoère	89 d.8	University Park VELIZY	PA 10	S E	16802	June 30, 2000 December 31, 1999
Object Descered Brandt Mand, NY		(914) 945-6423	345 Scarborough Road			MILLACOUBLAY Boardiff Manor	3	91	0610	000 1 1 1 1 000 0
Philips Semiconductors Hamburg			Röhren: und Halblenerwerke	Stresemennalise 101		Hamburg	į	3 8	D-22529	February 28 2000
Philips Seneconductors, Nimegen	A van der Wai		Gerstwag 2	6534 AE Nymegen				ž		March 31, 2000
MOLLAND Philips Semiconfuctors Stadiskanaal	Nick Koper	00 31 5990 32429	Elektronikawag i	9500 AA Stadekanasi				ž		November 30, 2000
HOLLAND	G Backnoathy	91 80 5276866						;		
PIC.	Matthias Grammer			Abgunet 24		Osterode		oe s	0 37520	March 31, 2000 December 31, 1999
PMESUK	Kevn Moses	01689 585151	PMES	Arminge Road		Rugeley	STFF	89	ın.	June 30, 2000
Prince Corporation	Mohana	(616) 394-2412	One Prince Center			Holland	Ē	2	10R 49423	No: ember 30, 1999
	041	341 69 18 76 18	DTII/PINT/MIT	62 Routeverd Vindo		Í				
PSA PEUGEO! / CITHUEN						SEINE CEDEX	Ğ	Œ	92208	December 31, 1999
PSA PEUGEOT/CITROEN	Edwige THOREY	01 56 47 83 29		Charlebourg/La Garenne	18. rue des Fauvelles	LA GARENNE COLOMBES		т ц	95226	December 31 1999
	DAY OF TAXAB	01 % 47 83 23		Charlehoum's a Garage		CEDEX				
PSA PEUGEOT: CITHUEN	raine and raine				IS no des Fauvelles	LA GARENNE COLOMBES CEDEX		<u>π</u>	32256	Dacember 31 1999
PSA Pengrati Olitakn	Frederic QUENTIN (14) 136-2592	4 (14) 136-2592		Roule de Gisy		VELIZY		FR 7	78140	December 31 1999
PSA Porgea Citour	Edwige I HOREY	0. 56 4/ 83 29	DE 1AMSC:1EA	DETA:CAR:EQU B	BP 16	VILLACOUBLAY	ō	G.	F 92250	December 11 1999
PSA VELGORIAL CONTRA	End DEPOLIER	(14) 748-5376	DETA/MSC/16A	DETAICARIEQU B	BP 16	Calombes La Garenna	ğ	9		
120000000000000000000000000000000000000	Section of Appendix	5/18/13/15/17	1282 Evertien in proceeding			Cutambes			2	Dacember 31 1999
the short of the short of a second of the se))		P.10F 14		V Lalayette	- <u>:</u>	* S-	5 /36/+	September 10 9 km
				1 - 4:						

ş
₩
_
~
_
Ä
×
٠,
٠
α
7)
~
=
=
2
C
ď.
=
=
æ
₹
~
⊽
-
S
Ē.
¥.
5
۷.
⋖
_

POPULAR NATION	Contact Name	Contact Phone	Address1	Address 2	Address3	Crty	State	Countr Po	Poetel	MaintExpDt
	100 Page 100	(410) 236-2311	1660 S. Magadam. Suite #290			Colaine) ISC 3	5	er sn	48823	March 31 2000
manning Consollants for	E Lansing, Dave Marin						: :			0000
Mi Quantum Consultants find El Linguing Clinives Sun	g Qinwei Sun	1810, 236 9454	4660 S. Magadom, Sude #290			East Lareing	ž	9	48823	March III 2000
MI CANADA	A P Hose	1613) 533-2804	Bollerell Hall			Kirugston	Š	CA K	K7L 3NG	March 31, 2000
Quena's Unwarsty Onland Childh		(416) 599-8777	Botterell Hall			Kingston	š		K7L 3N6	March 31, 2000
OLDER S UNIVERSITY OF MAIN OF STATE OF		(650, 361.2766			300 Constitution Drive	Manie Park	ა		94025	April 30 2000
Haychem Corporana. Raycheon Systems Co. EDSG E1	Jim Smith	(310) 607 7629			2000 E El Sagundo Blvd	El Segundo	క	S S	90245	March 31, 2009
Seguerin CA Seguerin CA Seguerin Systems Company	Bruce Bozuvich	1805 562-2380			75 Coromar Dr.	Goleta	5	S 93	93117	December 31, 1999
	Buchard King		Queensway Industrial Estate			Glerrothes	FIFE	8 8 5	KY7 SPY	June 30, 2000
Raymein Systems Glerrothes Un Raymein II Systems	Dale Streetman	(972) 344-5074	13532 North Central Expressway			Oakas	×	us 75	75243	March 31 2000
	Dhuren Patel	91 261 228 394	Computer Engrg.			Surat. Gujaral			395 007	March 31, 2000
REC Sural INDIA	G. Regupathy	91 80 5276866				T state		<u>z</u>		March 31, 2000
Regional Engg College (REC)	G. Ragupamy Jean-Charles	01 34 95 73 91	Fectinocerare Renault	1 avenue du Golf		GUYANCOURT	ğ		78288	May 31, 2000
AENAU!	BERLIOUX Phippe VAOUEZ	(13) 495-2919	Technocentre Renauk	1. avenue du Golf		CEDEX GUYANCOURT	οć	FR 78	78286	December 31, 1999
RENAUL		275 306 (TV)	Cyrection des Études et de la	Rus Amadeo	9	CEDEX	440	9	36303	0001 16 391
RENAULT VI	Gaudens BERTOLACCI	6 (50,063 (74)	Recherche			VENISSIECA			5	
Research & Development	G Ragupathy	9099/75 08 16						≥		March 31, 2000
Establishment Establishment Control Indiana INDIA		000 3 Aug 100000 17	205 Phillips St			į	á	≩ å	6	March 31 2000
Resubrith in Motion Lid. Waterloo.	John Wynen	(5) 9888 (40-504	10 dual 1 Cay			Waterloo	3		NZL JW8	Seplember 30, 2000
Ortano SS 0414	YASUKO									December 31 1999
Hiten Science 30 0000	Reinhard Neul			Hobert-Bosch-Platz 1	Poerfach 10 60 50	Stuttgern			D-70839	December 31, 2000
Hobert Busch Gright FV/FLI/FLO	Thomas Presive	0711 811 34854		Robert-Bosch-Sir 2	Postfach 10 60 50	Stuffgard	CWINE	9 6 6 6	0.70639	December 31, 1999
Ruburt Busch GrinbH, K1/3/8/9/FV	Helmul Meyer					Side garran med			5	October 31, 2000
Robert Bosch GmbH, K 1/3/8/9/FV	Thomas Hinz	3711 811 33826		Robert-Bosch-Str. 2	Postfach 30 02 40	Stuffgart/Schwieb	BNWG	.d 90	D-71701	March 31 2000
Good Busch (Settle) KITSHBASHEV	Woilgang Haas	0711 811 20687		Robert-Bosch-Str. 2	Posifach 30 02 40	Stuttgart/Schwieb BruWG		0 30	D-71701	October 31, 2000
Robert Bosch Galbh. K4	Ciaus Schmede	Ciaus Schmederer 07223 82-2791		Hauptstraße 34	Poerfach 1163	Bürkenal		DE D.7	0.77830	March 31, 2000
	1	0211 811.521R2		Warnerstraße 51	Boundary 30 03 40	1				
Hopert Busch Childlet, K5	Trangon Degrer Dieter Jefter			Tubingerati. 123	Position 13 42	Reutingen		0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.72762	September 30, 2000 March 31, 2000
Robert Board Gribar, No Doors Board KarpES	Sven Hartmann			Robert-Boach-Sir 2	Position 30 02 40	Stuffgert	Ş			March 31 2000
Rochester Instrument Systems	Larry Hogg	DEST-067 (01/)			255 North Uraan Steam	Rochesier	ž	US 1460; 2699	<u>.</u>	May 31 2000
CACO	Jerry Roland	(319) 295-4330	CACD	M:S 106-180	350 Calins Road NE	sp.	₹.			April 30, 2000
Hockwell Collins Inc. Melbourne FL		(407) 768-7488				Melbourne		US 32901		April 30, 2000
All posses a succession of the state of the			PO Box 3	Filton						Merch 31 2000
Polis Rovce & Associates Derby UK	¥	71,52,001401	525 Brooks Pla				DBYS		ž	February 28, 2000
Home Laboratory's RDD	Sieve Urage							4505		September 30 2000
Although Called	is J Bhati			Jodhpor Tekra		Ahmedabau	-		360 05.3 N	Marcf. 31 2000
Caf Appropried	Nayan Shelat	2000,000 (0.10)		PO 803 5800						March 31 2000
Sanda habitana aka	Buddy Wans			•		anthactonom	<u> </u>	F2128 - S0		September 30, 2009
				0,000						

age 15

-0
×
_
Ξ
8
ē
Œ
8
ĕ
ā
5
ž
æ
3
5
Đ.
>
ñ
ŧ
2
>

Company Name	Contact Name	Contact Phone	Address1	Address 2	Address3	Cuty	State		Countr Postal	MaintEupDt
	Anna Linieli	-46 8 553 82601						,	Code	
						SE 151 87		ď.		June 30, 2060
Seagate Lechnology Stuffs Valley CA Mituria Cuevas	A Mauricio Cuevas	(831) 439 /161	4585 Scutts Valley Drive			Scotts Valley	*	S	95056	Aug. 31, 2000
Solver Epicon, SB (0398)	YASURU	1004 000 1300 130	0 - 0 - 0			•		ď		March 11 2000
Seiko Epson Corpuration	Vana Larasenko	1994-600 (699310)	HIP-DADD GAD C-C-C			Naganoken		<u> م</u>	392 8502	Fahrian 28 2000
Seiko/Epsan SB 0393	TASUNO Maria Spalan	120,010,0121	3200 Canada Dama			•		q		October 31 2000
	VACUED	eso do (escap)				Sioux Falls	S	S	57107	January 31 2000
Shurp/Opto-Eloctrones SB-0420	Book Was	1682 89 11190						4		December 31 1999
Signeds A&U Franchaumicher	Not Augustia	Not Avadable		r invenauracher Sir 80	Posttach 3269	Enangen		DE	0.91056	December 31 1999
Signitions Ats	Hans-Peles Kasine	Hans-Peter Kasiner +43 1 1707 23985	EWW Elektronikwerk Wien	C.1.35-SAGGK8-SEr. 1.3		Svake		ΣĘ	D-26919	March 31 2000
Sierrens Ald Usiericial Cyvin				Semensell, 66-92		Wen		Y	A-1210	March 31 2000
Signeris Alpharetta, GA	Raed Armad	(770) 740-3692	Energy & Automation Industrial Systems	100 Technology Dave		Alpharetta	3	Sn	30005	December 31, 1999
SIEMENS Automotive	Mane-Françoise	(56) 119-8644		Avenue du Misal	BP 1149	TOULDUSE	Mpy	a d	3900	
Mis Halls Miss Authority Authority Miss Miss Miss Miss Miss Miss Miss Mis	Robert Cook	(248) 253-2729	2400 Executive Hills Drive			CEDEX		:	}	CUMON ID. 2000
Serial Se						Auburn Hills	₹	SO	48326	September 30, 2000
Sierrens Comms Bekslon UK	Mick Goddard	0115 943 4981	Technology Drive	Beeston		Notingham	ALL	6.0	2980 NG9 11 A	March 1. 2000
Signers Corporate Research				755 College Road East		Princeton	₹	S	08540	March 31 2000
	Don Bardett	(970) 204-0057	2421 Sumburgi Drive				1		2099	
Singapore Polytechino	Elic Cheung Lar Yip 65 772131	p 65 772131	Electronic & Comm. Engrg. Dept	500 Dover Road		Singapore	8	s c	80525	July 31, 2000
Charles Alon decomposition	Philip S.M. Chin	65 772131	Electronic & Cornm Engra Depr	500 Dover Board		, I		:	3	Jan 1941, 2000
Singapore romedime	•	•				Singapore		SG	139 651	January 31, 2000
Singapure Polytectinic	Rayindra Joshi	65 772131	Electronic & Comm. Engrg. Dept	500 Dover Road		Singapore		SG	139 651	January 31, 2000
Singapore Polytechnic	Xia We	(65) 772-1192	Electronic & Comm. Engry. Depi	500 Dover Road		Singapore		56	130.641	0000
	Okvier	(14) 018-6812	Direction de l'Ingenene	162 cum d. E. t						January 31 2000
SNCF	HOLT?MANN			Marin		PARIS CEDEX 10 IDF	10 (DF	FR 7	75475	June 30, 2000
SODERN	Enc JEUDY	(14) 595-7129		20, ave Descentes		CIMER.	Ş	FR 9	94451	March 31 2000
SRON MOLLAND	Piet de Groene		Sorbonnelaan 2	3584 CA Utrecht		BREVANNES				
ST Microelectronics	Rossi Augusto		Via Tolomeo. 1			CORNABEDO	7	; ≓ ‡		December 31, 1999
The state of the s	Yans Tarasenko	(617225) 009 4221	Via Folomeo, 1			(M)			1 01002	December 31, 1999
S MICLORIBERITY III.5	•	990 09 00000				CORNAREDO	LMBD	17 2	20010	August 6: 2000
ST Microelectronics	Perer birkeristing	007 60 76000		Haidking 17		Grafing / Munchen	_	0	0.085667	
Stryker Corporation	Chris Philippo	(6163237) 700-3332	Instruments	4100 East Meham Avenue		Kolamaroo	ŝ		;	200 (10 company)
SUPELEC	Bilding LORCET	0101-006 (011	Z E I	Plateau de Moulon		CHE CHAIN WATER		S (November 30, 1999
	Jaan-Pierre	(13) 085-3900		O September 19		CEDEX			, 20:18	January 15 2000
STLEA	BAUDRY			A STATE OF THE STA		MONTIGNYLE	10F	FR 78	79180 N	November 30, 1999
System Design Concepts, Inc.	Joe Troutmen	9008 393 (609)	Suile 106			WAS TRACES				
SZ Tesisysteme AG	Walicand Hadi	08075 17-249	Entwicklung Software	Wasserburger Str. 44		TTEN INCHES	₹			December 31 1999
Technische Universität Gresden	Andreus Binner	0351 463 3379	a . 1	Heimholtzstr. 18		Orenden				March 31, 2000
	()	BOSC CALLESTI		Barkhausenbau			-	ă	P 69015 O	July 31, 2000
Technische Universität Bresden	Tail Colles	200	_ 00	Hekmholtzein 16. Barkhauseoben		Dresden	_	0 30	et. 69010 0	COOK 31 VOICE
Manager terrorisated Manager	Oliver Alesenor	WIS 289 28519		Arcisstraße 21						0007 10 4
Egylins the Universitat Wien	Fuir timmelsloss	+131 58801 35977	.,	Gusshauseir 27.29		Munchen	u			November 30 1999
	Singapor di con c	.131.562:4036	-	4			•	- 4	A 1940 Se	September 30, 2000
TELECOM Paros	PLAVINE B		•	Manage of or		Pans CEDEX 13	ان ا	o5. Hd	.50 84 De	December 11 1993
				Page 16						

PATENT

59de: 018-051

REEL: 011820 FRAME: 0398

Ś
*
Ξ
ŏ
Ģ.
ž
تة
ü
Ē
õ
Ξ
ē
≥ .
8
3
5
Ξ
š
•

Company Name	Contact Name	Contact Phone	Address 1	Address2	Address3	Clty	State		County Posts	
	to across Hearthard	9341.881.2405				•			Code	Maintexput
FEMIC Telebrokan Microelectronic				Hinglerstr 17		Ingolstadt		DE	D-85057	14.5 11 July
Towas instruments Dallas	David Newman	1972: 486 2426	P O Box 66)199			ı				7362 Ausc
TOKALLIMVEISITY SB-0053	YASUKO		i		emeded to 7 (i	Dallias	ž		75266	December 31 1999
Toppun Printing Yaesu SB 0235	YASUKO		Electronics Division		2.2.7 Yaesu	Chicatoura coly	KANA		259-12	June 30 2003
Lushiba Corporation Fuchu Works SB-	YASUKO				1 Toshiba-cho		֝֞֝֞֝֓֞֝֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓֓		104 0028	July 10, 2000
0151/50 0152	3		Control of the second second	,		r denu-Jan	Š	<u>a</u> ,	183	March 51 2000
Tushiha isogo 58-0006	YASUKU		Crima motiva Daveroproent Centor 33 shin-osogocho	ilor 33 shin-osogocho	naogos:	yokahama city	KANA	ď		March 31, 2000
Tushiba LEC, SB-0418	YASUKO					3		ţ		
Toyo Denkind'i Factory SB-0220	YASUKO				4-6-32, tougava	Photogram				December 31, 1999
Toyota Auto Lourn Works SB-0043	YASUKU	1000 1007			2-1 lovolacho	Kennes Off.	A		243.04	January 31 2000
Trinitio Navigation Sunnyvale, CA	Mark Nage	(234) 366 4671	See Design & April			Supposeda	Į.	L 4	4 .	December 31 1999
TRW Automotive Livoria Mi	Wex Zhan	07733 8/19 947	of the Control of the	12000 Tech Center Drive		Livoria	5 :	3 5	90	June 30, 2000
TRW Fahrzeugelektrik Galut & Co. Kis Offich Backer	Asset Make	(248) 442-5298	Transportation Electronics	Industriestr 2-8	Positach 14 20	Radolfrati	Ē	3 6	98130	March 31, 2000
TRW, inc			Division	ATT CHESTON DRVE		Farmington Hills	₹	รร	48335	January 31, 2000
Unitrode Corporation	Paul Bompastore	(603) 429-8637	7 Confinental Blvd			Marriemack	2	ç.	2642	•
	Asset Treat		Southeast Design Center				Ē	3	03054	June 30, 2000
Unitode Corporation				071 200	111 Coming Road	Cany	Š	S	27511	June 30 2000
Unitrode Corporation	Anh Nguyen			4010 Moorpark Avenue	#220	San los	ć		1626	
BRAZIL	Dr Danzon						5	3	95117. 1843	June 30, 2000
	Andrade				UFU - Campus Senta Mension	Uberlanda MG		84	3 902	March 31, 2003
Unv Federal de Uberlandia BRAZIL	Herivello	55 B1 229 4087			UFU Campus Santa	(Bedeched)				
1/2 # CB	Jose Carlos de	55 34 236 5099			Menica			ž	M400 902	October 31, 2000
Unv rederal de Dografiula, unación	Okvera				UFU Campus Santa Monea	Uberlanda MG		H8	38400-902	March 31 2003
Univ Forteral de Uberlandia BRAZIL	Prof. Editherlo Tenxeira				UFU - Campus Santa	Ubertanda MG		G	18400 902	38400 902 Marrh 31 2003
Univ Foderal do Rio de Janeiro	Edson Watanabe	(5521) 260-5010	COPPENFA	Power Electronic Group	Chixa Postal 68504	P. C.	ŝ		!	7007
	Design Monthen	(803) 777-6784		EE Dept		Outer ne neuron	Z Q	89 84	1945-970	21945-970 June 30, 2000
Unv. of South Carolina, Counting, 30 UNIVERSIDAD DE LA RIOJA		(3454) 129-9496	Centro Cientifico Técnico	Lus de Ulloa 20	Elec & Computer Editico Politécnico	Columbia Lograño	ეგ ქ	US 2	29208	September 30, 2000
UNIVERSIDAD DE ZARAGOZA	Abelardo	-34 976762111	Centro Politiecnico Supenor de	Ci Maria de Luna, 3						June 30, 7000
	MARTINEZ Andres I LOMBART 976 75 1925	T 976 75 19 25	Ingeniero Centro Politechios Subenor de	C. Mana		ZAHAGOZA	A BGN	ES	50015	June 30, 2000
UNIVERSIDAD DE CARAGOZA	ESTOPINAN		Ingeruero	V. maria de Lista, 3		ZARAGOZA	AAGN	ES S	50015	June 30 2000
UNIVERSIDAD PUBLICA DE	Pero G4 12CO	34 9 48 168 255	Opto Automatica Electronica	J	Pous de Arrogadia	PAMPLONA				
NAVARRA	Werner Dierker	0511 762 3769	institut for Leistungseleittromik	Appeter 9a			¥	2	31006	June 30, 2000
Universitati Hannover Mikroeledronik	Rail Popp	0511 762 19654	Inst. für Mikroelektromsche	Appelatr 4		Hannover	٠			November 30, 1999
	Stelan Butzmann	0234 700 6652	Bisancia	: : : : :		Hemover	0	÷ 0 30	0-30167 04	December 31 1999
University Bothum	Dagmar Peters	0421 218 00	FBI Elektrotechnik	Kufatener Str. 150		Bochum	٥	DE D'	D-44780 Ca	Contamber 20 2000
Louversilal der Burdeswehr	Oskar Kowarik	UB9 6004 365 5	Fakultat Elektrook	MO.Wen	Positech 33 04 40	Bremen	8			December 31 1000
Linversity der Bundeswehr	Manfred Ruber	040 6541 2766			Post of the Post	Neubberg	8			Auto 20 Serio
Universität Duisburg	Georg Pelr	5203 3783 295				Hamburg	20			November 30 2000
Universital Edangen	Haraid Grams	91.31 65 51 16.00		Caueratr. 6		Dinsburg	PE			June 30 2000
Universität Kaiserstäutern	Sommer Workname Fresher	6391 671 2462		ij		Enangen Kaiserslandern	ä k			June 30 2000
Universitat Magdecurg	Michael Sestement 0711 21 4180	n 6711 -21 4180	Lehrstahl nir Mikrosystemiechoik	Universitatisplatz 2	Postlach 412C N	Madeture	7 2			March 31, 2000
Programme Stuttgard				Prenscheidstr 25		Stuttona	מ ב			June 30 2000
							2		0.70174 Jun	June 30 2000

Page 17

S)
=
•
て
Q
Ö
≘
I
بە
≗
۲.
Ō.
٠.
=
~
Ē
₹
€
¥
ō
>
≥ .
9
=
۳
S

Na Ma	Contact Name	Contact Phone	Addrags 1	Address2	Address3	Ě)		County Posts	3,000
1						ì			400 0	
UNIVERSITE DE BORDEAUX 1	Painck Vit I ESUZANNE	:55) 684 8345	LABORATOIRE IXL ICNRS)	351, cours de la Liberation	<u> </u>	TALENCE CEDEX	Ψ.	H.	33408	686. 18 Jequia ca d
	Married Married	1013 948 6101	434UU Serdand Sejannov							
Linversali Futra Malaysia	Xo. s.d Actam	(60.1) 348-6101	43400 Serdang Selange					¥		May 31 2000
University Pullin Malaysia	Robert Yacarian	353 21 90	Dept of Electrical Engineering	& Microelectropics				<u>}</u>		May 31 2000
University College Con their man	Antrea Forsyth	(21, 414.3495	School of Electronic & Electrical	Enghaston	University College	* 0		<u>ښ</u>		November 30 1999
University of Burnershild III University of Burnershild III			Eng			Bernmyham	Š	99	B15 21T	February 28 2000
University of Birmingham UK	Dr. C.J. Goodman	(21) 414-4287	School of Electronic & Electrical Eng	Edgbaston		Birmingham	3	6 9	815217	February 28 2000)
Angeles.	Ellici Brown	(310) 206-3102	a i		2567 Boetter Hall	1 05 Anneles	5	ý.	30000	1000 SC 1000
CA CONTRACTOR OF THE CONTRACTO		67 65 65 65 65					; }	3	1600	may 31. 4000
University of California Trus Angeles.	John Yer	(310) 206-2642			2567 Boelter Hall	Los Angeles	ď	US	90095	May 31 2000
CA Common Berkeley	Jacques Milland	(510) 486-4169			2700 - 7th Stone	9	ċ	<u>.</u>	1600	
Unversity of Camurica Const.	•					Deinge	5	ß	94/10	February 28, 2000
University of Florida Gameswille, FL	Khai Ngo	(352) 392-8934			334-B Larsen Hall	Gamesville	ದ	Sn	32611	July 31, 2000
Month of Glasgow UK	Calum Cossar	0141 330 6003	Dept of Electronics & Electrical	Rankine Bidg		Glasgow		89	2044 G12 BL1	January 31, 2001
	31	01707 284151	Haffeeld Campus	odian I						
University of Hanfordshire UK	Harry Jeneries	(8522) 859-2168	Electrical & Electronic Engra	CYCB 401		Hatted	HERT		AL 10 9AB	July 31 2001
University of Mong Kong	Georges Gielen	+321 632 1047	Kardinaal Mercienaan 94	B-3001 Heventee	TOKICLE HONO			Ť		December 31 1999
University of Leuven (AUL) DELLEGORE	Wong Man Chung		Faculty of Science and	FST University of Macau		Macau		ב ב		December 31 1999
		0239 754 15677	lecthology 1342	Tapa Macau		•)		
University of Michigan	Gerdy Canchille	מרום אינו לפרון	767	siec engrg & Computer Sci	1301 Boal Street	Ann Arbor	Ē	ns	60187	September 30, 2000
Silogenation excession with the control of	Amit Kumar Jain	(612) 625-4583		2-130 EE/CSCI	200 Union Se SE	Monage	4	9	2122	
MN		1001	20 mg : 20 CCC v. 1 344		5		į	ŝ	66960	Jay 11 2000
University of Nissour Rolla	Steven Pekalek	61 49 21 6082	Flectical & Computer France			Rolla	9	SO	65401	September 30, 2000
University of Newcastic	Harry Daniel	(115, 951.5545	Dead of Electronic	University Drive		Callaghan	NSN	₹	2308	February 28, 2000
University of Nothingham UK	Chastopoulos		Eng	Nie Andrew		Nottingham		98	NG7 2RD	October 31 2000
Section Of the section of	Ziwen Yao	(819) 376-5219		Dept of Electrical	3351 Bd DesFormer	Trace Baragas	8	č		,
Craversity of Coopers	Phy Melior	0114 2225196	Dept. of Electronic & Electrical	Mappin Sireel		Sheff-ald	?	5 8	GSA-SH?	October 31 2000
The state of the s			Eng.					9	25	September 30, 2000
Usinversay of Tennessee	Charles Wilson	(901) 448-5233	The Health Science Center	Anatomy Dept 528	855 Monroe	Метрия	ž	SN	38163	December 31 1999
University of Lorento Ordano	A. Salama	(416) 970-0030			10 Kings College Road		N O	ŏ	M55 1A4	January 31 2000
University of Toronto Ontand	Alguin Porture		Dept of Computer Spence		10 Kings College Road	Toronio	N _O	₹		September 30 2000
University of Wales, Aberystwyth UK	Chris rince	(60d) 265-3878	1500 Engineering Drive	811		Abenyatwyth	DYFD	GB	SY23 3DB	June 36, 2002
University of Wisconsin Madison, vvi	Mark Dischett	(405) 736-4198	OC-ALC	3001 Staff Desca		Madison	š	Sn	53706	December 31, 2000
US Ar Force - Tritler AFB					Damong 1083	Inker AFB	ð	C.S	73145	March 31 2001
VALEO Ficetronique	Hervé GUEGNAUD (14) 513-7774	0 (14) 513-7774	Centre d'Etudes Etertronique	2. avenue Fernand		CRETEIL CEDEX	Ō	ď	94042	December 31 1999
	1	ACOT CAR TAN		rouling						
VALEO Flectronique	Patrick PFRU	(14) 513-7034	Contra d'Elioces Electrongue	4. avenuc Fernand Poutton		CRETEIL CEDEX	ίΟ	Œ	94042	December 31 1999
(Kowalsch	06195 802 474		Gundelharter 72						
Varia Batterie AG	Ursula Teutsch	06196 564673		Guintoffuntstr 72		Kelkhein				December 31, 1999
Vada Ballerie AO		(248) 478-7281			37800 Hills Tecr. Dree	Kelkingum Farmiootoo Hills	ž		6	December 31 1999
5		6				S MILL TO STATE OF THE STATE OF	ž	20	4833. D	December 31, 1999
Vikturn Studdhy Spin e Centre (VSSC). K. Navsen Kumaf.	ij k Naveen Kumaf	(9147, 156-2373	CAD APLO ELG			Vel., Trivandrum		Z Z	695 U22 N	March 31 (600
INDIA	A Eishabra-Riad	(540) 231 4465		•	18 White					
Virginia Polytecholo matmini	שטורים שטר	01266-495100		ראיטשטא	STO VVINIONNOTE HOM	Blacksburg	∀ 4			Acvember 30 1999
Visigori (1981 According	Schoesser	0221 5496 550	Rody Fredhamas & Security	**************************************		Basildon Kan			ш	Occ8mber 31 1999
						I W.C.		i. S	() () () () () () () () () () () () () ((-ecembel 31 - 135
				P. raje 18						

P. tyte

PATENT

				Six IIOdayaanini ahaan	ş					
Company Name	Contact Name	Contact Phone	Address1	Address2	Address3	C _M 2	State	Coch	Countr Posts!	MaintExpDt
513 - Anthomolean System F1C	Sukiwi Gunawan	(213) 246 6283	Electronics Division	17000 Hotunda Dave		Destribons	3	. sn	18121	December 31 1992
Blag Visteon Automotive Systems	Hoshi Tanguch	551161465-9281	CAD/CAE/CAM Design Services - Avi Orlanda Bergamo 1000 - Cumbica	Av. Orlanda Bergamo 1000 - Cumbica		Guarumos	SPLU	88	07220-901	0/220-901 December 31 1999
Guarathos, Sau Pate	Suerx Sustano	7341 458-0306	22642 Nash			Dearborn	3	S	48124	April 30 2000
Visiteds Fracing, Community Institute:	G Angupathy	91 80 5276866						Z		March 31, 2000
O TITO	Andress Lübke	05361 9 29448		Berliner Ring 2		Wolfsburg		DE	D-38436	October 31 2000
Valkswagen AG	Henrik Backe	(3) 159-5583				S-405-08		SE		December 31 1999
VOIVO CAT SYNEDEN	•	C3E 3 307 10 34.				Goteborg		į		
Volvo Car SWEDEN	Solen Bomsted	1000 500 15 044				S-405-08		מ		August 31, 2006
	Peter Tempkn	+46 31 772 4431				S-412 BB		SE		March 31, 2000
Volvo Research SWEDEN		-				Göleborg				
Name Trans SWEDEN	Goran Boklund	(3) 166-0000				\$ 405 08		SE		December 31 1999
	David Mumford	(604) 718-2053	1691 West 75th Avenue			Vancouver	9	5	V6P 6P2	hav 31 2000
Westport Imavations, Inc.					Whiteheld	Bangalore	l I	Z		March 31 2000
Wipro GET smiled Ballgalore, inch.			Lutton Coud	Bernard Terrace		Edinburgh		95	EMB 9NX	August 31 2000
Wolfson Microelections of	YASUKO		ı			•		ā		March 31 2000
Yazahi Meter Surusiyi Yazahi Meter Surusiyi		(734) 983-2110	6700 Haggerly Road			Carrion	2	S	48187	December 31, 1999
Yazaki None Amerikan SB 0069				•	2-9-32 Nake-Cho	Museshing	TOKY	౼	9	March 31, 2000
Yokngawa Elekult Corp. Co.	Ivan Jadric	(717) 771-7476		631 South Richland	PO Box 1592	York	Ā	S	17405	December 31 1999
	Zhao-Mino Oran	(86571) 795-1891	Power Electronics Rsch Inst	20 YuGu Road		Hanozhou		Š	1502	1000 OF acut
Zhejiang University China Zhengzhou University	Guoyu Hu				75 Daxue Lu	Zhengzhou Henan	5	S	450052	March 31, 2002
Zhangany Felecom Lid	Tang bankang Liu Yin	(86733) 649-8350	Locomotive Research Institute	Transong Rd, Transon		Zhuzhou City	H CN	8 S	412001	February 28, 2002 December 31, 2000
CHINA 7.Junglic Suit Fech Dev Curp CHINA Bo Li	VA Bo Li	(86106) 322-4320	Chinese Academy of Railway Science	2 Dalwshu Road	Xichimenwa	Bering		S	100001	June 30, 2000

Schedule 2.21(j).

Softwarelizienvertrag between Robert Bosch and Analogy GmbH dated October 1, 1994

Software License Agreement between Electronic Data Systems Corporation and Analogy, Inc. dated January 1, 1992

Software License Agreement between Volvo Truck Corporation, Volvo Car Corporation and Analogy, Inc., dated September 30, 1997.

Source Code License Agreement between Kanematsu Electronics Ltd. and Symmetry

Design Systems dated February 1, 1996.

Government Development Agreements (limited right to obtain source code to projects funded):

B-2 Depot Analog Device Model Development Agreement between the Department of the Air Force OC-ALC/DIR of Aircraft and Analogy, Inc. dated June 27, 1996 and amended sequentially through June 18, 1998

Advanced Technology Program Cooperative Agreement between the National Institute of Standards and Technology and Analogy, Inc. awarded August 31, 1995 and as subsequently amended.

Cooperative Agreement under 10 U.S.C. § 2358, between the United States of America U.S. Air Force, Air Force Materiel Command, Rome Laboratory PKPF and Analogy, Inc., dated September 4, 1996 and sequentially amended through September 11, 1998.

Open Access Agreement between GEC Marconi Ltd (MES) and Analogy UK, dated September 24, 1998 and as renegotiated October 25, 1999.

60488/4/TVW/052859-0047

Schedule 2.21(k).

Software License Agreement between Acutex Division—SPX Corporation AA, dated September 4, 1997.

Software License Agreement between Adtran, Inc. and Analogy, Inc., dated June 10, 1998.

Software License Agreement between AMP Incorporated and Analogy, Inc., dated October 24, 1997.

Software License Agreement between Breed Technologies. Inc. and Analogy, Inc., dated December 12, 1997.

Software License Agreement between GE Industrial Systems and Analogy, Inc., dated December 15, 1998, amended August 26, 1998.

Software License Agreement between GE Appliances and Analogy, Inc., dated June 26, 1998.

Software License Agreement between GM Powertrain and Analogy, Inc., dated January 17, 1998.

Purchase Agreement, Software and Related Services, between Intel Corporation and Analogy, Inc., dated June 15, 1997, amended August 14, 1998.

Software License Agreement between The Valeo Group (a.k.a. ITT Automotive) and Analogy, Inc., dated October 2, 1998.

Software License Agreement between KAPL, Inc. and Analogy, Inc., dated October 1, 1997.

Software License Agreement between Liebert Corporation and Analogy, Inc., dated June 23, 1998.

Volume Purchase Agreement between Northrop-Grumman Electronic Sensors and Systems Division, dated June 1, 1999.

Software License Agreement between Tenneco/Newport News Shipbuilding and Analogy, Inc., dated August 26, 1998.

Software License Agreement between Varity Kelsey-Hayes and Analogy, Inc., dated March 24, 1998.

Software License Agreement between York International Corporation and Analogy, Inc., dated December 16, 1998.

60488/4/TVW/052859-0047

Software License Agreement between Zuken, Inc and Analogy, Inc., dated October 1, 1998.

Connections Partner Membership Agreement between the Company and Cadence Design Systems, Inc., dated August 1, 1996.

UK:

Open Access Agreement between GEC Marconi Avionics (Holdings) Ltd and Analogy UK, dated April 1, 1997, as amended March 31, 1999 and June 30, 1999.

Open Access Agreement between GEC Alsthom ERC and Analogy UK, dated December 19, 1997.

Open Access Agreement between GEC Alsthom Traction Ltd. and Analogy UK, dated December 19, 1997.

Open Access Agreement between GEC Marconi Radar & Defense Systems Ltd., Dynamics Division and Analogy UK, dated January 1998.

Open Access Agreement between GEC Marconi Ltd (MES) and Analogy UK, dated September 24, 1998 and as renegotiated October 25, 1999.

Open Access Agreement between Marconi Avionics GMEO - extension of GEC Marconi Avionics and Analogy UK, dated September 24, 1998 and March 31, 1998.

Open Access Agreement between Lucas Aerospace and Analogy UK, dated December 19, 1997, as amended March 31, 1999.

France:

License D'Utilisation de Logiciel between PSA/ Peugeot-Citroen and Analogy SARL, dated March 20, 1998.

Germany:

Softwarelizenzvertrag zwischen Analogy GmbH und DaimlerChryslerAG, vertragsbeginn 01.04.1999.

60488/4/TVW/052859-0047

Schedule 2.21(l).

B-2 Depot Analog Device Model Development Agreement between the Department of the Air Force OC-ALC/DIR of Aircraft and Analogy, Inc., dated June 27, 1996 and amended sequentially through June 18, 1998.

Advanced Technology Program Cooperative Agreement between the National Institute of Standards and Technology and Analogy, Inc. awarded August 31, 1995, and as subsequently amended.

Cooperative Agreement under 10 USC 2358 between The United States of America US Air Force, Air Force Materiel Command, Rome Laboratory PKPF and Analogy, Inc., dated September 4, 1996 and sequentially amended through September 11, 1998.

Agreement for Product Development Services between Tsinghua University and Symmetry Design Systems, Inc., signed December 4, 1996, amended September 15, 1998 by Tsinghua University and Analogy, Inc.

Development Agreement between FuDan University and Analogy, Inc., dated May 29, 1997.

Agreement for Product Development Services between Analogy, Inc. and the National Research Centro for Applied Power Electronics of Zheijang University, dated July 1, 1998.

60488/4/TVW/052859-0047

2.21(m)

See QA Incident Report—10/27/99 attached

QA Incident Report

Туре	Problem Report
Severity2	Critical, Major

Count of QAId		Severity	Status Grou	Sched.Rel.		
		Critical		Major		Grand To
		Open	Deferred	Open	Deferred	
ProductGroup	Product Name					
AIM+AIM2	AIM	1	4	2	35	42
	AIM2	6	2	11	26	
Documentation	Documentation: Co	Ų O	0	16		
Integrations/FWs:	F/W - CUI	0	0	3		
	F/W - EDIF) 0	0	0	1	1
	F/W - SITK	2	0	0	13	, ,
	F/W - Sketch	2	0	o	(4
	F/W-Cadence	0	0	1	1	
	F/W-Mentor	1 0	1	0	1	2 2 10
	F/W-Viewlogic	1	1	2	6	10
	Saber/ModelSim	0	1	0	4	5
	Saber/Verilog	0	1	o	1	2
	Saber/ViewSim	0	1	o	3	4
icensing/Installation	Install Tool	3	0	3	12	18
y	Licensing	4	1	6	9	20
Modeling	Component	0	0	3	1	4
•	NSPITOS	ol	0	1	ol	1
	Symbol	1	o	2	o	3
	Table Look-Up Mo	ol	o	0	1	1
	Template	o	o	7	4	11
SaberApplications	Inspecs	0	0	1	0	1
эдоол фриодионо	LSA	ol	0	0	3	3
	SaberBundle	o	0	4	1	5
	SaberGuide	1	0	12	4	17
	SaberHarness	18	o	49	5	72
	SaberLink	1	o	1	3	5
	SaberScope	4	2	33	9	48
	SaberSketch	8	3	16	9	36
Simulator/MAST	MAST	O	o	0	54	54
SITTUIA(OI/IVIAS I	SaberSimulator	0	4	. 1	55	60
ibrary Tools	Parts Gallery	Ō	0	2	0	2
LIDITALY TOOLS	Property Editor	1	0	4	0	5
Grand Total	It topolity conta	53	21	180	308	562

CONFIDENTIAL AND PROPRIETARY
TO ANALOGY INC.

10/27/99

Page 1

ProductGroup	Product Name QAId	e CAId Type L				1				O A C	chase	66/92/01	10/26/99 No
AIM+AIM2	ΑįΜ	23173 Problem F		Engineerin New	ě	To Be Deli 1. Critical	1. Critical	180		da Guene	chart.	06/20/8	ON 90/2/01
AIM : AIM2	ΑIM	22194 Problem /	1	Engineerin New	e.	et		TBD	5.1.0 17	OA Oueue	e tanfels	8/10/4 9	8/10/49 No
AIM+AIM2	AIA	21959 Problem #	ge does not include scroit bars on long	Engineerin New	ě¥	2 High	2. Major	081	5 10 16	200 AD		S S	
	7114	18447 Problem	nessages Scope crash using OSE 9800	Customer Deferred	eferred	To Be Defi 2 Maior		:TBD	432	Gazetta, Marialistair	arlalistair	1/19/99	8/5/99 Yes
AIM+AIM2	SIX.		•	-								٠	-
AIM. AIM.	AIM	17304 Problem A	failed request: 91	Customer Deferred		To Be Defi 2 Major	2 Major	T'B0	6.4	Gazetta, M	Gazetta, Maricallahan	11/17/98	3/2/99 Yes
YIM + YIM T		-	(X. QueryColors)	į						:		00000	00000
AIM AIM2	AIM	14698 Problem (14698 Problem Cannot print from Sketch/Scope on wind, an ultra Beaverton Deferred	Beaverton D		4: Low	2 Major	TBD	4 Ω	Gazetta, Mari bkelly	aribkelly	3/20/98	ON 66/2/6
			Sparc 1 machine	!			,	1	_		200	974/08	ON 10/1/01
AIM+AIM2	AIM	14241 Problem	lified locations of sketch	scope.log Beaverton Deferred		3. Medium 2. Major	2. Major	180	4. G	Gazetta, Mari Dretty	ar Dreiny	06/4/7	01106/1101
		.	- -	-21		!	:			(- Capadaga	1737/08	2710,00,017
AIM AIM2	AIM	14090 Problem	pods the wrong value	Engineerin' Deferred	elerred	3: Medium 2: Major	2. Major	180	E 4 .	'Bedrosian, 	Bedrosian, Diagocuros	0611311	011 06/01/7
			for -inf slope waveforms	!				1	· (1	of this	10/08	2/3/08.No
AIM+AIM2	AIM	13976 Problem	13976 Problem AimBusyDialog collides with self	Engineerin Deferred		2. High	2. Major	180	2 4 2	Murdock, Sc. pomier	C. pomer	70/11/1	ON 190/6/6
AIM. AIM?	N. Y	13257 Problem!	-	Engineerin Deferred	•	2. High	2. Major	180	4 0	Murdock, 2	Scotling	16/11/97	ON 96/6/2
A IA 4 A IA 7	A.	12754 Problem	sn't throw an exception	Engineerin Deferred		To Be Defi 2.	2. Major	180	4.2.1	Murdock, Scopolnier	copolnier	18/82/6	2/3/98 No
AIM AIMS	AIM	12744 Problem	12744 Problem 0 bytes in schematic file	Engineerin Deferred	eferred	To Be Defi 2	2 Major	180	4.1	Russell, Ro	Russell, Rob(edmundp	9/26/97	ON 66/11/8
AIM AIMS	AIM	12707 Problem	12707 Problem Sketch hangs when invoking aim help form	Test Engin Deferred	eferred	To Be Defi 2.	2. Major	180	4 3	Murdock, S	Murdock, Scc CBrowne	76/61/6	5/3/98/No
AIMIT AIMIE			command line.	-									
4114.01143	MiA	12465 Problem	12465 Problem inconsistent plotfile timestamps + millenium	Engineerin Deferred		To Be Defi 2. Major	2. Major	180	Undefined	Undefined Gazetta, Marichinsc	arichnsc	8/20/97	8/3/99,No
AIM+AIMS			problem			_							
2000	MIM	12125 Problem	12125 Problem Scroll bars and Icons stop working with age.	Customer Deferred		2. High	2. Major	TBD	4.1.2	Murdock, Scopconley	cd pconley	6/20/97	2/4/98 Yes
AIM+AIMZ	NIA.	11915 Problem	11915 Problem Macro Recorder problem	Beaverton Deferred		3: Medium 2.	2 Major	180	Undefined		Scopkelly	5/19/97	5/3/98,No
AIM+AIM2		11786 Problem	11786 Problem expr gives syntax error on string compare	Engineerin Deferred		4: Low	2. Major	TBD	Undelined	Undefined Murdock, S	Scopothier	5/1/97	2/3/98 ₁ No
AIM+AIM2	. T	11776 Problem	11776 Problem aimcal corrupts index	Engineerin Deferred	berred.			180	Undefined Murdock	Murdock, S	Scq pothier	4/30/97	2/3/98 No
AIM+AIMZ	Ž	11576 Problem	11576 Problem Questions on NT permissions. What permissions Field AE		Deferred	2. High	2. Major	TB0	Undefined	Undefined Murdock, Scd callahan	cecallahan	3/25/97	2/4/98·Yes
			do an aim mkdir cause on an NT. Where do the										
	• • •		permissions for the out files come from.										
	_			-:			!	:					1 00000
AIM, AIM2	AIM	11550 Problem math tst		Test Engin Deferred		4 Low	1. Critical	TB0	Undefined	Undefined Gazetta, Mar(rlau	ar(dau	3/11/9/	07/99/5/6
AIM, AIM2	AIM	11432 Problem	11432 Problem Form: AIM commands need man pages	Engineerin Delerred	Jelerred	3. Medium 2. Major	2. Major	180	Undelined	Undelined Muraock, Scientisc	CCCIIISC	10,407	ON 00/91/0
AIM AIM2	AIM	11331 Problem	11331 Problem French Keyboard doesn I support or etc.	Customer Deferred	eferred	3 Medium 2	2. Major	180	4.2	Gazena, Marigrey	Sec Great	16/47/1	5/4/02/NO
AIM+AIM2	AIM.	10953 Problem	10953 Problem Core dump when trying to create Aimmessage	Engineerin Deferred	Jelerred	3: Medium 1. Critical	Calical	081	Undelined	Murdock, 3	Undelined Murdock, Schugbeurgs	200	Okt. 15/1-/0
			dialog				1		, ,	Code Control of the C	ocapo de la companya	11/15/96	9/11/99 ¹ No
AIM+AIM2	AIM	10936 Problem	10936 Problem Changes to text line Width causes text to	Engineerin Delerred	Jerred	4: Low	2. Major	2 8	Cudenned	MUIDOCK			
			disappear	Loss Cario Cario	Polocial				- Constitution	Disagn energy position	. Direction	10/19/96	ON: 66/6/8
AIM+AIM2	AIM	8353, Problem	8353 Problem Stack Iface effo Wileli Open Gesign will long pain	1161111601	Delealed	J. Medium	z. Major	2	-	Care and	?		
:	,	on 70 Broblem	Possible Macro recorder uses absolute Graph names	Customer Deferred	Deferred	To Be Deli 2 Maior	2 Maior	TRO	Undefined	Undelined Weinerman, Juwek	Luwek	10/12/96	8/3/99 Yes
AIM+AIM2	N I	6272 Froblem	6272 Frighten Only able to reload plottile in Scool once	Field AF	Delerrad	To Be Defi	Major	TBO I	Undefined	Gazetta, Mari gestep	arquestep	10/11/96	8/3/99 No
AIM+AIM2	AIM	8254 Problem	er det	_		2 High 2 Major	Major	2 6	i Indefined	Indefined Murdock Scocallahan	cdcallahan	10/7/96	4/2/98 No
AIM AIM2	W W	8049 Problem	, ,			<u></u>		3	}				
	7 11 4	7090 Deoblem	2000 Deckler NT Hummingbird windows top left edit selection	Engineerin Deferred	Seferred	To Be Defi 2 Major	2. Major	180	Undelined	Undelined Murdock, Scochrisc	ccchrisc	10/3/96	2/3/98 No
AIM+AIM2	ž.	1000	menu not working					! !					
A114. A1143	AIRA	7952 Problem	7952 Problem units were incorrect when two voltages wer	Test Engin Deferred	,	3. Medium 2. Major	2. Major	180	Undefined	Undefined Gazetta, Mari Ming	ariMing	10/2/96	0N.66/6/8
		·	multiplied								,:		
AIM, AIM2	AIM	7904 Problem	7904 Problem flashing between scope start-up window and tel	lest Engin Deforred		2. High	2. Major	180	0	Murdock, Sck Ming	cc Ming	10/1/96	27.3798 Yes
			error window error window	System	Deferred	P. Modium 2 Maior		TAD	Prior Beleg	Prior Beles Murdock: Scc paryk	ccaarvk	9/13/96	9/5/99 No
AIM AIM?	¥¥	7005 Problem	Can I Dilli W/ dilli User will	ı				2			1		

System Deferred 2 High 2 Major TBD System Deferred 2 High 1 Critical TBD System Deferred 2 High 1 Critical TBD System Deferred 2 High 1 Critical TBD System Deferred 10 Be Defit 2 Major TBD Customer Deferred 10 Be Defit 2 Major TBD Customer Deferred 10 Be Defit 2 Major TBD Customer Deferred 2 High 2 Major TBD Customer Deferred 2 High 2 Major TBD System Deferred 2 High 2 Major TBD System Deferred 2 High 2 Major TBD Customer Deferred 2 High 2 Major TBD System Deferred 2 High 2 Major TBD Sy	10414	MACT	10554	10554 Problem suito declaration of unconnected connection	System	Deferred	o Tich	2 Maior	TAD	Prior Refe	Prior Refes Christen, Ern ar	6/23/93	8/2/33 res
West 1022 Problem reparts of course System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Courses Employ System Deterred 2 High 2 Major 1700 Prob Resid Course Employ System Deterred 2 High 2 Major 180 Prob Resid Course Employ System System Deterred 2 High 2 Major 180 Prob Resid Course Employ System System Deterred 2 High 2 Major 180 Prob Resid Course Employ System Syste	SimulatorimAS	- 20		Storon	•				1		-		
1017 Poberinal content of the part of		1000	0 60000	Supplementation of the	System	Deferred	,		TBD	Prior Bels	Chasten Ernipati	6/16/93	0N 66/5/8
MAST (1922 Probabil register) political and controlled by Same Mast (1922 Probabil register) political and controlled by Same Mast (1922 Probabil register) political and controlled by Same Mast (1922 Probabil register) political and controlled by Same Mast (1922 Probabil register) political and controlled by Same Mast (1922 Probabil register) probability of the same factors and controlled by Same Mast (1922 Probabil register) probability of the same same same same same same same sam	Simulator/MAS I	MASI	77001		200						Chairles Fro do	3/8/93	9/5/99 No
MAST 1017 Poberim Variante of Variante Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1917 Poberim Variante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production of Mariante System Delivered 2 Hyp 2 Major 180 Prior Resecritation 1 1952 Production 1 1953 Production 1	Simulator/MAST	MAST	10423.F	Problem frequency domain modeling	System	Celerred			180	Prof Hel	Consider, Erre der		014.00/3/0
System Colored State	TSAMMOON	MAST	10344 F	Problem auto var creation	System	Deferred			T80	Prior Rele	e Christen, Ern ar	76/11/11	ON 66/6/0
Standard Mark Mark 1017F) Problem Land Address System Obtered 2 Hyg 2 Augr 180 Part Releg Christon, Eric Christon 6493 859	SCHOOL STATE	TOWN	10217	Sockiem two antivalence r. mom and number	System	Deferred			TRO	Prior Rele	Christen, Ern boonley	10/15/92	8/5/99 No
SemilationANS MAST	Simulator/MAS	MAG			Custom	0			2 6	Driot Dale	Chasten Frn at	8/31/92,	7/16/99 Yes
SemulaconAST WAST (0007) Podemical copies of countries of plants (1997) Podemical copies of plants (1997) Podem	Simulator/MAST	MAST	10272	Problem enum and external) yalalı	חבום ופס			2		Section 1	100/40/3	8/5/00 N/A
Maintaine/MAST MAST 1016 Propagation accessment of places System Deletines 2 May 180 Proof Places Charlette, Err Catalogue 180 Proof Places Charlette, Err Catalogue 180 Proof Places 180 P	Simulator/MAST	MAST	10207	Problem union initialization	System	Deferred		2 Major	180	Prior Hele	SCAMSION, CITICALISIEN	20,07,0	0 1 5 C C C C C C C C C C C C C C C C C C
Main Control Mai	TOWNSON	MAST	10169 F	Problem duplicate declaration of pl_set	System	Deferred		2. Major	180	Prior Rele	Christen, Ern christen	76/8/4	
SmulacoMAST MAST (1016) Folgemin ergod is noncropate annable annable with the control process and a goal of home and goal of	DUM/ORDERO		3.7464	Daylem if expression semantics	System	Deferred	10	2 Maior	TRO	Prior Rele	Christen, Ern ar	3/25/92	
Samulation/MAST MAST 1010 Proposition again to contact the state of the contact o	Simulator/MAS!	MASS	6		Siretom	900	i e	5	3.5	- Indofine	Christen Frn here	2/18/92	8/5/99 No
Simulation/MST 1987 (1009) Probential in the international part of the	Simulator/MAST	MAST	10115.1	Problem Sinc Delaurs vs instance delaurs	yaldılı			v	2	Condense		2/11/02	A/5/99 No
SympationAMS 1 State Strondom State Strongom Control of System Control of Control of Control of Control of Control of Control of Control o	TSW/Joteline/ST	MAST	10109	Problem argdet in incomplete template: enum	System	Deferred		رز.	TB0	Prior Hele	ie Christen, Erri darrell		24 00 110
SmitatonAAST SabeSimulaton State Caracterinate Base in converge on circuit has SPICE can present conversationAST SabeSimulation State Caracterinate Base in converge on circuit has SPICE can be conversationAST SabeSimulation State Caracterinate Base in conversation	10	MACT	10101	Problem small signal 4th paremeter is number	System	Deferred	2 High		180	Prior Rele	EChristen, Ern alan	76///2	ON 55/C/9
Smulator/MAST State/Smulator (1976) Problem (September) State-Smulator/MAST State-Smul	Simulator/MASI	MAG	1	Described European Principle in renumbering alo	System	Deferred	High		*	Prior Rele	Christen, Ern christen	7/12/91	8/23/99'No
SmulatorMAST SaberSmulation 1976 probent National Language and Conveyagened and American SamulatorMAST SaberSmulation 1986 probent National SamulatorMAST SaberSmulation 1986 probent National SaberSmulator National SaberSmulator National SaberSmulator National National SaberSmulator National National SaberSmulator National National SaberSmulator National SaberSmulator National SaberSmulator National SaberSmulator National SaberSmulator National National SaberSmulator National S	Simulator/MAST	SaberSimulator		CONTRACTOR OF THE PROPERTY OF	Charlomor	-			, C		Goldneisser Tharold	10/8/98	10/6/99 Yes
SamuelatorMAST SaberSimulation 8828 Probber National Probers (1840) Processes back agromatis to partner of Sharm SaberSimulation 1840 Processes back agromatis to partner of Sharm Deferred 2 High 2 Major 1850 Processes Processes Deferred 2 High 2 Major 1850 Processes Deferred 2 High 2 Major 1850 Processes Processes Deferred 2 High 2 Major 1850 Processes Deferred 2 Hig	Simulator/MAST	SaberSimulatori		Problem'Saber can I converge on circuit inal or ion can	relicited of		A S. Mediun	v	2	?	100		
Samulation/MAST SaberSimulation/MAST SaberSimulatio	!		_	handle				-					. 44 00000
Simulation/MAST SuberSimulation (MST) SuberS		S. Charle	8008	Problem FRROR MESSAGE	System	Deferred	2. High		180	Prior Rele	KVIach, Martin ar	2/14/90	0N 66/6/9
Simulation/MAST SuberSimulation 2016 Problem (Sate and anguments to partner of Sixth States) SuberSimulation 2016 Problem (Sate and Sate a	SimulatoriMAS	Calcinition of the calcinition	0.00	Posting Mixed mode in DT	System	Deferred	E E	2 Major	TRO	Prior Rele	Vlach, Martin Jabarby	8/31/90	8/9/99 Yes
Simulation/MAST SuberSimulation (2014) Problem (201	Simulator/MAS1	SaberSimulator	200	Trouble in the second of the s	Engineering		1		2 6	40.1	Deferred Our roof	8/11/8	10/14/99 No
SamulatoriAAST SaberSimulation (1975) Froblem (1975	SimulatorMAST	SaberSimulator	22044	Problem Saber passes bad arguments to partition	L'inglineau		7	nic major	2		100000	10/2/01	8/16/00 No
Suber-Simulation (ASS) Suber-Simulation (1978) Problem (Connergance alik with 5 of Suber-Simulation (1978) Problem (Connergance alik (Conn	Cimilator/AAA ST	SaberSimulator		Problem STATE ARRAY PERF HIT	System	Deferred			180	Prior Hele	E VIACO, MARINI GALLEII	06/0/01	ON CENDING
Simulator/MAST Sabe/Smulator 2556 problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (1.2) Sabe/Smulator (2004) Problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (2004) Problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (2004) Problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (2004) Problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (2004) Problem (Convergence talls with 5 0, works with 4.3 Customer Deferred 2 Middlem (2004) Problem (Convergence talls with 5 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 5 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2004) Problem (Convergence talls with 2 0, works with 4.3 Convergence (Incident of 2 0, works with	TO THE TOTAL STREET	Cabacimulator		Problem VALUES SECT. EVALUAT	System	Deferred			TBD	Prior Rele	KVIach, Martin Jabarby	10/10/30	9/16/31/8
Simulator/MAST Sabe/Simulator (ASST) Poblem (Lordon Problem) State Christian Enfischment (ASST) Sabe/Simulator (ASST) Poblem (Lordon Problem) State Christian Enfishment (ASST) Sabe(Simulator) (ASST) Poblem (Bart and Problem) State Christian Enfishment (ASST) Sabe(Simulator) (ASST) Poblem (Lordon Problem) State Christian Enfishment (ASST) Sabe(Simulator) (ASST) Poblem (Christian Enfishment (ASST) Sabe(Simulator) (ASST) Poblem (Bart and Problem) State (Christian Enfishment (ASST) Sabe(Simulator) (ASST) Poblem (Christian Enfishment (ASST) Poblem (Chr	Simulator/MAS	SaberSimulator		Description of the second fails with 5.0 works with 4.3.1	Customer		-	-	TBO		Christen, Ernipchapman	66/61/2	10/22/99 Yes
Smulator/MAST Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Sabe-Simulator (1973) Problem/ Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Act Analysis becomes awfull slow over 50k Hz (1974) Sabe-Simulator (1973) Problem/ Sabe-Simulator (1974)	Simulator/MAST	SaberSimulator	_:	Problem Land Companies in the series of the		_	1	<u>:</u> .			Charles Ero rachwein	6/11/60	7/16/99 Yes
Simulator/MAST Sabe-Simulator (1952) Problem (Lu decomposition titles) and simulator/MAST Sabe-Simulator (1953) Problem (Lu decomposition titles) and simulator/MAST Sabe-Simulator (1953) Problem (Lu decomposition titles) and simulator/MAST Sabe-Simulator (1954) Problem (Lu decomposition titles) and simulator/MAST Sabe-Simulator (1954) Problem (Lu decomposition titles) and simulator/MAST Sabe-Simulator (1954) Problem (Lu decomposition titles) and simulator/MAST (Lu decomposition titles) and simula	Simulator/MAST	SaberSimulator	20942	Problem Saber crashes after memory problem	Customer			Ξ,	GH.	4.3.2 Par	Chillisteri, Criticischiwerg		100/2/04 *IV
Simulator/MAST SaberSimulator 19739 Problem (Lut decomposition talis) Simulator/MAST SaberSimulator 19739 Problem (Lut decomposition talis) Simulator/MAST SaberSimulator 19739 Problem (Lut decomposition talis) Simulator/MAST SaberSimulator 19739 Problem (Lut disease) Simulator/MAST SaberSimulator 19739 Pr	CimidatorAAAST	SaberSimulator		Problem Fundamental bug in renumbering alg.	System	Delerred	Z High	_	180	Pror Hele	E Deferred Clue miked	16/61/1	081 667 101
Simulator/MAST SaberSimulator (1985) problem lack enrolation proces incorrect notation with significant separation of simulator/MAST SaberSimulator (1986) problem lack enrolation with significant separation of significant separation (1986) problem lack of significant separation (1986) problem lack enrolation (1986) problem (1986) prob		CabarSimulator		Problem LU decomposition fails	Beavenor		2: High	ci	180	4.3.2	Christen, Ern christen	66/8/6	9/13/99 Y 65
Simulator/MAST SaberSimulator 1736 Problem different step size gets different results for xinds Simulator/MAST SaberSimulator 1736 Problem different step size gets different results for xinds Simulator/MAST SaberSimulator 1736 Problem Grant results when using tracolituse Simulator/MAST SaberSimulator 1736 Problem Control Results when using tracolituse Simulator/MAST SaberSimulator 1512 Problem Problem associated with attention of Simulator/MAST SaberSimulator 1513 Problem Results when using tracolituse Simulator/MAST SaberSimulator 1513 Problem Results when using tracolituse Simulator/MAST SaberSimulator 1513 Problem Results when using tracolituse Results	Simulator/MAS	Section Control	:	Punhlem Rack annotation gives incorrect results	Beavenon		To Be De	<u>ہ</u>	180	5.0	Goldgeisser, davec	2/11/99	9/23/99 No
Simulator/MAST Sabe/Simulator 1939 problem luternal separations Simulator/MAST Sabe/Simulator 1938 problem luternal separations of Simulator/MAST Sabe/Simulator 1938 problem luternal separation set of the continue of the c	Simulator/MASI	SaperSimulator		Original distriction are a set of different results for with	Test Foot	Deferred	-	O Major	TBD	5.0	Christen, Ernlihu	1/8/99	9/23/99 No
Simulator/MAST SaberSimulator 17389 Problem Binasini service of trigger 17339 Problem Binasini service of trigger	Simulator/MAST	SaberSimulator		Problem united and state gets official record to	Tool India	2000	+	o Major	3.6	9 0	Coldonisser in	11/19/98	9/23/99 No
Simulator/MAST SaberSimulator 17289 Problem Different results when using troohling Simulator/MAST SaberSimulator 17118 Problem Different results when using troohling Simulator/MAST SaberSimulator 17118 Problem Wast I was a mequation was no system Deferred 2. High 2. Major 1810 4.3 i Goldgeisser, ischweig 9/21/98 9/21/98 Simulator/MAST SaberSimulator 17118 Problem Wast I was not adaylor was no system Deferred 2. High 2. Major 1810 4.1 i Deferred Que callahan 11/16/98 7/19/98 7/1	Simulator/MAST	SaberSimulator		Problem spmosim 132 (egression les) ger misomos melorid		r Delerred	90 O		2	0.0	Conspersaci. Just		
Simulator//AKST SaberSimulator 17289 Problem Different results when using if conlinue Customer Deferred To Be Deli 2 Major TBD 4.3 Patch Deferred Localenha Can not see Inductance VAL from LNLC Customer Deferred TBD 4.3 For the Configuration of TST2 Problem Probl				"ALG ITERATIONS" Too many lierations					! ا ا			00,000	
Simulator/MAST SaberSimulator 17118 Problem Can not see Inductance VAL from LNLC Customer Deferred 7 Be Deferred 1 Medium 2 Major 18D 4 3 Christen, Em boonley 11/4/38 Simulator/MAST SaberSimulator 15721 Problem Problem associated with altering temperature in SaberSimulator 15721 Problem South PFEX SaberSimulator 15731 Problem South PFEX SaberSimulator 1533 Problem South PFEX SaberSimulator 1534 Problem South PFEX SaberSimulator 1535 Problem South PFEX SaberSimulator 1534 Problem South PFEX SaberSimulator 1534 Problem South PFEX SaberSimulator 1535 Problem South PFEX SaberSimulator 1534 Problem South PFEX SaberSimulator 1535 Problem South PFEX SaberSimulator 1535 Problem South PFEX SaberSimulator 1535 Problem South PFEX SaberSimulator 1531 Problem South PFEX SaberSimulator 1534 Problem South PFEX SaberSimulator 1535 Problem So	TOVER	CaberSimulator	:	Problem Different results when using ir continue	Customer			12 Major	T80	4.3 - Patc	h Deferred Que callahan	11/16/98	3/8/99 Yes
Simulator/MAST SaberSimulator 15219 Problem Problem associated with altering temperature in Beaverton Deferred 3: Medium 2: Major TBD 4.3 i Goldgeisser, Ischweig 9/2/198 Simulator/MAST SaberSimulator 15219 Problem RAS-99: FT time interval (kbegin + period) >> Customer Deferred 3: Medium 2: Major TBD 4.1 i Deferred Out ischweig 5/15/98 Simulator/MAST SaberSimulator 15219 Problem SooM PFEX Simulator/MAST SaberSimulator 15239 Problem Wast romand to the series of	Simulator/MA31	October State of Stat		Problem Can not see Inductance VAL from LNLC	Customer		1	2. Major	TBD	6.4	Christen, Ern bconley	11/4/98	9/8/99 Yes
Simulator/MAST SaberSimulator	Simulator/MAS I	Saperalificiator		etal ce et							-		
Simulator/MAST SaberSimulator 1521 Problem Problem associated with altering temperature in Samulator/MAST SaberSimulator 1521 Problem Problem associated with altering temperature in SaberSimulator 1521 Problem SA-99: FFT time intervall (xbegin + period) -> Customer Deferred 3: Medium 2: Major TBD 4:11 Deferred Out reschweig 5/15/98 Simulator/MAST SaberSimulator 15319 Problem Sook PEX Simulator/MAST SaberSimulator 1539 Problem Sook PEX Simulator/MAST SaberSimulator 1539 Problem Sook PEX Simulator/MAST SaberSimulator 1539 Problem With stressmeasurelist command System Deferred 2: High 2: Major TBD 4:11 Christen. Err Johnson 10/16/97 Simulator/MAST SaberSimulator 1539 Problem with stressmeasurelist command System Deferred 2: High 2: Major TBD 1: Christen. Err Johnson 10/16/97 Simulator/MAST SaberSimulator 1539 Problem with stressmeasurelist command System Deferred 4: Low 2: Major TBD 1: Christen. Err Johnson 10/16/97 Simulator/MAST SaberSimulator 1539 Problem with stressmeasurelist command System Deferred 2: High 2: Major TBD 1: Christen. Err Johnson 10/16/97 Simulator/MAST SaberSimulator 1539 Problem with stressmeasurelist command System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 5/17/95 Simulator/MAST SaberSimulator 1539 Problem with stressmeasurelist command System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 5/17/95 Simulator/MAST SaberSimulator 1539 Problem (to the exception System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 5/17/95 Simulator/MAST SaberSimulator 1528 Problem (to the exception System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 1/17/95 Simulator/MAST SaberSimulator 1528 Problem (to the exception System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 1/17/95 Simulator/MAST SaberSimulator 1528 Problem (to the exception System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 1/17/95 Simulator 1528 Problem (to the exception System Deferred 2: High 2: Major TBD 1: Christen. Err Division. 1/17/95 Simulator 1528 Problem (to the except				Date of Congression (Property 1)	Cietomer		P. Mading		Cat	1 2 7	Goldonisser Ischweig	9/21/98	9/29/99 Yes
SaberSimulator 15721 Problem Problem associated with attenting temperature in Deferred 2. High 2 Major 15139 Problem Sociated with stressmeasurelist command 2. High 2 Major 1780 Problem Sociated went statement of the SaberSimulator 12311 Problem Sociated Washington 12311 Problem Sociated Mark 1 Major 1780 Problem Poleman Pole	Simulator/MAST	SaberSimulator	/69.	Problem D1-Conveigence (miggai_1)) Medicil		2 1	2 (Secretary Colors	7/0/08	7/16/99 Vec
Simulator/MAST SaberSimulator 15219 Problem SA-99: FTT time interval (kbegin + period) -> Customer Deferred 3: Medium 2: Major TBD 4:11 Deferred Que rschwerg 5/12/98 Simulator/MAST SaberSimulator 15193 Problem SooM PEEX Simulator 12388 Problem vin kitres and soon of the customer deferred and soon of the custom	Similator/MAST	SaberSimulator	15721	Problem Problem associated with altering temperature in	Deavertor		Z. High		081		Haug, Matsor Jarapa	2000	10033 163
Simulator/MAST SaberSimulator 1519 Problem SA-99: FFT time interval (xbegin + period) -> Customer Deferred 3: Medium 2 Major TBD 4.1.1 Deferred Out richage 5/12/98 Simulator/MAST SaberSimulator 12319 Problem Hollow with stressmeasurelist compared during Calaveras Simulator/MAST SaberSimulator 1236 Problem MAST NO_NUM_EQUATION sends the customer the wrong way! The error implied that simulator/MAST SaberSimulator 9329 Problem 100u < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 certop or simulator/MAST SaberSimulator 12319 Problem 100u < 1000 < 1000 < 1000 certop or simulator/MAST SaberSimulator 12310 Problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u < 1000 certop or simulator/MAST SaberSimulator 12310 problem 100u certop or simulator/MAST SaberSimulator 12310 problem 100u certop or simulator/MAST SaberSimulator/MAST SaberSimulator 12310 problem 100u certop or simulator/MAST SaberSimulat				the design	:					:	1		
Simulator/MAST SaberSimulator 15193 Problem AC Analysis becomes awfull slow over 50k Hz Simulator/MAST SaberSimulator 1539 Problem AC Analysis becomes awfull slow over 50k Hz Simulator/MAST SaberSimulator 12319 Problem Problem Problem with stressmeasurelist command Simulator/MAST SaberSimulator 1539 Problem With stressmeasurelist command Simulator/MAST SaberSimulator 1539 Problem With stressmeasurelist command Simulator/MAST SaberSimulator 1539 Problem 100u < 10u <	_	CabarCimulator		Problem SA-99; FFT time intervall (xbegin + period) ->	Customer		3. Medium	2 Major	180	14.1	Deferred Querschweig	5/15/98	2/3/99 Yes
SaberSimulator 15193 Problem AC Analysis becomes awfull slow over 50k Hz SaberSimulator 12311 Problem SoM PFEX SaberSimulator 12311 Problem Was no saberSimulator 1538 Problem Tool of the word was no saberSimulator 1538 Problem Tool of the word was no saberSimulator 1538 Problem Tool of the word was no saberSimulator 1538 Problem Tool of the word was no saberSimulator 1538 Problem Was no saberSimulator 1539 Problem Was no saberSimulator 1539 Was no saberSimulator 1539 Was no saberSimulator 1539 Was no saberSimulator 1539 Was no saberSim				IFFT ⇒ ERROR	•			_			-		
SaberSimulator 12311 Problem SoM PEX SaberSimulator 12311 Problem With stressmeasurelist command SaberSimulator 12318 Problem Was no saberSimulator 12319 Problem With stressmeasurelist command System Deferred 2: High 2: Major 1231 Problem With stressmeasurelist command System Deferred 2: High 2: Major 1231 Problem With stressmeasurelist command System Deferred 2: High 2: Major 1240 Christen. Ern callahan 3/14/97 SaberSimulator 9329 Problem Toole 4: 1000 Christen. Ern Dkelly 12/4/95 SaberSimulator 9329 Problem Core during coverilow				Problem AC Analysis becomes awfull slow over 50k Hz	Customer	Deferred	3. Medium		TBD	421	Christen, Ern Ithapar	5/12/98	8/8/99 Yes
SaberSimulator 12866 Problem event scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling does not check refurn code SaberSimulator 12311 Problem vent scheduling vent vent code scheduling vent vent code scheduling vent vent code scheduling vent vent code vent vent code vent vent code vent vent code vent vent vent vent vent vent vent ven	Simulator/MAS	OBDITION DEC		that tuns fast with smaller free									
SaberSimulator 12886 Problem event scheduling does not check relum code Field AE Deferred 2: High 2 Major TBD 4.1.1 Christen, Ern bowmanm 10/16/97 (2.14/97) SaberSimulator 1538 Problem with stressmeasurelist command System Deferred 2: High 2 Major TBD 4.1.1 Christen, Ern ar 3/14/97 (2.14/97) SaberSimulator 9455 Problem 100u < 1000 < 1000 < 3/24/95 (3.14/97) SaberSimulator 9455 Problem 100u < 1000 < 1000 < 3/24/95 (3.14/97) SaberSimulator 9455 Problem 100u < 1000 < 1000 < 3/24/95 (3.14/97) SaberSimulator 9455 Problem 100u < 1000 < 1000 < 3/24/95 (3.14/97) SaberSimulator 9455 Problem 100u < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1000 < 1				DO TOTAL COOK DEEX	Svetern	Deferred	1	2 Maior	Tan	Dring Rala	Christen Ernicraios	5/9/91	8/23/99 Yes
SaberSimulator 12886 Problem With stressmeasurelist command SaberSimulator 1538 Problem Work with stressmeasurelist command SaberSimulator 1538 Problem Work wong way! The error implied that an equation was no saberSimulator 9455 Problem 100u < 10u < 1	Simulator/MAST		•	ribbilities and the design of the live code	Field AF			No.	201		Christen Em howmann	10/16/97	12/7/98 Yes
SaberSimulator 12311 Problem Vals Incorrectly Computed Daily Caracteristics Carac	- Simulator/MAST	-			Beauton			Z	; 2.6	1.6	Defend One daylor	7/14/97	ON 99/2/01
Toliback Toliback Toliback SaberSimulator 1538 Problem Problem With stressmeasurelist command Field AE Deferred 2. High 2. Major TBD 4.1.1 Christen, Ern ar 5/17/96 B SaberSimulator 9536 Problem MAST_NO NUM EQUATION sends the System Deferred 2. High 2. Major TBD 3.4 Christen, Ern Dkelly 12/4/95 B SaberSimulator 9329 Problem 100u < 1000 - 5 System Deferred 2. High 2. Major TBD 3.4 Christen, Ern Dkelly 12/4/95 B SaberSimulator 9329 Problem 100u < 1000 - 5 TBD 3.4 Christen, Ern Dkelly 12/4/95 B SaberSimulator SaberSimulator Deferred 2. High 2. Major TBD 3.4 Christen, Ern Dkelly 12/4/95 B SaberSimulator Deferred 2. High 2. Major TBD 3.4 Christen, Ern Dkelly 12/4/95 B SaberSimulator Deferred 2. High 2. Major TBD 3.4 Christen, Ern Dkelly Deferred Deferred 2. High 2. Major TBD 3.4 Christen Deferred Deferred Deferred 2. High 3.4 Christen Deferred De	Simulator/MAST			Problem Vals incorrectly computed doming Caravellas	DIBABOO		90 90 o	z. wajor	2	Danie DuO			
SaberSimulator 11538 Problem With stressmeasurelist command red AE Deferred 2: High 2 Major 18D 4.1.1 Christen, Erri calianam 5/17/96 8 SaberSimulator 9536 Problem WAST_NO NUM EQUATION - sends the System Deferred 4: Low 2. Major 18D 4.0 Christen, Erri arr 5/17/96 8 System Deferred 2: High 2 Major 18D 4.0 Christen, Erri davec 7/31/95 8 SaberSimulator 9329 Problem Core dump in ron hp; exception System Deferred 2: High 2 Major 18D 3.4 Christen, Erri davec 7/31/95 8	\ T			rollback	1				!	· .		7014110	200,000
SaberSimulator 9536 Problem MAST_NO_NUM_EQUATION sends free System Deferred 4: Low 2. Major TBD 4.0 Christen, Erri Arr 377730 customer the wrong way! The error implied that an equation was no saberSimulator 9455 Problem 100u < 100u	TSimulator/MAST	•	_	Problem Problem with stressmeasurelist command	rield AE	Deferred		2 Major	90	1.1.1	Christen, Ern cananan	3/14/9/	9//99 165
customer the wrong way! The error implied that an equation was no SaberSimulator 9455 Problem 100u < 100e -6 SaberSimulator 9329 Problem Core dump in tr on hp.exception SaberSimulator 9329 Problem Core dump in tr on hp.exception SaberSimulator 9329 Problem Core dump in tr on hp.exception SaberSimulator 9329 Problem Core dump in tr on hp.exception SaberSimulator 9329 Problem Core dump in tr on hp.exception SaberSimulator 9329 Problem Core dump in tr on hp.exception	CimilatorAAAST			Problem MAST_NO_NUM_EQUATION sends the	System	Deferred	4: Low	2. Major	TB0	0.4	Christen, Ern ar	26//1/6	0/10/33/165
SaberSimulator 9455 Problem 100u < 100e-6 System Deferred 2. High 2. Major TBD 4.0 Christen. Ern bkelly 12/4/95 SaberSimulator 9329 Problem Core dump in tr on hp.exception System Deferred 2. High 2. Major TBD 3.4 Christen. Ern davec 7/31/95 handling overflow				customer the wrong way! The error implied that				_			-		
SaberSimulator 9455 Problem 100u < 100e-6 System Deferred 2. High 2. Major TBD 4.0 Christen, Ern bkelly 12/4/95 System Deferred 2. High 2. Major TBD 3.4 Christen, Ern davec 7/31/95 saberSimulator 9329 Problem Core dump in tr on hp.exception System Deferred 2. High 2. Major TBD 3.4 Christen, Ern davec 7/31/95 handling overflow	Τ_	-		an equation was no		;	- :						
SaberSimulator 9329 Problem Core dump in tr on hp.exception System Deferred 2. High (2. Major TBD 3.4 Christen, Ern davec 7/31/95 saberSimulator 9329 Problem Core dump in tr on hp.exception		·		Problem 100u < 100e-6	System	Deferred		2. Major	180	04	Christen, Ern bkelly	12/4/95	8/16/99 Yes
	Simulator/MAS	SaberSimulator		Problem Core dump in tr on hp.exception	System	Deferred	High	2. Major	180	3.4	Christen, Ern davec	7/31/95	8/16/99 Yes
	SIMURACIONAL			handling overflow									

REEL: 011820 FRAME: 0409

Open Deferr. . roblem Reports - Critical and Major -

Simulator/MAST SaberSimulator 9266 Problem Alter complains of stamp change when it specifically should be shouldn't "FEROR "Altar would ca shouldn't "FEROR "Altar would ca shouldn't shou	and in scribed in seried in DC ng didn't work is				780 780 780 780 780 780 780	3.4 3.4 9.100 Reter	3.4 Christen, Erri tome 3.4 Vlach, Martin ar 3.4 Deferred Que lelanda 3.4 Vlach, Martin dm 3.4 Christen, Ern ar Prior Retee Vlach, Martin saswanso	3/24/95 3/21/95 3/14/95 1/25/95 1/16/95	8/5/99 No 8/5/99 Yes 8/5/99 Yes 8/5/99 No 8/5/99 No
SaberSimulator 9264 Problem Has a nettat that he's spent days to lind out that System changing the order fives tillAthough Ron Evans I changing works I control section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section section as well as those not, the control section as well as those not, the control section saber disconcise plot correct message section as well as those not, the control section saber disconcise to the correct of the correc			· · · · · · · · · · · · · · · · · · ·		180 180 180 180 180	3.4 3.4 3.4 Prior Relec	Viach, Martin ar Deferred Que lelanda Viach, Martin dm Christen, Ern ar Viach, Martin saswans		8/5/99 Yes 9/16/99 Yes 8/5/99 No 8/5/99 No
SaberSimulator 9257 Problem an trying to alter a profit L4 variable. Saber is Saber in Saber is Saber in Saber is Saber in Saber in Saber is Saber in Saber in Saber is Saber in Saber in Saber in Saber is Saber in Saber in Saber in Saber is Saber in Saber			E		780 780 780 780 780	34 34 Prior Relec	Deferred Que lelanda Vlach, Martin dm Christen, Ern ar Vlach, Martin saswans EViach, Martin saswans		9/16/99 Yes 8/5/99 Yes 8/5/99 No
SaberSimulator 9211 Problem compile error in Im 39 SaberSimulator 9200 Problem do page ahoud show dependencies specified in System Deferred 3. Medium 2. SaberSimulator 9196 Problem grants and the low end in DC. System Deferred 2. High 2. SaberSimulator 9014 Problem grants and the low end in DC. System Deferred 2. High 2. SaberSimulator 8867 Problem Error Message SaberSimulator 8867 Problem Saber distornoise pt core dump System Deferred 2. High 2. SaberSimulator 8868 Problem saber distornoise pt core dump System Deferred 2. High 2. SaberSimulator 8869 Problem integrator malfunction SaberSimulator 8869 Problem integrator malfunction SaberSimulator 10735 Problem integrator methods of SaberSimulator 10735 Problem integrator methods of SaberSimulator 10735 Problem integrator methods of SaberSimulator 10897 Problem integrator methods of SaberSimulator 10897 Problem in SSP a component library part is still listed as System Deferred 2. High 2. SaberSimulator 10873 Problem in SSP a component library part is still listed as System Deferred 2. High 2. SaberSimulator 10873 Problem in SSP a component library part is still listed as System Deferred 2. High 2. SaberSimulator 10873 Problem in SSP a component library part is still listed as System Deferred 2. High 2. SaberSimulator 10873 Problem Threshold crossings are different for 'order 1' System Deferred 2. High 2. SaberSimulator 10873 Problem Threshold crossings are different for 'order 1' System Deferred 2. High 2. SaberSimulator 10873 Problem Threshold crossings are different for 'order 1' System Deferred 2. High 2.	· · · · · · · · · · · · · · · · · · ·		E		780 780 780 780 780 780	3.4 Prior Relec	Viach, Martin dm Christen, Ern ar EViach, Martin saswans Grinisten, Ern christen		8/5/99 No 8/5/99 No 8/5/99 No
SaberSimulator 9196 Problem q2n4378 Out of regions on the low end in DC System Deferred 2: High 2 horizontation of the problem saber distonoise pt core dump SaberSimulator 8670 Problem Error Message SaberSimulator 8718 Problem aber distonoise pt core dump System Deferred 2: High 2 SaberSimulator 8718 Problem aber distonoise pt core dump System Deferred 2: High 2 SaberSimulator 8718 Problem integrator meltunction and System Deferred 2: High 2 SaberSimulator 10755 Problem integrator meltunction System Deferred 2: High 2 SaberSimulator 10759 Problem simulation saber memory config 2 System Deferred 2: High 2 SaberSimulator 10721 Problem simulation saber distribution of 10751 Problem integrator message SaberSimulator 10721 Problem simulation saber distribution 10721 Problem Simulation saber distribution 10721 Problem In SSP a component library part is still listed as System Deferred 2: High 2 SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2: High 2 SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2: High 2 SaberSimulator 10673 Problem In SSP a component library part is still listed as System Deferred 2: High 2 System Deferred 2: High 2 SaberSimulator 10673 Problem The should disseppear. System Deferred 2: High 2 System Deferred 2: High 3 System Deferred 2: High 3 System Deferred 3: High 3 System Deferred 4: High 3 System Deferred 4: High 3 System Deferred 4: High 3 System Deferred 5: High 3 System Deferred 5: High 4 System Deferred 5:		Deferred 2 Deferred 2 Deferred 2 Deferred 2 Deferred 2			780 780 780 780	Prior Relea	EVIach, Martin saswans		8/5/99 No
SaberSimulator 8680 Problem saber distornoise pf core dump System Deferred 2 High 2 SaberSimulator 8718 Problem saber falled on compiling bit, hemt and mestal System Deferred 2 High 2 SaberSimulator 8680 Problem integrator matfunction SaberSimulator 10725 Problem integrator matfunction SaberSimulator 10729 Problem sabermemory config 2 System Deferred 2 High 2 SaberSimulator 10729 Problem sabermemory config 2 System Deferred 2 High 2 SaberSimulator 10729 Problem sabermemory config 2 System Deferred 2 High 2 SaberSimulator 10729 Problem sabermemory config 2 System Deferred 2 High 2 SaberSimulator 10673 Problem the sabermemory of the samulation in 3.1 nn System Deferred 2 High 2 SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2 High 2 GaberSimulator 10673 Problem in SSP a component library part should disappear. SaberSimulator 10653 Problem Threshold crossings are different for order 1 System Deferred 2 High 2 SaberSimulator 10653 Problem Threshold crossings are different for order 1 System Deferred 2 High 2 SaberSimulator 10653 Problem Threshold crossings are different for order 1 System Deferred 2 High 2 SaberSimulator 10653 Problem Threshold crossings are different for order 1 System Deferred 2 High 2 SaberSimulator 10653 Problem Threshold crossings are different for order 1 System Deferred 2 High 2 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem Threshold crossings are different for order 1 SaberSimulator 10653 Problem SaberSimulator 10653 Problem SaberSimulator 2 SaberSimulator 10653 Problem SaberSimulator 2 SaberSimulator 10653 Problem SaberSimulator 2 SaberSimulator 1 SaberSimulator 2 SaberSimulator 2 SaberSimu		Deferred 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	H H H H H H B B B B B B B B B B B B B B	1 1 1	780 780 780	Prior Relec	Christen, Ernichristen		
SaberSimulator 6680 Problem integrator malfunction SaberSimulator 6680 Problem integrator malfunction SaberSimulator 10721 Problem samulation: saber: grainsteps SaberSimulator 10673 Problem saber: grainsteps SaberSimulator 10673 Problem in SSP a component library part should GaberSimulator 10673 Problem in SSP a component library part should disappear. SaberSimulator 10653 Problem in SSP a component library part should disappear. SaberSimulator 10653 Problem in SSP a component library part should disappear. SaberSimulator 10653 Problem in SSP a component library part should disappear. SaberSimulator 10653 Problem Threshold crossings are different for 'order 1' System Deferred 2: High 2. SaberSimulator 10653 Problem Threshold crossings are different for 'order 1' System Deferred 2: High 2.	System System System System		oi'o	Maior		Prior Relet	Prior Rele¢ Deferred Que christen Prior Rele¢ Deferred Que ghruan	8/23/94 5/26/94 4/28/94	0N, 66/21/01 0N, 66/2/01 0N, 66/7/01
SaberSimulator 10755 Problem simulation:saber:resianslesses System Deferred 3: Medium SaberSimulator 10755 Problem simulation:saber:resianslesses System Deferred 3: Medium SaberSimulator 10729 Problem simulation:saber:resianslesses System Deferred 2: High SaberSimulator 10721 Problem Saber:dc:lsonr HP-only no_ramp alg System Deferred 2: High SaberSimulator 10697 Problem The ram2k circuit, which previously ran in 3.1, no System Deferred 2: High longer completes the simulation in 3.2. Running SaberSimulator 10673 Problem In SSP a component library part is still listed as System Deferred 2: High disappear. SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Syste	System System System		·	1	081	Prior Relea	Prior RelegChristen, Ern ricke	3/29/94	8/5/99 Yes
SaberSimulator 10755 Problem simulation: saber memory contig 2 SaberSimulator 10721 Problem simulation: saber: stairsteps SaberSimulator 10673 Problem The ram2k circuit, which previously ran in 3.1, no System Deferred 2: High SaberSimulator 10673 Problem In SSP a component library part is still listed as System Deferred 2: High disappear. SaberSimulator 10673 Problem In SSP a component library part is still listed as System Deferred 2: High disappear. SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High disappear.	System	ì	Mediumiz	ī	<u>8</u>	Prior Relec	Prior Relect Deterred Quear	10/29/93	ON 66/1/01
SaberSimulator 10673 Problem simulation satisfactors of the same saber if satisfactors of the saberSimulator 10697 Problem saber circuit, which previously ran in 3.1, no System Deferred 2: High 2: SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2: High 2: Agnosial sapers in the same same satisfactors of the same same satisfactors of the same same same satisfactors of the same same same same same same same sam			Ε	7	180	Prior Helek	Prior Heliet Christelli, Emiliere	10/18/93	9/3/99 Yes
SaberSimulator 10673 Problem The ram2k circuit, which previously ran in 3.1, no System Deferred 2: High 2. SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2: High 2. Ag2n2222.143p.1. The qsp part should disappear. SaberSimulator 10653 Problem Threshold crossings are different for "order 1" System Deferred 2: High 2.			100	Major	<u> </u>	Prior Relec	Prior RelecColdgeisser, willf		oN 99/3/99
SaberSimulator 10673 Problem in SSP a component library part is still listed as System Deferred 2: High 2. disappear. SaberSimulator 10653 Problem Threshold crossings are different for order 1. System Deferred 2: High 2.	_		<u>~i</u>	Major	180	Prior Relea	Prior Refe¢Vlach, Martin dgbedros	66/01/6	ON SERVIN
SaberSimulator 10653 Problem Threshold crossings are different for order 1. System Deferred 2: High 2.	isted as				1BO	Prior Relec	Prior ReleaChristen, Ernichristen	8/30/93	9/3/99 Yes
	ခ		<u> </u>	Major	180	Prior Relea	Prior Reles Christen, Ern darrell	8/18/93	8/5/99 No
Simulator/MAST SaberSimulator 10602 Problem in a simple 1 transistor circuit to measure DC or System Deferred 2: High 2. Major Actional Simulator/MAST SaberSimulator (10602 Problem in a simple 1 transfer chars, the AC transfer (using vary I		į.	:	i	TB D	Prior Relea	Prior Relea Christen, Ern ar	7/26/93	8/5/99 ⁽ Yes
R System Deferred	System	i 1	1 !	\top	TB0	Prior Relect	Prior Relea Christen, Em boonley	7/7/93	9/7/99 Yes
SaberSimulator 10046 Problem hp 700 bullers 177 output System Deferred 2. High	System	1			000	Prior Relect	Prior Relec Christen, Ern ar	5/27/93	8/5/99 Yes
SaberSimulator 10009 Floblem saber mast: EXT IN ROOT origin System Deferred 3. Medium			11≝.	!	TBD	Prior Relea	Prior Reles Christen, Ern.ar	5/4/93	8/30/99 No
	never-			1	180 180	Prior Relection Relection	Prior Releč Goldgeisser, alistair Prior Relež Goldgeisser, alistair	3/24/93	8/5/99 Yes

Open Deferred Problem Reports - Critical and Major -

Simulator/MAST	SaberSimulator	10422 Problem	Simulator/MAST SaberSimulator 10422 Problem Wrong results. If vcvs. ud2 is changed for k=1u System to k=10u and r.tlin2 is changed to 10meg and	System	Deferred	Deferred 2 High 2 Major TBD	2. Major		Prior Rele≀	Prior Rele∈Christen, Ern ar		3/9/93	8/5/99 Yes
Simulator/MAST	SaberSimulator	10413 Problem	SaberSimulator 10413 Problem Saber: dc: oscillations	System	Delerred	2 High	2 Major	TBD	Prior ReleaC	System Deferred 2 High 12 Major TBD Prior Relet Goldgeisser, Christen	hristen	2/25/93	8/5/99/No
Simulator/MAST	SaberSimulator	10342 Problem		System	Delerred	2: High	2 Major	180	Prior ReleaC	Christen, Ernic	hriss	11/9/92	8/5/99 Yes
	SaberSimulator	10226 Problem	vs. hierarchy. Hier, it ok. See prob lites	System	Deferred	High F	2. Major	180	Prior ReleaC	Soldgeisser, a		6/26/92	8/5/99 Yes
Simulator/MAST	SaberSimulator	10206 Problem	cobian during TR	System	Deferred	Deferred 2. High 2. Major TBD	2. Major	180	Prior Relect	Prior Relec Goldgeisser, boonley	conley	5/28/92	6/30/99 No
Simulator/MAST	SaberSimulator 10179 Problem slow dc	10179 Problem		System	Deferred	2 Hoh	2. Major	180	Prior Relect	soldgeisser, le	alanda	4/17/92	8/30/99 Yes
Simulator/MAST	SaberSimulator	10178 Problem	SaberSimulator 10178 Problem ALG NO SOLUTION	System	Deferred	2. High	2 Major	ł .	4.3	soldgeisser, d	arrell	4/15/92,	8/5/99 No
Simulator/MAST	SaberSimulator	10113 Problen	SaberSimulator 10113 Problem During Ir, the next time step after a scheduled	System	Deferred	2. High	2. Major		Prior Releav	Prior Relea Vlach, Martin christen	hristen	2/18/92	8/23/99 No
			lime is independent of tstep if the analog time is					-				-	

Schedule 2.21(n).

None.

60488/4/TVW/052859-0047